

No.	2	3	-	5	8	

AUTHORIZING THE EXECUTION OF A GROUND LEASE WITH HALEWILIKO HIGHLANDS LP, TO DEVELOP A SENIOR AFFORDABLE RENTAL PROJECT AT THE FORMER AIEA SUGAR MILL SITE, 99-385 POHA! PLACE, AIEA, OAHU, HAWAII 96701 (TMK NO.: (1) 9-9-078:006).

WHEREAS, on August 6, 2018, pursuant to Revised Ordinances of Honolulu 1990 ("ROH"), Section 28-3.4, the Department of Land Management ("DLM") issued a Request for Proposals (DLM-1226505) ("RFP") to solicit proposals from qualified entities to plan, design, construct, operate, and manage, under a long-term leasehold interest, a senior affordable housing project at the former site of the Aiea Sugar Mill, located at 99-385 Pohai Place in Aiea, and identified as TMK No. (1) 9-9-078:006 (the "Property") to provide housing for low to moderate income households, and make ancillary improvements to the Property; and

WHEREAS, on or before October 3, 2018, developers submitted their development proposals in response to the RFP and subsequently on January 11, 2019, the qualified developers submitted their Best and Final Offers to the City; and

WHEREAS, on August 9, 2019, pursuant to ROH Section 28-3.4(c)(2), by Departmental Communication 571 (2019), DLM submitted to the City Council a written report identifying the proposal by EAH Inc., a California nonprofit corporation ("EAH"), as the selected proposal for development of the Project (as herein defined), and describing the City's negotiations with EAH regarding key elements of its proposal; and

WHEREAS, the Project will consist of a total of 139 affordable studio and one-bedroom rental units that will be available to individuals 62 years of age or older, whose household income is 60 percent or below of the area median income for Honolulu (plus one two-bedroom unit will be reserved for an on-site resident manager), in a three-story building with a common elevator core, surface parking lot, tech lounge, outdoor recreation area, and outdoor garden for use by residents as well as a community center available for use by residents and from which third-party services and programs may also be available to the public (the "Project"); and

WHEREAS, on October 9, 2019, the City Council adopted Resolution No. 19-194, CD1, which authorized the Mayor or the Mayor's designee to enter into a development agreement ("Development Agreement") between the City and EAH for the development of the Project; and

WHEREAS, the Development Agreement approved by the City Council in Resolution No. 19-194, CD1, included (i) a draft form of Ground Lease which was intended to be finalized at a later date, subject to the final approval of the City Council;



No.	23 - 58	
		•

and (ii) authorization for the Mayor or the Mayor's designee to make minor modifications to the Development Agreement so long as such modifications did not substantially deviate from EAH's proposal; and

WHEREAS, EAH assigned its interest and rights in and to the Development Agreement to an affiliated entity, Halewiliko Highlands LP, a Hawaii limited partnership ("HH LP", and together with EAH, the "Developer"), which is the proposed lessee under the Ground Lease; and

WHEREAS, the Developer has obtained entitlements for the Project from the City Council under the Hawaii Revised Statutes ("HRS") 201H program pursuant to Resolution No. 20-302, CD1, adopted on December 9, 2020 ("Entitlements"), and has also obtained commitments to finance the development and construction of the Project from institutional lenders, low-income housing tax credit investors, the State of Hawaii and a grant award of Affordable Housing Funds from the City (together, the "Financing Commitments"); and

WHEREAS, all conditions precedent to the execution of a Ground Lease, as set forth in the Development Agreement, are or will be satisfied by the Developer prior to the execution of the Lease, except for obtaining City Council's approval of the Ground Lease by resolution, which is the final remaining condition precedent to the execution of the Ground Lease; and

WHEREAS, the City and the Developer have now negotiated a final Ground Lease incorporating changes based on the Entitlements, the Financing Commitments, and final Project plans; and

WHEREAS, the Developer seeks a 75-year Lease term upon mutual agreement with the City to comply with the affordability requirements of the Financing Commitments; and

WHEREAS, all of the affordable rental units will be rented at rates at or below 60% of the U.S. Department of Housing and Urban Development ("HUD") area median income levels for the City, with 88 of 139 of the affordable rental units rented at 50% of the HUD area median income level; and

WHEREAS, the Developer will pay the City a nominal lease rent (\$1.00 per year) during the Lease term; and

WHEREAS, the City and Developer are seeking the City Council's approval and authorization to enter into the Ground Lease with HH LP pursuant to the Revised



Ordinances of Honolulu 2021, Section 38-3.3(g), which recodified ROH Section 28-3.4(g); and

WHEREAS, the City Council, having duly considered all of the submittals and testimony, desires to accept and approve the terms of the Lease, as set forth in the form attached hereto as Exhibit "A"; now, therefore,

BE IT RESOLVED by the City Council that:

- 1. The Ground Lease is approved substantially in the form attached hereto as Exhibit "A"; and
- 2. The Mayor or the Mayor's designee is hereby authorized to:
 - a. Execute the Ground Lease in substantially the form attached hereto as Exhibit "A", with any amendments recommended by the Department of the Corporation Counsel as necessary or desirable; and
 - b. Execute any incidental or related agreements or documents necessary to carry out the transaction described above as recommended by the Department of the Corporation Counsel as necessary or desirable, as long as such incidental or related documents do not increase either directly or indirectly the financial obligation of the City; and
- 3. Monies received from the payment of the lease rent shall be deposited in the General Fund.



No.	2	3	-	5	8	

BE IT FINALLY RESOLVED that copies of this resolution be transmitted to the Mayor; the Director of Department of Budget and Fiscal Services; the Director of Department of Land Management; and Halewiliko Highlands LP, whose address is 1001 Bishop Street, Suite 2880, Honolulu, Hawaii 96813.

	INTRODUCED BY: (b)
DATE OF INTRODUCTION:	
APR 18 2023	
Honolulu. Hawaii	Councilmembers

Halewiliko Highlands

LEASE

between

CITY AND COUNTY OF HONOLULU, a municipal corporation of the State of Hawai'i

"Lessor"

and

Halewiliko Highlands LP, a Hawaii limited partnership

"Lessee"

TABLE OF CONTENTS

ARTICLE 1:	DEFINITIONS	1
ARTICLE 2:	PREMISES	12
2.1	Premises	
2.2	Acceptance in Existing Condition	
2.3	Release of Lessor	
4 D T 101 E 0		
	TERM	
3.1	Term	
3.2 3.3	Delivery of PossessionNo Option to Extend; No Renewal	
3.3	No Option to Extend; No Renewal	14
ARTICLE 4:	RENT	
4.1	Fixed Rent	
4.2	No Offsets	
4.3	Payment - Generally. [RESERVED]	14
ARTICLE 5:	ADDITIONAL PAYMENTS BY LESSEE; REAL ESTATE TAXES	14
5.1	Net Lease	
5.2	Real Estate Taxes	
5.3	Assessments in Installments	
5.4	BID Decisions	
5.5	Direct Payment by Lessor	
5.6	Utilities	15
5.7	Excise Tax	16
5.8	Conveyance Tax	16
5.9	Taxes on Lessee's Business and Personal Property	16
5.10	Tax Exemptions	16
5.11	Lessor Expenses	16
ARTICI F 6:	USE	17
6.1	Permitted Use	
6.2	Prohibited Uses	
6.3	Exclusive Ownership of Improvements and Control of	
	Premises	18
6.4	Compliance with Laws	18
6.5	Copies of Notices	19
6.6	Entitlements	
6.7	Illegal Activities.	
6.8	Public Accommodations Laws	19
6.9	Use of Public Buildings by Blind or Visually Handicapped	
	Persons	
6.10	Project Name	20
ARTICLE 7:	MAINTENANCE, REPAIR AND CAPITAL IMPROVEMENTS	20
7.1	Obligation to Maintain	
7.2	Capital Improvement Obligations	
ADTICI E 0.	CONSTRUCTION	20
8.1	General	_
8.1 8.2	Commencement of Construction	
0.2	Commencement of Construction	∠∪

8.3	Manner of Construction	
8.4	Permits	
8.5	Applications	
8.6	Bond	
8.7	Completion.	
8.8	Construction Insurance	
8.9	Ownership	
8.10	Inspection	21
8.11	Lessee's Covenant	22
8.12	Title Encumbrances	22
8.13	Lease Termination	22
ARTICLE 9: H	IAZARDOUS SUBSTANCES	23
9.1	Restrictions	23
9.2	Compliance; Clean-Up	
9.3	Surrender Obligations	
9.4	Copies of Environmental Reports	24
9.5	Survival	
9.6	Discharges before Commencement	
ADTICI E 10.	INDEMNIFICATION; LIABILITY OF LESSOR	24
10.1	Obligations	
10.1	Liability of Lessor	
10.2	Indemnification Procedures	
10.3	indennincation Procedures	Z
	RIGHT OF CONTEST	
11.1	Lessee's Right; Contest Conditions	
11.2	Lessor Obligations and Protections.	
11.3	Miscellaneous	
11.4	Contest Security	27
ARTICI F 12.	INSURANCE	27
12.1	Lessee's Insurance	
12.2	Nature of Insurance Program	
12.3	Policy Requirements and Endorsements	
12.4	Waiver of Certain Claims	
12.5	Waiver of Subrogation.	
12.6	Evidence of Insurance.	
12.7	Deductibles, Retentions and Co-Insurance	
12.7 12.8	Failure to Maintain Insurance.	
12.9	Additional Insurance	
12.10	No Representation	31
	LOSSES AND LOSS PROCEEDS	
13.1	Notice	
13.2	Effect of Casualty	
13.3	Adjustment of Claims; Use of Property Insurance Proceeds	
13.4	Substantial Condemnation	
13.5	Insubstantial Condemnation	
13.6	Temporary Condemnation	
13.7	Immaterial Loss	33
13.8	Surrender	33
ADTIOL E 44	LESCORIS RESERVED RICHTS	20

14.1	Inspections	33
14.2	Other Entries by Lessor.	
14.3	Water, Oil, Gas and Mineral Rights	34
14.4	Easements	34
14.5	Development	
14.6	No Light or Air Easement	
14.7	General	35
ARTICLE 15	: LESSOR'S TRANSFERS	35
15.1	Transfer of Lessor's Interest	35
15.2	Release of Lessor	
15.3	No Right of First Refusal/No Option to Purchase	
ARTICLE 16	: REGULATORY PROVISIONS	36
16.1	Reserved	
ARTICLE 17	: LESSEE'S TRANSFERS	36
17.1	Lessee's Transfer Right	
17.2	No Partial Transfers	
17.3	Notice of Transfer	
17.4	Expenses	
ARTICI F 18	: SUBLEASES	37
18.1	Residential Tenants.	
18.2	Commercial Tenants. [RESERVED]	
18.3	Recordkeeping	
ARTICLE 19	: LEASEHOLD MORTGAGES	41
19.1	Leasehold Mortgage	
19.2	Protection of Leasehold Mortgagee	
ARTICLE 20	: EQUIPMENT LIENS	45
20.1	Lessee's Rights	
20.2	Required Provisions for Equipment Liens	
ARTICLE 21:	: QUIET ENJOYMENT	45
ADTICLE 22	. EVENTS OF DEFAULT, DEMEDIES	45
	: EVENTS OF DEFAULT; REMEDIES Definition of "Event of Default"	
22.1	Remedies	45
22.2	Proceeds of Reletting	
22.3 22.4	Lessee's Late Payments; Late Charges	
22. 4 22.5	Lesse's Late Payments, Late Charges	
22.6	Holding Over	
22.7	Waivers: Jury Trial, Redemption	
22.7	Accord and Satisfaction; Partial Payments	
22.9	Lessor's Default.	
22.10		
ARTICI F 23	: END OF TERM	ΔΟ
ARTICLE 24	: NOTICES	50
ADTICLE OF	ADDITIONAL DELIVEDIES, THIRD DARTIES	50

25.1	Estoppel Certificates	50
25.2	Further Assurances	
25.3	Memorandum of Lease	50
25.4	Modification	50
25.5	Lessor's Right to Amend	
25.6	Successors and Assigns	
ADTICI E 26.	CULTURAL AND ARCHEOLOGICAL	E1
26.1	Native Hawaiian Rights	
26.1 26.2	Human Remains; Artifacts; Historical Items	
20.2	numan Remains; Artifacts; historical items	
ARTICLE 27:	MISCELLANEOUS	
27.1	Due Authorization and Execution	52
27.2	Costs and Expenses; Legal Costs	52
27.3	No Consequential Damages	52
27.4	No Waiver by Silence	
27.5	Performance Under Protest	53
27.6	Survival	53
27.7	Unavoidable Delay	53
27.8	Broker	53
27.9	Service of Process	53
27.10	Sexual Harassment Policy	53
27.11	Non-Discrimination Policy	53
27.12	Neither Party Agent, Joint Venturer or Partner of the Other	
ARTICI E 28:	INTERPRETATION, EXECUTION, AND APPLICATION OF	
ARTIOLL 20.		
	IFASE	54
28 1	LEASE	
28.1 28.2	Captions	54
28.2	CaptionsCounterparts	54 54
28.2 28.3	Captions Counterparts Delivery of Drafts	54 54 54
28.2 28.3 28.4	Captions Counterparts Delivery of Drafts Entire Agreement	54 54 54
28.2 28.3 28.4 28.5	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue	54 54 54 54
28.2 28.3 28.4 28.5 28.6	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue Partial Invalidity	54 54 54 54 54
28.2 28.3 28.4 28.5 28.6 28.7	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue Partial Invalidity Principles of Interpretation	54 54 54 54 54 54 54
28.2 28.3 28.4 28.5 28.6 28.7 28.8	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue Partial Invalidity Principles of Interpretation Time of the Essence	54 54 54 54 54 54 54 55
28.2 28.3 28.4 28.5 28.6 28.7 28.8 28.9	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue Partial Invalidity Principles of Interpretation Time of the Essence Computation of Deadlines	54 54 54 54 54 54 54 55
28.2 28.3 28.4 28.5 28.6 28.7 28.8	Captions Counterparts Delivery of Drafts Entire Agreement Governing Law and Venue Partial Invalidity Principles of Interpretation Time of the Essence	54 54 54 54 54 54 55 55

Exhibit A	Legal Description and Encumbrances
Exhibit B	Additional Permitted Exceptions
Exhibit C	Form of Income Certification
Exhibit D	Form of Certificate of Continuing Program Compliance
Exhibit E	Notice Addresses (Including Required Copy Recipients)
Exhibit F	Service of Process

٧

LEASE

This **LEASE** (this "**Lease**") is made and entered into as of _______, 20____ (the "**Commencement Date**"), between the **CITY AND COUNTY OF HONOLULU**, a municipal corporation of the State of Hawai'i ("**Lessor**"), and **HALEWILIKO HIGHLANDS LP**, a Hawaii limited partnership ("**Lessee**").

For good and valuable consideration, the sufficiency of which is hereby acknowledged, Lessor hereby leases to Lessee, and Lessee hereby accepts and leases from Lessor, upon the terms and conditions set forth in this Lease and all Exhibits attached hereto, the Premises defined in <u>Section 2.1</u> below.

ARTICLE 1: DEFINITIONS

The following definitions apply in this Lease:

- "Additional Rent" has the meaning set forth in Section 5.1 of this Lease.
- "Affiliate" of any specified Person means any other Person Controlling or Controlled by or under common Control with such specified Person. "Affiliated" shall have the correlative meaning.
 - "Affordable Rent" has the meaning set forth in Section 18.1.3 of this Lease.
- "Affordable Rental Housing Area" means the gross square foot area of the entire Premises, excluding the Community Center Area.
- "Application" means any agreement, application, certificate, document, or submission (or amendment of any of the foregoing): (a) necessary or appropriate for any Construction this Lease allows, including any application for any building permit, certificate of occupancy, utility service or hookup, easement, covenant, condition, restriction, subdivision, or such other instrument as Lessee may from time to time reasonably request for such Construction; (b) to allow Lessee to obtain any abatement, deferral, or other benefit otherwise available for Real Estate Taxes; (c) to enable Lessee from time to time to seek any Approval or to use and operate the Premises in accordance with this Lease; or (d) otherwise reasonably necessary and appropriate to permit Lessee to realize the benefits of the Premises under this Lease.
- "Approvals" means any and all licenses, permits (including building, demolition, alteration, use, and special permits), approvals, consents, certificates (including certificate(s) of occupancy), rulings, variances, authorizations, or amendments to any of the foregoing as shall be necessary or appropriate under any Law to commence, perform, or complete any Construction, use, occupancy, maintenance, or operation of the Premises.
- "Bankruptcy Law" means Title 11, United States Code, and any other or successor state or federal statute relating to assignment for the benefit of creditors, appointment of a receiver or trustee, bankruptcy, composition, insolvency, moratorium, reorganization, or similar matters.
- "Bankruptcy Proceeding" means any proceeding, whether voluntary or involuntary, under any Bankruptcy Law.

"Building Equipment" means all fixtures incorporated in the Premises owned by Lessor or Lessee and used, useful, or necessary to operate the Improvements (including boilers; compactors; compressors; conduits; ducts; elevators; engines; equipment; escalators; fittings; heating, ventilating and air conditioning systems; utility systems; machinery; and pipes) as opposed to operating any business in the Improvements.

"Bureau" means the Bureau of Conveyances of the State of Hawaii.

"Business Day" means all days except for Saturdays, Sundays and legal holidays observed by the State, County or the federal government. If a due date determined under this Agreement falls on a Saturday, Sunday or an official State, County or federal holiday, such due date will be deemed to be the next Business Day.

"Casualty" means any damage or destruction of any kind or nature, ordinary or extraordinary, foreseen or unforeseen, affecting any or all Improvements, whether or not insured or insurable.

"Casualty Termination" means a termination of this Lease because of a Substantial Casualty, when and as this Lease expressly allows such a termination.

"Certifying Party" has the meaning set forth in Section 25.1 of this Lease.

"CFR" means the United States Code of Federal Regulations, as may be amended from time to time.

"Commencement Date" has the meaning set forth in the opening paragraph of this Lease.

"Community Center" has the meaning set forth in Section 6.1.4 of this Lease.

"Community Center Area" means the gross square footage underlying the Community Center located on the first floor of the Premises.

"Completion of the Project" means the date when the applicable Government agency issues a temporary or permanent "certificate of occupancy" for the Project.

"Compliance Period" means the fifteen (15) year tax credit compliance period as described in Section 42(i)(1) of the Internal Revenue Code, as amended.

"Condemnation" means any temporary or permanent taking of (or of the right to use or occupy) all or any portion of the Premises by condemnation, eminent domain or any similar proceeding.

"Condemnation Award" means any award(s) paid or payable (whether or not in a separate award) to either party or its mortgagee after the Commencement Date because of or as compensation for any Condemnation, including: (a) any award made for any improvements that are the subject of the Condemnation, (b) the full amount paid or payable by the condemning authority for the estate that is the subject of the Condemnation, as determined in Condemnation, (c) any interest on such award, and (d) any other sums payable on account of such Condemnation, including for any prepayment premium under any mortgage.

"Condemnation Effective Date" means, for any Condemnation, the first date when the condemning authority has acquired title to, or possession of, any portion of the Premises subject to the Condemnation.

"Construction" means any alteration, construction, demolition, development, expansion, reconstruction, redevelopment, repair, Restoration, or other work affecting any Improvements, including new construction and replacements. Construction consists of Minor Construction and Major Construction.

"Consumer Price Index" means the index published by the United States Department of Labor, Bureau of Labor Statistics, and now known as the Consumer Price Index for All Urban Consumers (1982-84 = 100), U.S. City average, All Items (or such comparable index as may be utilized in substitution for or as the successor to the stated index). If such index is not published by the United States Bureau of Labor Statistics, or successor agency thereof, at any time during the Term, then the most closely comparable statistics on the purchasing power of the consumer dollar as published by a responsible financial authority and selected by Lessor shall be utilized in lieu of such index.

"Contest" has the meaning set forth in Section 11.1 of this Lease.

"Contest Conditions" has the meaning set forth in Section 11.1 of this Lease.

"Contest Security" has the meaning set forth in Section 11.1.1 of this Lease.

"Control" means possession of the power to direct or cause the direction of the management and policies of such Person, whether by ownership of Equity Interests or by written management authority.

"County" means the City and County of Honolulu.

"Default" means any Monetary Default or Nonmonetary Default.

"**Default Interest**" means interest at an annual rate equal to the lesser of: (a) the Prime Rate plus four percent (4%) per annum; or (b) the Usury Limit.

"**Depository**" means a bank or trust company mutually designated by Lessor and Lessee, which is qualified under the Laws of the State and having its principal office in Honolulu, Hawai'i.

"Development Agreement" means that certain Development Agreement dated November 6, 2019, by and between Lessor and EAH Inc., a California nonprofit public benefit corporation ("EAH"), approved by the Honolulu City Council via Resolution No. 19-194, CD1, the interest of EAH having been assigned to Lessee by that certain unrecorded Assignment and Assumption of Developer's Rights and Obligations under Development Agreement; and Consent, by and between EAH and Lessee, with the consent of Lessor.

"Easements" has the meaning set forth in Section 14.4.1 of this Lease.

"Environmental Law(s)" means any Law regarding the following at, in, under, above, or upon the Premises: (a) air, environmental, ground water, or soil conditions; or (b) clean-up, disposal, generation, storage, release, transportation, or use of, or liability or standards of conduct concerning, Hazardous Substances.

"Equipment Lien" means any security interest, financing lease, personal property lien, conditional sales agreement, chattel mortgage, security agreement, title retention arrangement or any similar arrangement (including any related financing statement) for Lessee's acquisition or leasing of any Financed FF&E used in the Premises that is leased, purchased under conditional sale or installment sale arrangements, encumbered by a security interest, or used under a license, provided that each Equipment Lien encumbers or otherwise relates only to the Financed FF&E for which such secured party provides

bona fide purchase-money financing or a bona fide equipment lease, after the Commencement Date. A Leasehold Mortgage is not an Equipment Lien.

"Equity Interest" means all or any part of any equity or ownership interest(s) (whether stock, partnership interest, beneficial interest in a trust, membership interest, or other interest of an ownership or equity nature) in any Person.

"Event of Default" means those any one of those certain events of default described in <u>Section 22.1</u> of the Lease, including without limitation a Monetary Default and a Nonmonetary Default.

"Excise Tax" has the meaning set forth in Section 5.7 of this Lease.

"Expiration Date" means the date when this Lease terminates or expires in accordance with its terms, whether on the Scheduled Expiration Date, by Lessor's exercise of remedies for an Event of Default, or otherwise.

"Federal and State Affordable Housing Requirements" means any and all federal and state statutory, executive order and regulatory requirements applicable to any residential units on the Premises as such requirements now exist or as they may be amended from time to time, including, but not limited to, the Housing Act of 1937 as amended (42 U.S.C. §1437 et seq.), or its successor, the HOME Program, all requirements of law relating to any multi-family revenue bonds authorized in connection with the financing of Lessee's acquisition of the Lease, the Tax Credit Requirements (as defined below), and any other housing assistance program funded, insured, or operated by HUD or the State.

"Fee Estate" means Lessor's fee estate in the Premises, including Lessor's reversionary interest in the Premises after the Expiration Date.

"FF&E" means all movable furniture, furnishings, equipment, and personal property of Lessee or anyone claiming through Lessee (excluding Building Equipment) that may be removed without material damage to the Premises and without adversely affecting: (a) the structural integrity of the Premises; (b) any electrical, plumbing, mechanical, or other system in the Premises; (c) the present or future operation of any such system; or (d) the present or future provision of any utility service to the Premises. FF&E includes items such as factory equipment, furniture, movable equipment, telephone, telecommunications and facsimile transmission equipment, point of sale equipment, televisions, radios, network racks, and computer systems and peripherals.

"Financed FF&E" means any FF&E subject to an Equipment Lien in favor of a lessor or lender that: (a) is not an Affiliate of Lessee, and (b) actually provides *bona fide* financing or a *bona fide* equipment lease after the Commencement Date for Lessee's acquisition or use of such FF&E.

"Fixed Rent" has the meaning set forth in Section 4.1 of this Lease.

"Fixed Term" has the meaning set forth in Section 4.1 of this Lease.

"Ford Report" means (i) Environmental Hazard Evaluation dated September 2, 2022, and prepared by Ford & Associates, Inc.; and (ii) Construction Environmental Hazard Management Plan dated September 2, 2022, prepared by Ford & Associates, Inc.

"Government" means each and every governmental agency, authority, bureau, department, quasi-governmental body, or other entity or instrumentality having or claiming jurisdiction over the Premises (or any activity this Lease allows), including the United States government, the State and County

governments and their subdivisions and municipalities, and all other applicable governmental agencies, authorities, and divisions thereof.

"Hazardous Substances" includes flammable substances, explosives, radioactive materials, asbestos, asbestos-containing materials, polychlorinated biphenyls, chemicals known to cause cancer or reproductive toxicity, pollutants, contaminants, hazardous wastes, medical wastes, toxic substances or related materials, explosives, petroleum and petroleum products, and any "hazardous" or "toxic" material, substance or waste that is defined by those or similar terms or is regulated as such under any Law, including any material, substance or waste that is: (a) defined as a "hazardous substance" under Section 311 of the Water Pollution Control Act (33 U.S.C. §1317), as amended; (b) defined as a "hazardous waste" under Section 1004 of the Resource Conservation and Recovery Act of 1976, 42 U.S.C. §6901, et seq., as amended; (c) defined as a "hazardous substance" or "hazardous waste" under Section 101 of the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended by the Superfund Reauthorization Act of 1986, 42 U.S.C. §9601 et seq. or any so-called "superfund" or "superlien" law; (d) defined as a "pollutant" or "contaminant" under 42 U.S.C.A. §9601(33); (e) defined as "hazardous waste" under 40 C.F.R. Part 260; (f) defined as a "hazardous chemical" under 29 C.F.R. Part 1910; or (g) subject to any other Law regulating, relating to or imposing obligations, liability or standards of conduct concerning protection of human health, plant life, animal life, natural resources, property or the enjoyment of life or property free from the presence in the environment of any solid, liquid, gas, odor or any form of energy from whatever source. The term "Hazardous Substances" for purposes of this Lease shall also include any mold, fungus or spores, whether or not the same is defined, listed, or otherwise classified as a "hazardous substance" under any Environmental Laws, if such mold, fungus or spores may pose a risk to human health or the environment or negatively impact the value of the Premises.

"Hazardous Substances Discharge" means any deposit, discharge, generation, release, or spill of Hazardous Substances that occurs at or from the Premises, or into the Land, or that arises at any time from the use, occupancy, or operation of the Premises or any activities conducted therein or any adjacent or nearby real property, or resulting from seepage, leakage, or other transmission of Hazardous Substances from other real property to the Land, whether or not caused by a party to this Lease and whether occurring before or after the Commencement Date.

"Historic Items" has the meaning set forth in Section 26.2.1 of this Lease.

"Housing Act" means the Housing Act of 1937, as may be amended from time to time.

"HRS" means the Hawaii Revised Statutes, as may be amended from time to time.

"HUD" means the United States Department of Housing and Urban Development.

"Immaterial Loss" means a Casualty or Condemnation where the Loss Proceeds or the Condemnation Award, respectively, is less than \$500,000.

"Improvements" means all buildings, structures, and other improvements and appurtenances located on the Land.

"Indemnify" means, where this Lease states that any Indemnitor shall "Indemnify" any Indemnitee from, against, or for a particular matter (the "Indemnified Risk"), that the Indemnitor shall indemnify the Indemnitee, defend and hold the Indemnitee harmless from and against any and all loss, cost, claims, liability, penalties, judgments, damages, and other injury, detriment, or expense (including Legal Costs, interest and penalties) that the Indemnitee suffers or incurs: (a) from, as a result of, or on account of the Indemnified Risk; or (b) in enforcing the Indemnitor's indemnity. Indemnitor's counsel shall be subject to

Indemnitee's approval, not to be unreasonably withheld. Any counsel satisfactory to Indemnitor's insurance carrier shall be automatically deemed satisfactory.

"Indemnitee" means any party entitled to be Indemnified under this Lease and its agents, directors, employees, Equity Interest holders, mortgagees, and officers.

"Indemnitor" means a party that agrees to Indemnify any other Person.

"Institutional Lender" means a state or federally chartered savings bank, savings and loan association, credit union, commercial bank or trust company or a foreign banking institution (in each case whether acting individually or in a fiduciary or representative (such as an agency) capacity); an insurance company organized and existing under the Laws of the United States or any state thereof or a foreign insurance company (in each case whether acting individually or in a fiduciary or representative (such as an agency) capacity); an institutional investor such as a publicly held real estate investment trust, an entity that qualifies as a "REMIC" under the Internal Revenue Code, a syndicator or other entity in the business of low-income housing tax credit investment, or other public or private investment entity (in each case whether acting as principal or agent) which at the date hereof or in the future is involved in the business of investing in real estate assets, security interests or private activity bonds; a brokerage or investment banking organization (in each case whether acting individually or in a fiduciary or representative (such as an agency) capacity); an employees' welfare, benefit, pension or retirement fund; any entity the liabilities of which are insured by a governmental agency, or any combination of Institutional Lenders; or a government-sponsored enterprise, such as the Federal National Mortgage Association (Fannie Mae) or the Federal Home Loan Mortgage Corporation (Freddie Mac); provided, however, that each of the entities above shall gualify as an Institutional Lender only if (at the time it becomes an Institutional Lender) it shall not be an Affiliate of Lessee.

"Insubstantial Condemnation" means any Condemnation except a Substantial Condemnation, a Temporary Condemnation, or an Immaterial Loss.

"Land" has the meaning set forth in Section 2.1 of this Lease.

"Land Court" means the Office of the Assistant Registrar of the Land Court of the State of Hawaii.

"Law" or "Laws" means all laws, ordinances, requirements, orders, proclamations, directives, rules, and regulations of any Government affecting the Premises, this Lease, or any Construction in any way, including any use, maintenance, taxation, operation, or occupancy of, or environmental conditions affecting, the Premises, or relating to any Real Estate Taxes, or otherwise relating to this Lease or any party's rights and remedies under this Lease, or any Transfer of any of the foregoing, whether in force at the Commencement Date or passed, enacted, or imposed at some later time, and as may be amended from time to time.

"Lease" means this Lease, together with the exhibits attached hereto, as the same may be amended from time to time.

"Lease-Related Documents" has the meaning set forth in Section 27.1 of this Lease.

"Leasehold Estate" means Lessee's leasehold estate in and to the Land, and all Improvements constructed by Lessee thereon, and all of Lessee's rights and privileges under this Lease, upon and subject to all the terms and conditions of this Lease, and any direct or indirect interest in such leasehold estate.

"Leasehold Mortgage" means a mortgage or other security instrument (a) that encumbers the Leasehold Estate or any interest in the Leasehold Estate; (b) a copy of which is promptly after execution

delivered to Lessor and recorded in the Bureau, with a certification by Leasehold Mortgagee that the copy is accurate and stating Leasehold Mortgagee's name and Notice address; and (c) that is held by a Leasehold Mortgagee that is an Institutional Lender, subject to the jurisdiction of the courts of the State of Hawai'i.

"Leasehold Mortgagee" means the holder of any Leasehold Mortgage and its successors and assigns.

"Legal Costs" of any Person means all reasonable costs and expenses such Person incurs in any legal proceeding (or other matter for which such Person is entitled to be reimbursed for its Legal Costs), including reasonable attorneys' fees, court costs, and expenses, and in or as a result of any Bankruptcy Proceeding.

"Lessee" means Halewiliko Highlands LP, a Hawaii limited partnership, and its successors and permitted assigns.

"Lessor" initially means the City and County of Honolulu, a municipal corporation of the State of Hawaii. After every transfer of the Fee Estate made in accordance with <u>Article 15</u>, "Lessor" means only the owner(s) of the Fee Estate at the time in question.

"Liability Insurance" means commercial general liability insurance against claims for personal injury, death, or property damage occurring upon, in, or about the Premises or adjoining sidewalks, providing coverage limits (and subject to increases) as provided in <u>Article 12</u>.

"Limited Partner" means, collectively, U.S. Bancorp Community Development Corporation, a Minnesota corporation, and USB Hawaii State Investor II, LLC, a Missouri limited liability company, and their respective successors and assigns.

"Limited Partnership Agreement" means that Amended and Restated Agreement of Limited Partnership of Halewiliko Highlands LP by and between the Limited Partner and HKI Halewiliko LLC dated on or about the date hereof, as may be amended from time to time.

"Loss" means any Casualty or Condemnation.

"Loss Proceeds" means Property Insurance Proceeds and/or Condemnation Award(s).

"LUO" means the County's Land Use Ordinance, codified in Chapter 21 of the ROH.

"Major Construction" means any Construction that is reasonably anticipated to cost in excess of \$1,000,000 (which amount shall be increased in proportion to the percentage increase, if any, in the Consumer Price Index since the Commencement Date).

"Market Value" of the Fee Estate or the Leasehold Estate means, as of any date of determination, the present fair market value of such estate (including the fair market value of the rights of the holder of such estate in and to any Improvements) as of such date, considered: (a) as if no Loss had occurred; (b) without adjusting for any expectation of any Loss; (c) as if the Leasehold Estate had not been terminated; (d) taking into account the benefits and burdens of this Lease (including, without limitation, all cash flows and revenues, including developer fees, accruing to or reasonably anticipated to accrue to the holder of the Leasehold Estate), the remaining Term, all Permitted Exceptions, and all other matters affecting such estate and its valuation; and (e) discounting to present value all the obligations and benefits associated with such estate (including, in the case of the Fee Estate, the Rent and Lessor's reversionary interest). The Market Value shall be determined as if the Term were to continue until the Scheduled

Expiration Date, and shall be determined independently of, and without regard to, any valuation established in a Condemnation.

"Minor Construction" means any Construction that Lessee elects in its discretion, or this Lease requires Lessee, to undertake from time to time, except Major Construction.

"Modification" means any abandonment, amendment, cancellation, discharge, extension, modification, rejection, renewal, replacement, restatement, substitution, supplement, surrender, termination, or waiver of a specified agreement or document, or of any of its terms or provisions, or the acceptance of any cancellation, rejection, surrender, or termination of such agreement, document, or terms.

"Modify" means agree to, cause, make, or permit any Modification.

"Monetary Default" means Lessee's failure to pay any Rent or other money (including Real Estate Taxes and insurance premiums) when and as required under this Lease (subject to Lessee's right to cure during the applicable cure period after receipt of Notice as set forth in Article 22 of this Lease) or, subject to Section 28.4, the Development Agreement (after the giving of written notice and failure to cure within the applicable cure period as set forth in Section 13.1 of the Development Agreement).

"Nonmonetary Default" means Lessee's: (a) failure to comply with any affirmative or negative covenant or obligation in this Lease (subject to Lessee's right to cure during the applicable cure period after receipt of Notice as set forth in Article 22 of this Lease) or in the Development Agreement in any material respect after the giving of written notice and failure to cure within the applicable cure period as set forth in Section 13.1 of the Development Agreement and subject to Section 28.4), except a Monetary Default; or (b) breach by Lessee of any representation or warranty in any material respect (as of the date made or deemed made) made in this Lease (subject to Lessee's right to cure during the applicable cure period after the receipt of Notice as set forth in Article 22 of this Lease).

"**Notice**" means any consent, demand, designation, election, notice, or request relating to this Lease, including any Notice of Default. Notices shall be deemed effective only when in writing and delivered in accordance with <u>Article 24</u>.

"Notice of Default" means any Notice claiming or giving Notice of a Default or alleged Default.

"Notice to Proceed" means that certain notice which permits Lessee to commence with the Construction of the Project in accordance with <u>Article 7</u> of the Development Agreement.

"Notify" means give a Notice.

"Permitted Exceptions" means: (a) the recorded title exceptions affecting the Fee Estate and prior to this Lease as of the Commencement Date, listed as exceptions in Lessee's leasehold policy of title insurance for this Lease; (b) intentionally omitted; (c) any title exceptions (including new Subleases) caused by Lessee's acts or omissions, consented to or requested by Lessee, or resulting from Lessee's Default; (d) any Application made at Lessee's request; (e) any title exceptions resulting from the exercise of the rights reserved to Lessor in this Lease; (f) any grant of easement for access or utility purposes as may be requested by any Government public utility company in connection with the development, use and operation of the Project, and (f) the additional matters, if any, listed in Exhibit B attached hereto.

"Permitted Uses" means the uses described in Section 6.1 of this Lease.

"Person" means any association, corporation, Government, individual, joint venture, joint-stock company, limited liability company, partnership, general or limited partnership, trust, unincorporated organization, or other entity of any kind.

"Personal Default" means any Nonmonetary Default that is not reasonably susceptible of cure by a Leasehold Mortgagee, such as (to the extent, if any, that any of the following may actually constitute a default) a bankruptcy proceeding affecting Lessee; a prohibited transfer; failure to deliver financial information within Lessee's control; failure to remove or retain any particular officer, employee, or director of Lessee; failure to comply with restrictions on competition or other activities that relate to other sites owned or leased by Lessee; and any other Nonmonetary Default that by its nature relates only to, or can reasonably be performed only by, Lessee.

"Pre-Existing Hazardous Substances Condition" means any Hazardous Substances Discharge or other environmental condition that existed or that occurred prior to the Commencement Date arising out of the Lessor's use of the Premises, including for parking, storage and related purposes, but excluding the existing soil contamination from the former Aiea Sugar Mill and Refinery that was partially located on the Premises and is identified in the Ford Report.

"Premises" has the meaning set forth in Section 2.1 of this Lease.

"Prime Rate" means the prime rate or equivalent "base" or "reference" rate for corporate loans that is published in the Wall Street Journal as of the applicable date or, if such rate is no longer published, then a reasonably equivalent rate published by an authoritative third party mutually designated by Lessor and Lessee. Notwithstanding anything to the contrary in this paragraph, the Prime Rate shall never exceed the Usury Limit.

"Prohibited Lien" means any mechanic's, vendor's, laborer's, or material supplier's statutory lien or other similar lien arising from work, labor, services, equipment, or materials supplied, or claimed to have been supplied, to Lessee (or anyone claiming through Lessee), if such lien attaches to the Leasehold Estate or attaches (or may attach upon termination of this Lease) to the Fee Estate. An Equipment Lien is not a Prohibited Lien.

"Project" means that certain affordable rental housing project for seniors, together with a Community Center and other components, to be developed on the Land, and more particularly described in Article VI.

"Property Insurance Proceeds" means net proceeds (after reasonable costs of adjustment and collection, including Legal Costs) of property insurance, when and as received by Lessor, Lessee, Depository, or any mortgagee, excluding proceeds of Lessee's business interruption insurance in excess of Rent.

"Public Accommodations Laws" means all applicable Laws, including, without limitation, Title II and Title III of the Americans with Disabilities Act of 1990 (the "ADA"), the ADA Accessibility Guidelines promulgated by the Architectural and Transportation Barriers Compliance Board, the public accommodations title of the Civil Rights Act of 1964, 42 USC § 2000a et seq., the Architectural Barriers Act of 1968, 42 USC § 4151 et seq., as amended, Title V of the Rehabilitation Act of 1973, the Minimum Guidelines and Requirements for Accessible Design, 36 CFR Part 1190, and the Uniform Federal Accessibility Standards, and any similar Laws now or hereafter adopted, published or promulgated, as the same are now in effect or may be hereafter modified, amended or supplemented.

"Qualified Tenants" mean tenants that meet the age and income requirements for a Residential Rental Unit set forth in Section 6.1.1.

"Real Estate Taxes" means all general and special real estate taxes (including taxes on FF&E, sales taxes, use taxes, and the like), BID payments, assessments, municipal water and sewer rents, rates and charges, excises, levies, license and permit fees, fines, penalties and other governmental charges and any interest or costs with respect thereto, general and special, ordinary and extraordinary, foreseen and unforeseen, that before or during the Term and applicable to the Term, or any part of it, may be assessed, levied, imposed upon, or become due but only to the extent payable out of or in respect of, or charged with respect to or become a lien on, the Premises, or any FF&E, Building Equipment or other facility on the Premises and used in the operation thereof. "Real Estate Taxes" shall not, however, include any of the following, all of which Lessor shall pay before delinguent or payable only with a penalty: (a) any franchise, income, excess profits, estate, inheritance, succession, transfer, gift, corporation, business, capital levy, or profits tax, or license fee, of Lessor; (b) any item listed in this paragraph that is levied, assessed, or imposed against the Premises during the Term based on the recapture or reversal of any previous tax abatement or tax subsidy, or compensating for any previous tax deferral or reduced assessment or valuation, or correcting a miscalculation or misdetermination, relating to any period(s) before the Commencement Date; and (c) interest, penalties, and other charges for subparts (a) and (b) aforesaid. If at any time during the Term the method of taxation prevailing at the Commencement Date shall be altered so that any new tax, assessment, levy (including any municipal, state or federal levy), imposition, or charge, or any part thereof, shall be measured by or be based in whole or in part upon the Premises and imposed upon Lessor, then all such new taxes, assessments, levies, Real Estate Taxes, or charges, or the part thereof to the extent that they are so measured or based, shall be deemed to be included within the term "Real Estate Taxes".

"Rent" means the Fixed Rent, the Additional Rent, and any other amounts required to be paid by Lessee to Lessor under the provisions of this Lease.

"Requesting Party" has the meaning set forth in Section 25.1 of this Lease.

"Residential Rental Units" means those low income residential rental units subject to the Tax Credit Requirements that are designated for occupancy by persons age 62 and older who meet the income qualifications set forth herein, and shall not include the residential unit occupied by the resident staff person.

"Restoration" means, after a Loss, the alteration, clearing, rebuilding, reconstruction, repair, replacement, restoration, and safeguarding of the damaged or remaining Improvements, substantially consistent with their condition before the Loss, subject to such Construction as Lessee shall perform in conformity with this Lease, subject to any changes in Law that would limit the foregoing.

"Restoration Funds" means any Loss Proceeds (and deposits by Lessee) to be applied to Restoration.

"Restore" means accomplish a Restoration.

"RFP" means that certain Request for Proposals issued by the Lessor on August 2018 for the development of the Premises.

"ROH" means the Revised Ordinances of Honolulu, as may be amended from time to time.

"Scheduled Expiration Date" has the meaning set forth in Section 3.1 of this Lease.

"SHPD" has the meaning set forth in Section 26.2.2 of this Lease.

"State" means the State of Hawai'i.

"Sublease" means, for any portion of the Premises, any: (a) sublease; (b) agreement or arrangement (including a concession, license, management, or occupancy agreement) allowing any Person to occupy, use or possess; (c) sub-sublease or any further level of subletting; or (d) Modification or assignment of clause (a) through clause (c) above.

"Sublessee" means any Person entitled to occupy, use, or possess any premises under a Sublease.

"Substantial Casualty" means a Casualty that (a) occurs less than five (5) years before the end of the Term and renders the Improvements, in Lessee's reasonable judgment (with Leasehold Mortgagee's consent), not capable of being economically Restored, (b) occurs less than ten (10) years before the end of the Term and the cost of Restoration of the Improvements destroyed or damaged by such casualty equals or exceeds fifty percent (50%) of the replacement cost of the Improvements then situated on the Premises, or (c) pursuant to Law, prevents the Improvements from being Restored to substantially the same condition, and for the same use, as before the Casualty.

"Substantial Condemnation" means any Condemnation that takes the entire Premises or so much thereof that the remainder, in Lessee's reasonable judgment (with Leasehold Mortgagee's consent), is not capable of being Restored to an economically viable whole for the conduct of the Permitted Use specified in Section 6.1.

"Tax Credit Requirements" means any and all matters required by Section 42 of the Internal Revenue Code of 1986 (26 U.S.C.), as amended (and the cognate provisions of the laws of the State) and/or any applicable agreement or restrictions relating to the receipt of any federal or State low income housing tax credits in connection with Lessee's activities and obligations under this Lease, whether or not such requirement is explicitly stated in Section 42 or regulations thereunder (or the applicable provisions of the laws of the State and the regulations thereunder).

"**Temporary Condemnation**" means a Condemnation of the right to use or occupy all or part of the Premises for a temporary period of time.

"Term" has the meaning set forth in Section 3.1 of this Lease.

"Transfer" of any property means any of the following, whether by operation of law or otherwise, whether voluntary or involuntary, and whether direct or indirect: (a) any assignment, conveyance, grant, hypothecation, mortgage, pledge, sale, or other transfer, whether direct or indirect, of all or any part of such property, or of any legal, beneficial, or equitable interest or estate in such property or any part of it (including the grant of any easement, lien, or other encumbrance); or (b) any conversion, exchange, issuance, Modification, reallocation, sale, or other transfer of any Equity Interest(s) in the owner of such property by the holders of such Equity Interest(s). Except as provided below, a transaction affecting Equity Interests, as referred to in clause (b) above shall be deemed a Transfer by Lessee even though Lessee is not technically the transferor. A "Transfer" shall not, however, include any of the foregoing (provided that the other party to this Lease has received Notice thereof) relating to any Equity Interest: (i) that constitutes a mere change in form of ownership with no material change in beneficial ownership and constitutes a tax-free transaction under federal income tax law; (ii) to member(s) of the immediate family(ies) of the transferor(s) or trusts for their benefit; or (iii) to any Person that, as of the Commencement Date, holds an Equity Interest in the entity whose Equity Interest is being transferred or is an Affiliate of a Person who holds an Equity Interest in the entity whose Equity Interest is being transferred. Notwithstanding the foregoing, (a) any of the following shall not be considered a "Transfer" and shall not require Lessor's consent (provided that, in each instance, Lessor has received Notice thereof): (i) the exercise by the Limited Partner of its rights

pursuant to the Partnership Agreement to remove the general partner of Lessee upon the existence of an event of default by the general partner, after the giving of written notice and expiration of any applicable cure period, and appoint the Limited Partner or an Affiliate thereof as interim general partner of Lessee; (ii) the exercise by the Limited Partner of its right to enforce any obligation of the general partner to repurchase the Limited Partner's partnership interest in Lessee as set forth in the Partnership Agreement; and/or (iii) a transfer by the Limited Partner of its partnership interest in Lessee (including transfers within a limited partner's interest of Lessee) in accordance with the terms of the Partnership Agreement, and (b) any amendment to the Partnership Agreement to effectuate any of the permitted transfers in the preceding clause (a) of this sentence shall not require Lessor's prior written consent. In addition to the foregoing, the trading of an Equity Interest in any entity whose capital stock is listed on a nationally recognized stock exchange shall not constitute a Transfer.

"Unavoidable Delay" means delay in performing any obligation under this Lease (except payment of money) arising from or on account of any cause whatsoever beyond the obligor's reasonable control, despite such obligor's reasonable diligent efforts, including industry-wide strikes, labor troubles or other union activities (but only to the extent such actions affect similar premises at that time and do not result from an act or omission of the obligor), the obligor's inability to obtain required labor or materials after commercially reasonable efforts to do so, litigation (unless caused by the obligor) including, without limitation, injunctive or similar relief in connection with any litigation, Loss, accidents, Laws, governmental preemption, war, riots, terrorism, contamination by radioactivity or Hazardous Materials (unless caused by the obligor), fire, explosions, earthquakes, drought, tidal waves, floods, windstorms, natural disaster or other acts of God. Unavoidable Delay shall exclude delay caused by the obligor's financial condition, illiquidity, or insolvency. Any obligor claiming Unavoidable Delay shall Notify the obligee: (a) within ten (10) Business Days after such obligor knows of any such Unavoidable Delay; and (b) within five (5) Business Days after such Unavoidable Delay ceases to exist. To be effective, any such Notice must describe the Unavoidable Delay in reasonable detail. Where this Lease states that performance of any obligation is subject to Unavoidable Delay(s) or words of similar import, such Unavoidable Delay(s) shall extend the time for such performance only by the number of days by which such Unavoidable Delay(s) actually delayed such performance, plus a reasonable period of time thereafter to mobilize to commence performance.

"Unrelated Hazardous Substance Discharge" any Hazardous Substance Discharge migrating, entering or leaching onto, above, or beneath the Property at any time subsequent to the Commencement Date from adjacent or nearby property or properties over which Lessee has no control or in which Lessee has no ownership interest.

"Usury Limit" means the highest rate of interest, if any, that Law allows under the circumstances.

"Year" means a period of 365 days except in leap years, in which case the period is 366 days.

ARTICLE 2: PREMISES

Premises. At the Commencement Date, Lessor owns the following real property (collectively, the "**Premises**"): (a) the land described in **Exhibit A** attached hereto (the "**Land**"); (b) the Improvements existing thereon as of the Commencement Date; (c) the appurtenances and all the estate and rights of Lessor in and to the Land; and (d) all Building Equipment attached or appurtenant to any of the foregoing existing as of the Commencement Date. This Lease is subject to the encumbrances described in **Exhibit A**, the Permitted Exceptions, and any title exceptions resulting from the exercise of the rights reserved to Lessor in this Lease.

- 2.2 Acceptance in Existing Condition. Except as otherwise provided in the Development Agreement or in this Lease, (a) Lessee expressly acknowledges and agrees that Lessor has made no representations or warranties whatsoever, whether express, implied or statutory, with respect to the Premises or any portion thereof, and (b) that Lessor shall not be obligated to provide or pay for any work or services related to the Premises or the operation thereof. Lessee acknowledges that Lessee has inspected the Premises carefully, or has had the opportunity to inspect the Premises carefully, and except as otherwise provided in the Development Agreement or this Lease, accepts the Premises in "AS IS, WHERE IS AND WITH ALL FAULTS" condition without warranty, guaranty, liability, or representation whatsoever, express or implied, oral or written, on the part of Lessor, or any Person on behalf of Lessor, regarding the Premises or matters affecting the Premises, including the following:
- **2.2.1** Physical Condition. The physical condition of the Premises, including the quality, nature, adequacy and physical condition of (a) the Land, including any systems, facilities, access, and/or landscaping; (b) the air, soils, geology, topography, drainage, and groundwater, (c) the suitability of the Land for construction of any future Improvements or any activities or uses that Lessee may elect to conduct on the Land; or (d) the compaction, stability or composition, erosion or other condition of the soil or any fill or embankment on the Land for building or any other purpose;
- **2.2.2** <u>Improvements</u>. The quality, nature, adequacy and physical condition of any existing Improvements.
- **2.2.3** <u>Title.</u> The nature and extent of any recorded right of way, lease, possession, lien, encumbrance, license, reservation or other title condition of record as of the Commencement Date affecting the Premises including, without limitation, the existence of any easements, rights of ways or other rights across, to or in other properties that might burden and/or benefit the Premises; provided, however, that Lessor represents and warrants to Lessee that it owns fee simple title to the Land and that no other party has any right or option to purchase the Land or the Leasehold Estate therein as of the Commencement Date of this Lease.
- **2.2.4** <u>Compliance</u>. The development potential of the Premises and/or the zoning, land use, or other legal status of the Land or Improvements or compliance with any public or private restrictions on the use of the Land, as the same are in effect as of the Commencement Date or may be hereafter modified, amended, adopted, published, promulgated or supplemented, or the compliance of the Land or Improvements with any applicable Laws;
- **2.2.5** <u>Hazardous Substances</u>. The presence or removal of Hazardous Substances on, in, or under the Premises (as described in the Ford Report) but excluding any Pre-Existing Hazardous Substances Condition and any Unrelated Hazardous Substances Discharge;
- **2.2.6** <u>Economic Feasibility</u>. Economic conditions or projections, development potential, market data, or other aspects of the economic feasibility of the Premises and/or the business Lessee intends to conduct on the Premises;
- **2.2.7** <u>Utilities</u>. The availability, existence, quality, nature, adequacy and physical condition of utilities serving the Premises;
- **2.2.8** <u>Suitability</u>. The use, habitability, merchantability, fitness, suitability, value or adequacy of the Premises for any particular purpose (including, without limitation, the Permitted Use specified in Section 6.1);
- **2.2.9** <u>Boundaries</u>. The boundaries of the Premises, the location of any Improvements on the Land and/or the existence of any encroachments onto or from any adjacent lands;

- **2.2.10** Access. Access to the Premises, including from or through any particular route; and
- **2.2.11** Other Matters. Any matter whatsoever not referenced above that pertains to the Premises.
- **Release of Lessor**. Lessee, on behalf of itself, its agents, directors, employees, Equity Interest holders, mortgagees, and officers, hereby waives, releases and forever discharges Lessor and its agents, directors, employees and officers of and from any and all claims, actions, causes of actions, demands, rights, damages, costs, expenses or compensation whatsoever, direct or indirect, known or unknown, foreseen or unforeseen, which Lessee or any its agents, directors, employees, Equity Interest holders, mortgagees or officers now have or which may arise in the future on account of or in any way related to or in connection with any past, present or future aspect, feature, characteristic or condition of the Premises of the nature and type specified in Section 2.2.1 through Section 2.2.10; provided, however, that this release shall not cover, pertain to, or deem to release any claim of Lessee against Lessor for breach of this Lease or the Development Agreement or any claim under Article 9 or Section 10.2 of this Lease.

ARTICLE 3: TERM

3.1	<u>Ter</u>	<u>m</u> .	The	terms a	and pr	ovisions	of this	Leas	se sh	nall be effect	tive as	of the	Commenceme	nt D	ate.
The	term	of	this	Lease	(the	"Term")	shall	be	for	seventy-five	e (75)	years	commencing	on	the
Com	mence	me	nt Da	ite and	termin	ating at	11:59 p	o.m.	on _		_, 20	(the "	Scheduled Ex	pira	tion
Date	"), unle	ess	termi	nated s	ooner	in accord	dance	with	this I	Lease.					

- **3.2** <u>Delivery of Possession</u>. Lessor shall deliver the Premises to Lessee on the Commencement Date.
- **3.3 No Option to Extend; No Renewal**. Lessee shall have no right to extend or renew the Term. Upon expiration or earlier termination of this Lease, Lessor shall have no obligation to extend or renew this Lease, or to enter into a new lease with Lessee, unless otherwise mutually agreed to and as permitted by law.

ARTICLE 4:

- **4.1 Fixed Rent**. Lessee shall pay Lessor, without notice or demand, in lawful money of the United States of America, a net annual rental (the "**Fixed Rent**") of One and No/100 Dollars (\$1.00). Lessor acknowledges that on the Commencement Date, Lessee has paid to Lessor the amount of \$75.00, which represents payment of the Fixed Rent of \$1.00 per year that is due for the entire Term of this Lease.
- **4.2 No Offsets**. Lessee shall pay all Rent without offset, defense, claim, counterclaim, reduction, or deduction of any kind whatsoever.
- **4.3 Payment Generally**. [RESERVED]

ARTICLE 5: ADDITIONAL PAYMENTS BY LESSEE; REAL ESTATE TAXES

5.1 Net Lease. This Lease shall constitute an absolute "net" lease. Lessee shall pay as "**Additional Rent**" and discharge, before failure to pay creates a material risk of forfeiture or penalty, each and every

item of expense, of every kind and nature whatsoever, related to or arising from the Premises, or by reason of or in any manner connected with or arising from the leasing, operation, management, maintenance, repair, use, or occupancy of, or permitted Construction affecting, the Premises. Notwithstanding anything to the contrary in this Lease, Lessee need not pay, Lessee may offset against Rent any sums paid by Lessee on account of, and Lessor shall Indemnify Lessee against payment of, the following items payable, accrued, or incurred by Lessor: [(a) depreciation, amortization, brokerage commissions, financing or refinancing costs, management fees, or leasing expenses for the Fee Estate]; (b) consulting, overhead, accounting, tax preparation, other professional fees, legal and staff costs, and other costs incidental to Lessor's ownership of the Fee Estate and administration and monitoring of this Lease, including such costs Lessor incurs in reviewing anything Lessee delivers under this Lease or determining whether Lessee is in compliance with this Lease, except where this Lease expressly provides otherwise; (c) any costs or expenses that Lessor incurs in or for any litigation, except to the extent that this Lease requires Lessee to pay such costs or expenses; (d) any insurance premiums, utilities, operating expenses, or other costs related to the Premises that accrued before the Commencement Date; (e) any sums payable by Lessor under this Lease or expressly excluded from the definition of Real Estate Taxes; and (f) all other costs or expenses that, by their nature, are personal to Lessor or Lessor's ownership of the Fee Estate.

- **Real Estate Taxes**. Lessee shall pay and discharge all Real Estate Taxes, if any, payable or accruing for all periods within the Term, before failure to pay creates a material risk to Lessor of forfeiture or penalty, subject however to Lessee's right of Contest as this Lease expressly provides. Lessee shall also pay all interest and penalties any Government assesses for late payment of any Real Estate Taxes. Lessee shall within a reasonable time after Notice from Lessor give Lessor reasonable proof that Lessee has paid any Real Estate Taxes that this Lease requires Lessee to pay prior to the date when the same shall be due. Lessee shall have the sole right and authority to contest Real Estate Taxes, in compliance with the Contest Conditions. Lessee shall also have the right to apply for any applicable exemption from Real Estate Taxes applicable to the Premises, and Lessor shall cooperate as may be reasonably requested to provide such certifications and other information in support of Lessee's Application and subsequent renewals throughout the Term.
- **5.3** Assessments in Installments. To the extent allowed by Law, Lessee may apply to have any assessment payable in installments. Upon approval of such application, Lessee shall pay and discharge only such installments as become due and payable during the Term.
- **5.4 BID Decisions**. Intentionally Omitted.
- **5.5 Direct Payment by Lessor.** If any Additional Rent must be paid directly by Lessor then: (a) Lessor shall Notify Lessee of such Additional Rent and the payee entitled thereto, such Notice constituting Lessee's authorization to make such payment, insofar as applicable, on behalf of Lessor, and (b) if the payee nevertheless refuses to accept payment from Lessee, then Lessee shall Notify Lessor and shall pay such amount to Lessor in a timely manner with reasonable instructions on remittance of such payment. Lessor shall with reasonable promptness comply with Lessee's reasonable instructions. In such event, provided Lessee provides Lessor with payment and reasonable instructions on remittance of such payment in a reasonable amount of time prior to the payment due date, Lessor shall be responsible for any late charges or fees, interest or penalties arising from Lessor's payment if Lessor is late in making such payment.
- **5.6 <u>Utilities</u>**. Lessee shall arrange and pay directly, before the same become delinquent, for all fuel, gas, electricity, light, power, water, sewage, garbage disposal, telephone, and other utility charges, and the expenses of installation, maintenance, use, and service in connection with the foregoing, for the Premises during the Term. Lessor shall cooperate as may be reasonably requested by Lessee to arrange for such direct billing. Lessor shall have absolutely no liability or responsibility for the foregoing, provided that Lessor performs its obligations regarding any related Application.

- 5.7 Excise Tax. Lessee shall pay to Lessor, as Additional Rent, the applicable State general excise or surcharge tax on gross income, as the same may be amended, and all other similar applicable taxes, surcharges, rates and/or charges imposed upon Lessor with respect to rental or other payments in the nature of a gross receipts tax, sales tax, privilege tax, surcharge or the like, excluding federal, state or county net income taxes, imposed by any Government (collectively, the "Excise Tax"), such Excise Tax to be paid at the time and together with each payment of Fixed Rent and Additional Rent (which includes any and all charges required under this Lease to be made by Lessee to Lessor) to the extent they are subject to the Excise Tax. The Excise Tax due from Lessee shall be the amount which, when added to the applicable Rent due or other payment (whether actually or constructively received by Lessor), shall yield to Lessor (after deduction of all such tax payable by Lessor with respect to all such payments) a net amount which Lessor would have realized from such payment had no such tax been imposed. It is the intent of this Section 5.7 that Rent will be received by Lessor without diminution by any tax, assessment, charge, or levy of any nature whatsoever, except net income taxes imposed by any Government, and the terms and conditions of this Lease shall be liberally construed to effect such purpose. Lessee shall also have the right to apply for any applicable exemption from Excise Tax, and Lessor shall cooperate and execute such documents as may be reasonably requested by Lessee to support such Application.
- **Conveyance Tax**. Lessee shall pay the entire amount of any conveyance tax or related tax imposed by Law on account of this Lease or any amendment to this Lease (including, without limitation, to the extent resulting from any increase in Rent under this Lease and/or any renewal or extension of the Term), and for preparing, executing and/or filing when due such documentation as may be necessary or proper in connection therewith. If Lessor chooses, in its sole discretion, to collect said conveyance tax from Lessee and pay it to the tax authority on behalf of Lessee, Lessee shall promptly pay said conveyance tax to Lessor. At Lessor's request, Lessee shall promptly execute such affidavits and other documents as may be necessary or proper in connection with said conveyance tax. Lessee's obligations as aforesaid shall survive the expiration or earlier termination of this Lease. Notwithstanding the foregoing, Lessee has requested that the conveyance tax certificate for the Fixed Rent that is due and payable during the Term be paid, and has executed and delivered a draft to Lessor for review. Lessee shall submit the fully executed conveyance tax certificate to the Bureau, together with payment, within ten (10) days of the Commencement Date.
- **Taxes on Lessee's Business and Personal Property**. Lessee shall be responsible for and shall pay before delinquency all taxes assessed by any Government against Lessee by reason of the conduct of its business in the Premises or with respect to personal property of any kind owned by or placed in, upon or about the Premises by or at the expense of Lessee.
- **5.10** Tax Exemptions. Nothing contained herein shall prevent Lessee from applying for any exemptions which may be available to Lessee for its Real Estate Tax, Excise Tax, conveyance tax or other tax obligations; provided however, that it shall be Lessee's sole responsibility to apply for and maintain any such exemptions as and when required by Law. Lessor shall promptly cooperate and execute such documents as may be reasonably requested by Lessee to support such Application and renewals throughout the Term. Any such exemptions are subject to all Laws applicable thereto.
- **5.11** Lessor Expenses. Lessee shall pay to Lessor, within ten (10) Business Days after the date of mailing or personal delivery of statements, all reasonable out-of-pocket costs and expenses, including attorneys' fees, reasonably paid or incurred by Lessor: (i) required to be paid by Lessee under any covenant in this Lease (including without limitation any indemnity provision), (ii) in enforcing any of Lessee's covenants or obligations in this Lease, (iii) in remedying any breach of this Lease by Lessee, (iv) in recovering possession of the Premises or any part of the Premises, (v) in collecting or causing to be paid any delinquent rent, taxes or other charges payable by Lessee under this Lease, (vi) in connection with any estoppel certificate requested by Lessee, or (vii) in connection with any litigation (other than condemnation proceedings) commenced by or against Lessee to which Lessor shall without fault be made a party. All

such costs, expenses and fees shall constitute Additional Rent, and Lessee's obligations under this <u>Section 5.11</u> shall survive the expiration or earlier termination of the Term.

ARTICLE 6: USE

- **6.1 Permitted Use**. Subject to all applicable Laws and this Lease, Lessee shall use the Premises only for the following purposes:
- **6.1.1** Affordable Residential Rental Units for Seniors. Lessee shall design, construct, operate and maintain Residential Rental Units, in the number and of the type described below, for persons 62 years of age or older who meet the following income qualifications associated with said Residential Rental Units:

No. of units	No. of bedrooms	Income qualifications
8	studio	At or below 30% of HUD area median income
5	1	At or below 30% of HUD area median income
45	studio	At or below 50% of HUD area median income
30	1	At or below 50% of HUD area median income
30	studio	At or below 60% of HUD area median income
21	1	At or below 60% of HUD area median income
1	2	Staff unit

Total: 140 residential units: 139 Residential Rental Units

- **6.1.2** Parking and Loading. Lessee shall design, construct, operate and maintain a parking lot, to include a total of 95 parking stalls and 1 loading space, including a total of 10 parking stalls that shall be reserved for persons with disabilities in compliance with Public Accommodations Laws. Use of the parking stalls and loading space shall be restricted to residents, Community Center (defined below) visitors, and other approved guests. Lessee shall not charge residents and their guests for such use without the prior review and written approval of the Lessor, which Lessor may withhold in its reasonable discretion. Public parking shall not be allowed, with or without charge.
- **6.1.3** Resident Amenities. Except as otherwise set forth in Section 6.1.4, Lessee shall design, construct, operate and maintain the following amenities available exclusively for residents and their guests:
 - i) Parking and Loading;
 - ii) Community Center (as defined below); and
 - iii) Common Area Amenities (as defined below).
- **6.1.4** <u>Community Center</u>. Lessee shall design, construct, operate, and maintain a community center of approximately 1,037 square feet (the "**Community Center**"). The Community Center is incorporated into the first floor of the Improvements and is comprised of a multi-purpose room with kitchen

and storage. Lessee shall provide services and programs for residents and the community, as determined by Lessee in consultation with community organizations and agencies.

- **6.1.5** <u>Common Area Amenities</u>. Lessee shall design, construct, operate, and maintain common area amenities, including but not limited to, a tech lounge, outdoor garden, dog park, landscaping, and an outdoor passive recreation area (together, the "**Common Area Amenities**"). Indigenous species of plants shall be used for landscaping whenever and wherever reasonably feasible.
- **Prohibited Uses**. Lessee shall not cause, maintain or permit any waste or nuisance to exist on, in or about the Premises. Lessee shall not do or permit anything to be done in or about the Premises which will in any way damage the Premises, or use or allow the Premises to be used for any improper, offensive or unlawful purpose or any use other than the Permitted Use.
- Exclusive Ownership of Improvements and Control of Premises. During the Term, Lessee 6.3 shall be the exclusive owner of the Improvements including FF&E (collectively, the "Project Improvements"), which are located within the Premises. For federal income tax purposes, Lessee alone shall be entitled to all of the tax attributes of ownership of the Improvements and Building Equipment, including, without limitation, the right to claim depreciation or cost recovery deductions, and Lessee shall have the right to amortize capital costs and claim the low-income housing tax credit described in Section 42 of the Code and Lessee shall have the right to amortize capital costs and to claim any other federal or state tax benefits attributable to the Improvements and Building Equipment. Except as otherwise expressly provided in this Lease, Lessee shall have exclusive control, use, and management of the Project and Premises during the Term. Subject to any applicable Laws, Lessee, in its sole discretion, may enter into, terminate, or Modify any existing or future contract for management or operation of the Premises or provision of services to the Premises (provided that as to contracts existing as of the Commencement Date, (a) a copy of any such existing contract has been provided to Lessee prior to the Commencement Date, and (b) such existing contract has been assigned to and assumed by Lessee as of the Commencement Date (individually and collectively, "Assumed Existing Contract"), any such termination or Modification of an Assumed Existing Contract is done in conformity with the terms of such Assumed Existing Contract). Lessor shall be responsible for the termination of an existing contract that is not assigned to and assumed by Lessee. Lessee shall Indemnify Lessor arising out of such cancellation or termination of any Assumed Existing Contract. All such contracts shall expire automatically on or before the Scheduled Expiration Date, except for contracts entered into in the ordinary course of maintenance and operation of the Premises, which shall expire no later than one (1) year after the Scheduled Expiration Date and shall be assigned to and assumed by Lessor effective as of the Expiration Date pursuant to Article 23. Following the end of the Term, ownership of the Project Improvements located on the Land shall revert to Lessor and remain with the Premises.
- **Compliance with Laws**. Lessee shall not use the Premises, or do anything or suffer anything to be done in or about the Premises that will in any way conflict with or constitute a violation of any Laws applicable to the use, condition or occupancy of the Premises, subject to Article 9 of this Lease. At its sole cost and expense, Lessee shall, in all material respects and subject to Lessee's right of Contest, promptly comply with all such applicable Laws. Lessee shall, at its sole cost and expense, make all alterations to the Premises, and to any adjacent land between the Premises and any public street, that are required to comply with applicable Laws, whether in effect as of the Commencement Date or thereafter, subject to Article 9 of this Lease. Lessee's obligations under this Section 6.4 shall include the obligation that Lessee, at its sole cost and expense, in accordance with the terms of this Lease, make, build, maintain and repair all fences, sewers, drains, roads, curbs, sidewalks, parking areas and other improvements that are located within the Premises and are required by applicable Laws to be made, built, maintained and repaired in connection with Lessee's or its Sublessees' use of the Premises or any part of the Premises and are located on the Premises.

- **Copies of Notices**. Lessor shall promptly give Lessee and Lessee shall promptly give Lessor a copy of any notice of any kind regarding the Premises or any Real Estate Taxes (including any bill or statement), and any notice of nonrenewal or threatened nonrenewal of any Approval that Lessor receives from any Government, utility company or insurance carrier affecting the Premises.
- **6.6 Entitlements**. Lessee shall maintain in full force and effect all entitlements and permits necessary for the Permitted Uses specified in <u>Section 6.1</u>.
- 6.7 Illegal Activities. Upon Lessor's receipt of actual notice or if Lessor, in its reasonable business judgment, believes or suspects that illegal acts are taking place on the Premises, or the Premises are being used for an illegal purpose that could result in criminal or civil forfeiture, or both, of the Premises or any portion of the Premises, to any Government, Lessor may Notify Lessee of such suspected illegal activity and basis for Lessor's belief, and provide Lessee with a reasonable opportunity to investigate such illegal activity of not less than three (3) days following receipt of the Notice and, if such investigation confirms the conduct of such illegal activity, not less than two (2) days to cause such illegal activity to stop, and if not resolved within such 5-day period (or such longer period as may be approved by Lessor), Lessor may thereafter take all reasonable and appropriate action as may be necessary to stop such illegal activity. including entry onto the Premises (provided that any entry into a Residential Rental Unit shall be subject to compliance with applicable laws, rules and regulations). In such circumstances, Lessor shall have the right to investigate, including, without limitation, the right of entry to the Premises and a review of Lessee's records. For any entry onto the Premises, Lessor shall (a) provide Lessee with written Notice prior to such entry, unless it is an emergency (provided that any entry into a Residential Rental Unit shall be subject to compliance with applicable laws, rules and regulations, including 48 hours advance notice to the tenant of a Residential Rental Unit), (b) meet with Lessee's property manager and/or designated representative, or if neither is present, the individual who is physically present at the Premises on the day of the entry and claims to be Lessee's representative, and (c) have such property manager or representative accompany Lessor during its entry. If such investigation yields any evidence of any illegal activity on the Premises, Lessor shall immediately Notify Lessee of the results of such investigation and Lessor may immediately take all reasonable and appropriate action as may be necessary to stop such illegal activity and shall Notify Lessee of any such action to be taken by Lessor. If Lessee unreasonably refuses to commence any action to investigate such illegal activity within three (3) days of receipt of such Notice from Lessor, such failure or refusal shall constitute an Event of Default. By having the right to take certain actions in this Section 6.7, Lessor is neither obligated nor required to take any such action, and shall not be liable to Lessee, any Person or any Government if Lessor does not exercise such right.
- Public Accommodations Laws. Without limiting Lessee's obligation to comply generally with all applicable Laws, Lessee, at its sole cost and expense, shall cause the Premises, including all Improvements, and Lessee's use and occupancy of the Premises, and Lessee's performance of its obligations under this Lease, to comply with the requirements of the Public Accommodations Laws, and to take such actions and make such alterations or reasonable accommodations as are necessary for such compliance. If Lessee concludes that the Premises are not in compliance with Public Accommodations Laws as of the Commencement Date, or that the Premises thereafter fail to comply with Public Accommodations Laws, then Lessee shall provide to Lessor a plan for compliance within one hundred twenty (120) days of the Commencement Date or the date such subsequent noncompliance is determined. At minimum, such plan shall identify the work to be done to cause the Premises to be in compliance with Public Accommodations Laws and the timetable for completing such work. Any such work shall be subject to Lessor's reasonable approval, and to the terms and conditions of Article 7 and Article 8 as applicable.
- **6.9 Use of Public Buildings by Blind or Visually Handicapped Persons.** Lessee acknowledges its understanding that the Premises may be subject to the provisions of HRS Section 102-14 relating to the placement of certain vending machines and concessions in public buildings for operation by blind or visually

handicapped persons. If applicable, Lessee shall comply with the requirements of Section 102-14 and the rules adopted by the State Department of Human Services, to implement this Section.

6.10 Project Name. Lessee agrees that it will not change the name by which the Project is known or identified without the prior written approval of Lessor.

ARTICLE 7: MAINTENANCE, REPAIR AND CAPITAL IMPROVEMENTS

- 7.1 Obligation to Maintain. Lessee, at its sole cost and expense, shall keep and maintain the Premises in good order, condition and repair, reasonable wear and tear excepted, at all times during the Term (which obligation shall include all structural and non-structural, and capital and non-capital, repairs and replacements including, without limitation, plumbing, heating, air conditioning, ventilating, electrical, lighting, fixtures, walls, building systems, ceilings, floors, windows, doors, plate glass, skylights, landscaping, driveways, site improvements, curb cuts, parking lots, fences and signs located in, on or at the Premises). Lessee shall manage and operate the Premises and perform its duties and obligations under this Lease in a manner consistent with the standards followed by institutional quality owners and management companies that are managing comparable projects. Lessee shall cause the Improvements to be inspected periodically by qualified Persons to be certain the repair, maintenance and replacement obligations of Lessee pursuant to the terms of this Lease are being satisfied (including for purposes of ascertaining and curing infestation of the Improvements by termites, rodents and other pests). Lessee shall thereafter take all measures that may be reasonably required to prevent or cure any discovered repair, maintenance and/or replacement item. Lessee expressly acknowledges and agrees that Lessor has no obligation whatsoever for the repair, alteration, maintenance, and replacement of the Premises except as expressly provided herein.
- **7.2** <u>Capital Improvement Obligations</u>. Lessee shall demolish and clear the Improvements on the Premises existing as of the Commencement Date, and develop the Project on the Premises, all as provided and detailed in the Development Agreement. Lessee shall carry out the capital improvements in accordance with the timing, terms, and conditions set forth in the Development Agreement. Lessee shall thereafter make subsequent capital improvements during the Term as may be required to comply with any applicable Laws, or as necessary to comply with Lessee's obligations under this Lease.

ARTICLE 8: CONSTRUCTION

- **8.1** General. Lessee shall comply with all of the terms of this Article 8 and of the Development Agreement in connection with all Construction affecting the Premises (including, without limitation, any existing and new Improvements, alterations, any capital improvements to the Premises, Restoration after a Loss, and those required to comply with applicable Laws or otherwise required under this Lease).
- **8.2** <u>Commencement of Construction</u>. Lessee shall not commence Construction until a Notice to Proceed has been issued by the Lessor, and until all conditions precedent to the commencement of construction, as set forth in the Development Agreement, have been satisfied.
- **8.3** Manner of Construction. All Construction shall be diligently and continuously pursued from the commencement thereof through completion, and shall be performed in a good and workmanlike manner, strictly in conformance with any and all Laws, in accordance with any approved plans and specifications, and in accordance with the terms of the Development Agreement.

- **8.4** Permits. Lessee shall be responsible for obtaining, at its sole cost and expense, all Approvals required for any Construction, and for any issuance or re-issuance of all certificates of occupancy or equivalent permits required by Law for the use and occupancy of the Premises. Notwithstanding the foregoing, Lessee shall apply for and prosecute any required Government review process for a general plan amendment or rezoning only through and in the name of Lessor, or otherwise with the approval of Lessor.
- **Applications**. Upon Lessee's request, Lessor shall, without charge to Lessee (except reimbursement of Lessor's reasonable out-of-pocket costs and expenses), promptly join in and execute any Application as Lessee reasonably requests, provided that: (a) such Application is in customary form and imposes no material obligations (beyond obligations ministerial in nature or merely requiring compliance with Law, this Lease or the Development Agreement) upon Lessor; and (b) no uncured Event of Default exists. Promptly upon Lessee's request and without charge (except reimbursement of Lessor's reasonable out-of-pocket costs and expenses), Lessor shall furnish all information in its possession that Lessee reasonably requests for any Application.
- **Bond.** Lessee or its general contractor shall deposit with Lessor performance and payment bonds, underwritten by a corporate surety licensed to issue bonds in the State, that is listed on the United States Department of Treasury Circular 570, "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds" and that holds an A.M. Best rating of A or higher. The total amount of the bond shall not be less than 100% of the total Construction cost, naming Lessor and Lessee as co-obligees, in form and content reasonably satisfactory to Lessor, guaranteeing the full and faithful performance of the contract for such Construction free and clear of all mechanics' and materialmen's liens and the full payment of all subcontractors, labor and materialmen.
- **8.7** Completion. Upon substantial completion of any Construction: (a) Lessee shall properly publish and file a "Notice of Completion" in the Office of the Clerk of the Circuit Court of the State in the circuit where the Premises are located, a certified "filed" stamped copy of which shall be provided by Lessee to Lessor, (b) Lessee shall comply with any other applicable requirements of Law with respect to the completion of works of improvement, and (c) Lessee's general contractor or architect shall deliver to Lessor a certificate setting forth the total cost of such Construction and, if the Construction is Major Construction, certifying that the Construction has been completed in compliance with the approved plans and specifications for such work. In addition, Lessee shall deliver to Lessor a reproducible copy of the "as built" drawings of all Construction as well as all Approvals and other Government documents, if any, issued in connection with such Construction.
- **8.8** Construction Insurance. Lessee shall maintain insurance coverage in accordance with Article 12.
- **Ownership**. All Improvements from Construction that may be installed or placed in or about the Premises shall be owned by Lessee during the Term, deemed to become an integral part of the Premises and shall not be removed or otherwise severed from the Premises except as otherwise permitted by this Lease. Upon the expiration of the Term or earlier lawful termination of this Lease, all the then existing Improvements shall automatically revert to Lessor without compensation or payment of any kind to, or requirement of consent or other act of Lessee or any other Person, and without the necessity of executing a deed, bill of sale, conveyance or other act or agreement of Lessee or any other Person. If requested by Lessor, Lessee shall, without charge to Lessor, execute, acknowledge and deliver to Lessor appropriate documentation (in form and content reasonably satisfactory to Lessor) which acknowledges and confirms that Lessor retains all of right, title and interest in and to the then existing Improvements as of the expiration of the Term or earlier termination of this Lease.
- **8.10** <u>Inspection</u>. During and upon completion of any Construction, Lessor and its agents may, following not less than 24 hours written notice, inspect the Improvements and all work and materials as rendered and

installed, provided that following completion of Construction, any inspection of occupied units shall be subject to compliance with the rights of any residential tenant under a tenancy agreement and tenant-landlord notice requirements. Lessee shall permit Lessor and its representatives to examine the Construction at all reasonable times following such written notice and shall furnish Lessor with copies of all plans, shop drawings and specifications relating to any Construction. Lessor shall comply with all reasonable safety rules and requirements of Lessee or its contractors during any such inspections.

- **Lessee's Covenant**. Lessee covenants to keep the Premises free from all Prohibited Liens arising out of any work performed, materials furnished or obligations incurred by or on behalf of Lessee. Lessee shall remove any Prohibited Liens by bond or otherwise within 30 days after Lessee is informed of the existence of such lien or encumbrance, or if a Prohibited Lien cannot be removed within such 30 day period, and provided that Lessee promptly begins and diligently pursues the removal in good faith during the 30 day period, Lessee shall have such additional time as is reasonably necessary, not to exceed ninety (90) days in the aggregate to remove any Prohibited Liens, and if Lessee shall fail to remove such Prohibited Liens, Lessor may pay the amount necessary to remove such Prohibited Liens, without being responsible for investigating the validity thereof. The amount so paid shall be deemed Additional Rent under this Lease payable upon demand, without limitation as to other remedies available to Lessor under this Lease. Nothing contained in this Lease shall be deemed or construed in any way to constitute Lessor's consent or request, express or implied, to any contractor, subcontractor, laborer, equipment or material supplier for the performance of any labor or the furnishing of any materials or equipment for any Construction, nor as giving Lessee any right, power or authority to contract for, or permit the rendering of, any services, or the furnishing of any materials that would give rise to the filing of any liens against the Fee Estate.
- 8.12 Title Encumbrances. Lessee shall keep the Fee Estate free from any encumbrances against title, and shall not record or permit the recordation of any lien, encumbrance, easement, memorandum of Sublease or other document that affects the record title to the Fee Estate without Lessor's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed; provided that Lessor shall use commercially reasonable efforts to execute a written consent to such requested encumbrance or provide a written denial of consent (with specific reasons for Lessor's denial of such request) as soon as reasonably practicable following receipt of Lessee's written request for such consent (and, if such request involves an easement, documentation evidencing that final subdivision approval and grant of such easement area has been obtained from the applicable Governmental authority or public utility company). Lessee shall cause any recorded title encumbrances arising during the Term (other than those caused to be recorded by Lessor), which are consented to in writing by Lessor, to be released as of the earlier of (a) the date each individual encumbrance ceases to be valid (e.g., as of the expiration of a Sublease for which there is a recorded memorandum), and (b) the date this Lease expires or earlier terminates. Lessee shall cause any recorded title encumbrances on the Fee Estate arising during the Term (other than those caused to be recorded by Lessor), which are not consented to in writing by Lessor, to be cleared immediately, and in any event, within thirty (30) days of a written demand by Lessor. Should Lessee fail to clear any recorded title encumbrances, then Lessor may, but without obligation to do so, without notice or demand to or upon Lessee, and without releasing Lessee from any obligation hereof, clear any recorded title encumbrances, provided Lessor shall give Lessee one Business Day prior written notice of same. In exercising any such powers, Lessor may pay its reasonable costs and expenses, employ counsel and incur and pay reasonable attorneys' fees (prior to trial, at trial and on appeal), and shall receive reimbursement thereof from Lessee upon demand. Lessee hereby grants to Lessor an irrevocable power of attorney, coupled with an interest, to perform all of the acts and things provided for in this Section 8.12 as Lessee's agent and in Lessee's name. During the Term, Lessor shall not encumber the Premises or Lessee's leasehold interest therein.
- **8.13** <u>Lease Termination</u>. If this Lease expires or is terminated prior to the completion of any Construction, Lessee shall, at Lessor's option and at Lessee's sole expense, either (a) promptly complete such Construction, or (b) remove all such partially completed improvements, construction materials,

equipment and other items from the Premises and restore the Premises to their pre-Construction condition.

ARTICLE 9: HAZARDOUS SUBSTANCES

- **Restrictions**. Lessee shall not cause and shall not permit to occur on, under or at the Premises during the Term: (a) any violation of any Environmental Laws; or (b) the use, generation, release, manufacture, refining, production, processing, storage, or disposal of any Hazardous Substance, or transportation to or from the Premises of any Hazardous Substance, unless both: (i) reasonably necessary and customary in accordance with professional standards for the Permitted Uses of the Premises and (ii) in compliance with all Environmental Laws. Notwithstanding the foregoing, Lessee shall not have any clean-up obligation or other responsibility under this Lease with respect to any Pre-Existing Hazardous Substances Condition or any Unrelated Hazardous Substance Discharge.
- 9.2 Compliance: Clean-Up. Except with respect to any Pre-Existing Hazardous Substances Condition or Unrelated Hazardous Substance Discharge, Lessee shall, at Lessee's sole expense: (a) comply with Environmental Laws, including compliance with an environmental hazardous materials plan set forth in the Ford Report or similar requirements approved by the applicable Governmental agency and, to the extent Environmental Laws require, clean up any Hazardous Substances Discharge; (b) make all submissions to, deliver all information required by, and otherwise fully comply with all requirements of any Governmental agency under Environmental Laws; (c) if any Governmental agency requires any clean-up plan or clean-up because of a Hazardous Substances Discharge, prepare and submit the required plans and all related bonds and other financial assurances; (d) promptly and diligently carry out all such clean-up plans; and (e) Indemnify Lessor against any Hazardous Substances Discharge or violation of Environmental Laws that occurred during Lessee's possession of the Premises or that is attributable to the Project or Lessee's use, occupancy, or activities on the Premises (excluding however, any Pre-Existing Hazardous Substances Condition or Unrelated Hazardous Substances Discharge). Notwithstanding the foregoing, Lessee shall inform Lessor of any Pre-Existing Hazardous Substances Condition that is discovered by Lessee during the Term, and the parties shall negotiate in good faith to mutually agree upon a remediation or clean-up plan under the circumstances. In the event that Lessor and Lessee mutually agree upon a clean-up plan, only then shall Lessee have the right but not the obligation to clean up any Pre-Existing Hazardous Substances Condition, in which event Lessor shall promptly reimburse Lessee for the actual costs and expenses incurred by Lessee in connection with the clean-up of any such Pre-Existing Hazardous Substances Condition and, if Lessor is the responsible party, any Unrelated Hazardous Substance Discharge, in accordance with the requirements under Environmental Law. Lessor's obligations under this Section 9.2 to reimburse Lessee shall not limit Lessor's or Lessee's respective rights against third parties whatsoever, with respect to any Pre-Existing Hazardous Substances Condition or Unrelated Hazardous Substances Discharge, as applicable.
- **Surrender Obligations**. At the expiration or earlier termination of the Term, Lessee, at Lessee's sole expense, shall cause all Hazardous Substances at the Premises to be in compliance with Environmental Laws, which may include ongoing compliance with an environmental hazardous materials plan set forth in the Ford Report or similar requirements approved by the applicable Governmental agency, except that Lessee shall not be responsible for any Pre-Existing Hazardous Substances Condition or Unrelated Hazardous Substance Discharge. Lessee shall dispose of all Hazardous Substances required to be removed by Lessee pursuant to this Section 9.3 in accordance with all Environmental Laws. Lessee shall cause to be repaired any damage to the Premises caused by such removal. Prior to surrendering the Premises at the Expiration Date, Lessee shall clean up and/or remediate the Land to substantially the same environmental condition it was in as of the Commencement Date (but excluding any clean up and/or remediation of any Pre-Existing Hazardous Substances Condition or Unrelated Hazardous Substances Discharge), as established by the environmental studies and assessments conducted by Lessee prior to

the Commencement Date (including the Phase 1 and/or Phase 2 environmental site assessments of the Premises conducted by Lessee and the Ford Report), as necessary for an authoritative Government determination to allow the continued use of the Premises for residential purposes consistent with the use permitted under this Lease.

- **Copies of Environmental Reports**. Within thirty (30) days of receipt thereof, Lessee shall provide Lessor with a copy of any and all environmental assessments, audits, studies and reports with respect to the Premises, or ground water beneath the Land, or the environmental condition or any clean-up thereof that is conducted by or on behalf of Lessee. Subject to any court order or contractual obligation that is binding on Lessee, Lessee shall be obligated to provide Lessor, without any representation or warranty, with a copy of such materials without regard to whether such materials are generated by Lessee or prepared for Lessee, or how Lessee comes into possession of such materials; provided that Lessee shall not be liable to Lessor for the scope, information and/or contents of any such materials.
- **9.5 Survival**. Each covenant, agreement, representation, warranty and indemnification made by Lessee set forth in this <u>Article 9</u> shall survive the expiration or earlier termination of this Lease and shall remain effective until all of Lessee's obligations under this Article 9 have been performed and satisfied.
- 9.6 <u>Discharges before Commencement</u>. Lessee agrees to accept the Premises in "AS IS, WHERE IS AND WITH ALL FAULTS" condition as described in <u>Section 2.2</u>, including the soil contamination identified in the Ford Report, but Lessee shall have no responsibility for or to clean up or remediate any Pre-Existing Hazardous Substances Condition, as established and identified by environmental studies and assessments conducted by Lessee prior to or after the Commencement Date (including the Phase 1 and/or Phase 2 environmental site assessment conducted by Lessee for the Premises) or other proof of existence or occurrence of such Pre-Existing Hazardous Substances Condition prior to the Commencement Date. As between Lessor and Lessee, any Pre-Existing Hazardous Substances Condition is and shall continue to be Lessor's sole responsibility.

ARTICLE 10: INDEMNIFICATION; LIABILITY OF LESSOR

- Obligations. Lessee shall Indemnify Lessor against any: (a) wrongful act, wrongful omission, or negligence of Lessee (and anyone claiming by or through Lessee) or its or their partners, members, directors, officers, or employees relating to Lessee's use, operation, or occupancy of the Premises or this Lease; (b) breach or default by Lessee under this Lease; or (c) breach of any representation or warranty Lessee makes in this Lease. In addition, Lessee shall Indemnify Lessor against the following during the Term and so long as Lessee remains in possession after the Expiration Date: (i) any Contest Lessee initiates; (ii) any Application made at Lessee's request; (iii) Lessee's use, occupancy, control, management, operation, and possession of the Premises; (iv) any Construction and any agreements that Lessee (or anyone claiming through Lessee) makes for any Construction; (v) the condition of the Premises from and after the Commencement Date (except as otherwise provided in Article 9 and Section 10.2 of this Lease); and (vi) any accident, injury or damage whatsoever caused to any person or property in or on or originating from the Premises. Notwithstanding anything to the contrary in this Lease, Lessee shall not be required to Indemnify Lessor for claims, liabilities, or losses arising solely from Lessor's and/or its employees' (and anyone claiming by or through Lessor) intentional acts or omissions or negligence or from any Pre-Existing Hazardous Substance Condition, any Unrelated Hazardous Substance Discharge and as otherwise provided in this Lease.
- **10.2** <u>Liability of Lessor</u>. During the Term: (a) Lessee is and shall be in exclusive control and possession of the Premises; and (b) Lessor shall not be liable for any injury or damage to any property (of Lessee or any other Person) or to any person occurring on or about the Premises, except to the extent

caused by Lessor's and/or its employees' (and anyone claiming by or through Lessor) willful misconduct or gross negligence. Lessor's right to enter and inspect the Premises is intended solely to allow Lessor to ascertain whether Lessee is complying with this Lease and (to the extent this Lease allows) to cure any Default. Such provisions shall not impose upon Lessor any liability to third parties. Nothing in this Lease shall be construed to exculpate, relieve, or Indemnify Lessor from or against any liability of Lessor: (i) to third parties existing at or before the Commencement Date; or (ii) arising from Lessor's and/or its employees' (and anyone claiming by or through Lessor) willful misconduct or gross negligence.

- **10.3** <u>Indemnification Procedures</u>. Wherever this Lease requires any Indemnitor to Indemnify any Indemnitee:
- **10.3.1** Prompt Notice. Indemnitee shall promptly Notify Indemnitor of any claim. To the extent, and only to the extent, that Indemnitee fails to give prompt Notice and such failure materially prejudices Indemnitor, Indemnitor shall be relieved of its indemnity obligations for such claim.
- **10.3.2** <u>Selection of Counsel</u>. Indemnitor shall select counsel reasonably acceptable to Indemnitee. Counsel to Indemnitor's insurance carrier shall be deemed satisfactory. Even though Indemnitor shall defend the action, Indemnitee may, at its option and its own expense, engage separate counsel to advise it regarding the claim and its defense. Such counsel may attend all proceedings and meetings. Indemnitor's counsel shall actively consult with Indemnitee's counsel. Indemnitor and its counsel shall, however, fully control the defense.
- **10.3.3** <u>Cooperation.</u> At Indemnitor's request, Indemnitee shall reasonably cooperate with Indemnitor's defense, provided Indemnitor reimburses Indemnitee's actual reasonable out of pocket expenses (including Legal Costs) of such cooperation.
- **10.3.4** <u>Settlement</u>. Indemnitor may, with Indemnitee's consent, not to be unreasonably withheld, settle the claim. Indemnitee's consent shall not be required for any settlement by which: (a) Indemnitor procures (by payment, settlement, or otherwise) a release of Indemnitee by which Indemnitee need not make any payment to the claimant; (b) neither Indemnitee nor Indemnitor on behalf of Indemnitee admits liability; (c) the continued effectiveness of this Lease is not jeopardized in any way; and (d) Indemnitee's interest in the Premises is not jeopardized in any way.
- **10.3.5** <u>Insurance Proceeds</u>. Indemnitor's obligations shall be reduced by net insurance proceeds Indemnitee actually receives for the matter giving rise to indemnification.

ARTICLE 11: RIGHT OF CONTEST

- 11.1 <u>Lessee's Right; Contest Conditions</u>. Notwithstanding anything to the contrary in this Lease, Lessee shall have the exclusive right to contest, at its sole cost, by appropriate legal proceedings diligently conducted in good faith, the amount or validity of any Real Estate Taxes or Prohibited Lien; the valuation, assessment, or reassessment (whether proposed, phased, or final) of the Premises for Real Estate Taxes; the amount of any Real Estate Tax; the validity of any Law or its application to the Premises; the terms or conditions of, or requirements for, any Approval; or the validity or merit of any claim against which this Lease requires Lessee to Indemnify Lessor (any of the foregoing, a "Contest"). Lessee may defer payment or performance of the contested obligation pending outcome of the Contest, provided that Lessee causes the following conditions (collectively, the "Contest Conditions") to remain satisfied:
- **11.1.1** <u>No Criminal Act</u>. Such deferral or noncompliance shall not constitute a criminal act by Lessor or subject Lessor to a material risk of any fine or penalty, except civil penalties for which Lessee

has given Lessor a bond, letter of credit, or other security reasonably satisfactory to Lessor (the "Contest Security") in an amount equal to the reasonably estimated amount of such civil penalties, provided that such civil penalties do not otherwise prejudice Lessor.

- **11.1.2** No Liability. Such deferral or noncompliance creates no material risk of a lien, charge, or other liability of any kind against the Fee Estate, unless Lessee has given Lessor Contest Security equal to the reasonably estimated amount of such lien, charge, or other liability, and such Contest Security otherwise is acceptable to Lessor.
- **11.1.3** <u>No Forfeiture</u>. Such deferral or noncompliance will not place the Fee Estate in material danger of being forfeited or lost.
- **11.1.4** <u>No Cost to Lessor</u>. Such Contest shall be without cost, liability, or expense, and without risk of future cost, liability, or expense, to Lessor.
- **11.1.5** <u>Diligence</u>. Lessee shall prosecute such Contest with reasonable diligence and in good faith.
- **11.1.6** <u>Payment</u>. If required for such Contest, Lessee shall have paid the Real Estate Taxes or other matter subject to the Contest.
- **11.1.7** <u>Collection of Real Estate Taxes</u>. If such Contest relates to any Real Estate Tax, then such Contest shall suspend its collection from Lessor and the Fee Estate.
- **11.1.8** No Event of Default. No uncured Event of Default shall exist under this Lease at the time of such Contest.
- **11.1.9** <u>Named Parties</u>. If Lessor has been named as a party in any action consistent with this <u>Article 11</u>, then Lessee shall cause Lessor to be removed as such party and Lessee substituted in Lessor's place, if practicable and permissible under the circumstances.
- 11.2 <u>Lessor Obligations and Protections</u>. Lessor need not join in any Contest unless Lessee has complied with the Contest Conditions, and such Contest must be initiated or prosecuted in Lessor's name. In such case, Lessor shall cooperate, as Lessee reasonably requests, to permit the Contest to be prosecuted in Lessor's name. Lessor shall give Lessee any documents, deliveries, and information in Lessor's control and reasonably necessary for Lessee to prosecute its Contest. Lessor shall otherwise assist Lessee in such Contest as Lessee reasonably requires. Lessee shall pay all reasonable costs and expenses, including Legal Costs, of any Contest. Lessee shall, at Lessor's request, advance (when Lessor incurs them) such reasonable costs and expenses as Lessor incurs or reasonably anticipates incurring, for Lessee's Contest and Lessor's assistance with such Contest.
- 11.3 <u>Miscellaneous</u>. Lessee shall be entitled to any refund of any Real Estate Taxes (and penalties and interest paid by Lessee), to the extent attributable to periods within the Term, whether such refund is made during or after the Term. When Lessee concludes Lessee's Contest of any Real Estate Taxes, Lessee shall pay the amount of such Real Estate Taxes (if any) as has been finally determined in such Contest to be due, to the extent attributable to periods within the Term, and any costs, interest, penalties, or other liabilities in connection with such Real Estate Taxes. Upon final determination of Lessee's Contest of a Law, Lessee shall comply with such final determination. So long as the Contest Conditions remain satisfied, Lessor (in its role as owner, and not in its role as Government authority) shall enter no objection to any Contest. Lessor may contest any matter for which Lessee is entitled to (but does not) prosecute a Contest, but only if: (a) Lessor Notifies Lessee of Lessor's intention to do so; (b) Lessee fails to commence such Contest within ten (10) Days after receipt of such Notice; and (c) Lessor's contest complies with all

conditions and covenants that would apply to a Contest by Lessee transposing references to the parties and their interests as appropriate.

11.4 Contest Security. Lessor shall promptly release any Contest Security to Lessee after the Contest has been resolved and Lessee has performed its obligations, if any, as determined by such resolution.

ARTICLE 12: INSURANCE

- **Lessee's Insurance**. At all times during the Term and until such time as action against the Lessee, the developer, its contractor or subcontractor for death, injuries, losses and damages is barred by the provisions of Chapter 657, HRS, Lessee shall procure and maintain in full force and effect, at Lessee's sole expense, any and all insurance that may be required by any Laws as they may pertain to Lessee's use and occupancy of, and operations at the Premises, as well as the following policies of insurance in the following amounts:
- **12.1.1** Workers' Compensation and Employers' Liability Insurance. The Lessee shall maintain workers' compensation and employers' liability insurance. Workers' compensation coverage shall be in accordance with State statutes. Employers' liability coverage shall provide limits of not less than **\$1,000,000** each accident for bodily injury by accident or **\$2,000,000** each employee, **\$2,000,000** aggregate, for bodily injury by disease. The policy will include a waiver of subrogation in favor of the Lessee.
- **12.1.2** <u>Liability Insurance</u>. Lessee shall maintain commercial general liability (CGL) with a limit of not less than \$2,000,000 each occurrence, and \$4,000,000 general aggregate, and \$4,000,000 aggregate for this Project. CGL insurance shall be written on ISO occurrence form, CG 00 01 (or a substitute form providing equivalent coverage), and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract). The policy shall be endorsed to redefine the word occurrence to include construction defect coverage. Lessor shall be included as an additional insured under the CGL, using ISO additional insured endorsement CG 20 10 (11/85) (or equivalent), and as appropriate CG 20 37 or equivalent and under the commercial umbrella. The policy(ies) shall contain a waiver of subrogation against Lessor.
- **12.1.3** <u>Business Automobile Liability Insurance</u>. Lessee shall maintain business auto liability (including no-fault coverage) insurance with a limit of not less than **\$1,000,000** each accident. Such insurance shall cover liability arising out of any auto (including owned, hired, and non-owned autos) used in the performance of this Lease. Business auto coverage shall be written on ISO form CA 00 01 or CA 00 20, or a substitute form providing equivalent liability coverage. If necessary, the policy shall be endorsed to provide contractual liability coverage.
- **12.1.4** <u>Umbrella/Excess Liability</u>. Lessee shall maintain umbrella/excess liability insurance with limits of not less than \$10,000,000 per occurrence combined single limits and aggregate, with the aggregate to apply per project/per location. Such insurance shall be written on an occurrence basis in excess of the underlying insurance described in <u>Sections 12.1.1 through 12.1.3</u>, which is at least as broad as each of the underlying policies, and otherwise including "pay on behalf" wording, concurrency of effective dates with underlying primary coverages, blanket contractual liability, and construction defect coverage. The amounts of insurance required in <u>Sections 12.1.1 through 12.1.4</u> may be satisfied by Lessee purchasing coverage for the limits specified or by any combination of underlying and umbrella limits, so long as the total amount of insurance is not less than the total limits required by this Section 12.1.4.

- **12.1.5** Professional Liability Insurance. Lessee's consultants and design professionals shall maintain professional liability insurance with limits of not less than \$1,000,000 per occurrence/annual aggregate, covering the consultant or design professional, their employees and agents, any subcontractors and subcontractors' employees or agents, for liability arising out of errors, omissions, or negligence in the performance of professional services in connection with the Project. Such insurance shall remain in full force and effect continuously for the period of design and construction of the improvements, and for a period of 3 years following substantial completion of Construction, provided that such coverage is reasonably available at commercially affordable premiums, as mutually determined and agreed. For the purposes of this section Professional Services has the meaning set forth in Section 103D-104 of the Hawai'i Revised Statutes.
- **12.1.6** Environmental Impairment Liability or Contractors Pollution Liability Insurance. The Lessee shall maintain, or cause its contractor to maintain, environmental impairment liability or contractors pollution liability insurance covering third-party injury and property damage claims, including cleanup costs, as a result of pollution conditions arising from the Premises, Lessee's operations or completed operations. Such requirement shall commence upon the Commencement of Construction for the Project, and terminate no less than three (3) years after the Completion of the Project. Lessor will be named as an additional insured. The limits of coverage will not be less than **\$2,000,000**.
- **12.1.7** <u>Insurance Coverage For Construction Phase</u>. Prior to commencing Construction or site preparation work, the Lessee shall procure or cause to be procured and maintain (as provided herein), all insurance to cover the demolition, construction and development activities under this Lease and the Development Agreement, that may be required by any Laws, in addition to the coverages specified in above, and the following types and amounts of insurance described below.
 - (a) <u>Builders Risk Insurance</u>. Lessee shall maintain builders risk insurance covering all risks of physical loss except those specifically excluded in the policy and shall insure at least against the perils of fire and extended coverage, theft, vandalism, malicious mischief, and collapse, earthquake, flood and windstorm. The insurance shall apply on a replacement cost basis. This insurance shall cover the entire work at the site, including all Improvements, materials and equipment, and reasonable compensation for architects' services and expenses made necessary by an insured loss. Insured property shall include portions of the work, materials and equipment located away from the site but intended for use at the site, and shall also cover portions of the work, materials, and equipment in transit. The policy shall include as insured property scaffolding, false work, and temporary buildings located at the site. The policy shall cover the cost of removing debris, including demolition. The insurance required shall name as insured, the Lessor, the Lessee and all subcontractors in the work.

The Lessee is responsible for paying any portion of any loss not covered because of the operation of any deductible or co-insurance provision applicable to the insurance required herein.

Waiver of Subrogation. The Lessor and Lessee waive all rights against each other and each of their contractors, subcontractors, officers, directors, agents and employees, for recovery of damages caused by fire and other perils to the extent covered by builders risk insurance purchased pursuant to the requirements of this paragraph or any other property insurance applicable to the work.

(b) <u>Boiler and Machinery Insurance</u>. Lessee shall maintain boiler and machinery insurance covering insured objects, including rooftop HVAC units and any separate heating units or boilers which serve the Premises, during installation and testing and until final acceptance, and including mechanical breakdown. Such coverage shall be for the full replacement value without deduction for depreciation. This insurance shall name as insured the Lessor, the Lessee, and all subcontractors in the work.

- (c) Lessee shall be responsible for any and all loss or damage to equipment, tools and other personal property, and may at its option purchase insurance to cover such property and equipment.
- **12.1.8** <u>Insurance Coverage For Operations</u>. Upon final acceptance of the Construction, and termination of the builder's risk insurance required in <u>Section 12.1.7(a)</u> above and prior to the use or occupancy of any of the Improvements, the Lessee shall procure or cause to be procured and maintain all insurance to cover operations under this Lease, that may be required by any Laws, and the following types and amounts of insurance described below:
 - (a) <u>Commercial Property Insurance</u>. Lessee shall maintain commercial property insurance covering the buildings, fixtures and equipment, with limits equal to the full replacement cost of the improvements. Such insurance shall at minimum, cover the perils against the perils of fire and extended coverage, theft, vandalism, malicious mischief, and collapse, earthquake, flood and windstorm.
 - (b) <u>Boiler and Machinery Insurance</u>. Lessee shall maintain boiler and machinery insurance covering the buildings, fixtures and equipment from loss or damage caused by an insured peril. Lessee shall be responsible for the amount of any deductible or co-insurance requirements applicable to the policy.
 - (c) Lessor shall be included as an insured and loss payee under the commercial property insurance and the boiler and machinery policies.
 - (d) Lessor and Lessee hereby waive any recovery of damages against each other (including their employees, officers, directors, agents or representatives) for loss or damage to the building, fixtures, equipment and any other personal property to the extent covered by the commercial property insurance or boiler and machinery insurance required herein. If either policy does not expressly allow the insured to waive rights of subrogation prior to loss, Lessor shall cause such policies to be endorsed with a waiver of subrogation as required above.
- **Nature of Insurance Program**. All insurance policies this Lease requires shall be issued by carriers that: (a) have a policyholders' rating of "A-, VIII" or better, based on the latest rating publication of Property and Casualty Insurers by A.M. Best Company (or its equivalent if such publication ceases to be published); and (b) are lawfully doing business in the State. Lessee may provide any insurance under a "blanket" or "umbrella" insurance policy, provided that (i) such policy or a certificate of such policy shall specify the amount(s) of the total insurance allocated to the Premises, which amount(s) shall equal or exceed the amount(s) required by this Lease and shall not be reduced for claims made for other properties; and (ii) such policy otherwise complies with this Lease.
- **12.3** <u>Policy Requirements and Endorsements</u>. All insurance policies this Lease requires shall contain (by endorsement or otherwise) the following provisions:
- **12.3.1** Contractor insurance. Lessee shall either: (a) include all contractors as insureds under all insurance set forth above, or (b) cause each contractor employed by Lessee to purchase and maintain insurance of the types specified above. When requested by the Lessor, Lessee shall furnish copies of certificates of insurance evidencing coverage for each contractor.
- **12.3.2** <u>Insureds.</u> Insurance policies shall identify Lessor as an "additional insured" using ISO form CG 20 10 (or equivalent), and during any construction activity the 1 CD 20 37 will also be required. Property

insurance policies shall name Lessor and Lessee as loss payees as their respective interests may appear, and each mortgagee this Lease allows under a standard noncontributing mortgagee clause. Notwithstanding anything to the contrary, all Property Insurance Proceeds shall be paid and applied as this Lease provides. On all insurance policies where Lessor is named as an additional insured, Lessor shall be an additional insured to the full limits of coverage purchased by Lessee even if those limits are in excess of those required under this Lease.

- **12.3.3** <u>Primary Coverage</u>. All policies shall be written as primary policies not contributing to or in excess of any coverage that Lessor may carry, and Lessee's insurers will not seek contribution from other insurance available to Lessor.
- **12.3.4** Contractual Liability. Liability Insurance policies shall contain contractual liability coverage, for Lessee's indemnity obligations under this Lease, to the extent covered by customary contractual liability insurance coverage. Lessee's failure to obtain such contractual liability coverage shall not relieve Lessee from any indemnity obligation under this Lease.
- **12.3.5** Severability of Interest. Liability Insurance policies shall contain a clause clarifying that, except with respect to coverage limits, the insurance applies separately to each insured and that the policy covers claims or suits by one insured against other, to the extent customarily covered by liability insurance policies.
- **12.3.6** <u>Notice to Lessor</u>. All policies required hereunder shall be written to provide not less than sixty (60) days prior Notice of cancellation or material change to Lessor.
- **12.4** Waiver of Certain Claims. Notwithstanding anything to the contrary contained in this Lease, Lessee and Lessor each waive any right of recovery against the other party and against any other party maintaining a policy of property insurance with respect to this Lease or the Premises, for any loss or damage sustained by Lessee or Lessor, as the case may be, that is covered by any policy of property insurance maintained (or required to be maintained under this Lease) with respect to the Premises, or the contents of the same or any operation in the Premises, whether or not such loss is caused by the fault or negligence of Lessee or its agents, directors, employees or officers, or is caused by the fault or negligence of Lessee or its agents, directors, employees or officers.
- 12.5 <u>Waiver of Subrogation</u>. The Lessee will require all insurance policies in any way related to the Lease and secured and maintained by the Lessee to include clauses stating each underwriter will waive all rights of recovery, under subrogation or otherwise, against the Lessor and all tiers of contractors or consultants engaged by either of them. The Lessee will require of contractors and consultants, by appropriate written agreements, similar waivers. If Lessee's policy of insurance relating to this Lease or to the Premises does not permit the foregoing waiver or if the coverage under such policy would be invalidated as a result of such waiver, Lessee shall obtain from the insurer under such policy a waiver of all right of recovery by way of subrogation against Lessor in connection with any claim, loss or damage covered by such policy.

12.6 Evidence of Insurance.

12.6.1 Upon execution of the Lease by Lessee, Lessee shall furnish Lessor with current certificate(s) of insurance, executed by a duly authorized representative of each insurer, certifying that at least the minimum coverages required herein are in effect and specifying that the liability coverages are written on an occurrence form and that the coverages will not be cancelled, non-renewed or materially changed by endorsement or through issuance of other policy(ies) of insurance without 60 days advance written notice to Lessor.

- **12.6.2** Prior to commencing work at the Premises, Lessee shall furnish Lessor with a certificate(s) of insurance, executed by a duly authorized representative of each insurer, showing compliance with the insurance required under <u>Section 12.1.7</u> above.
- **12.6.3** Upon final acceptance of the Construction, and prior to the use or occupancy of any Improvements, Lessee shall furnish Lessor with a certificate of insurance, executed by a duly authorized representative of each insurer, showing compliance with the insurance required under <u>Section 12.1.8</u> above.
- **12.6.4** Lessee shall provide certified copies of all insurance policies required above within 10 days of the Lessor's written request for said copies.
- **12.6.5** Failure of the Lessor to demand such certificate or other evidence of full compliance with these insurance requirements or failure of Lessor to identify a deficiency from evidence that is provided shall not be construed as a waiver of Lessee's obligations to maintain such insurance.
- **12.6.6** The acceptance of delivery by the Lessor of any certificate of insurance evidencing the required coverages and limits does not constitute approval or agreement by the Lessor that the insurance requirements have been met or that the insurance policies shown in the certificates of insurance are in compliance with the requirements of this Lease.
- **12.7** <u>Deductibles, Retentions and Co-Insurance</u>. Lessee is solely responsible for any loss or portion of loss not covered by the insurance required herein by reason of the application of any deductible, self-insured retention or co-insurance provision of the respective policy(ies), or due to policy limits or exclusions.

12.8 Failure to Maintain Insurance.

- **12.8.1** Failure to maintain the required insurance may result in a Default of this Lease at Lessor's option. If the Lessor is damaged by the failure of the Lessee to maintain insurance as required in this paragraph, then the Lessee shall bear all reasonable costs properly attributable to that failure.
- **12.8.2** Lessor shall have the right, but not the obligation, to prohibit Lessee or any of its contractors from entering the Premises until Lessee has provided certificates or other evidence that insurance has been placed in complete compliance with these requirements and such certificates have been approved by the Lessor.
- **12.8.3** If the Lessee fails to maintain the insurance as set forth herein, the Lessor shall have the right, but not the obligation, to purchase said insurance at Lessee's expense. In no event shall the Lessor be liable for payment of premiums due under any policy issued to the Lessee by reason of the Lessor being added as an 'insured' as required herein.
- **12.9** Additional Insurance. Lessor reserves the right to require additional kinds or amounts of insurance, as may be mutually agreed from time to time. Lessee shall periodically, but not less frequently than once every three (3) years, reevaluate the scope of risks covered and the limits of its insurance and, if commercially reasonable, increase such coverage or limits in order to provide coverage for Lessee's and Lessor's protection for risks and limits that a prudent business person would provide for property being put to uses similar to those of the Premises.
- **12.10 No Representation**. Lessor makes no representation that the limits of liability required to be carried by Lessee pursuant to this <u>Article 12</u> are adequate to protect Lessee. If Lessee believes that any of such insurance coverage is inadequate, Lessee shall obtain such additional insurance coverage as Lessee deems adequate, at Lessee's sole expense. No approval by Lessor of any insurer, or the terms or

conditions of any policy, or any coverage or amount of insurance, or any deductible amount shall be construed as a representation by Lessor of the solvency of the insurer or the sufficiency of any policy or any coverage or amount of insurance or deductible, or to limit Lessee's contractual obligations and liabilities, and Lessee assumes full risk and responsibility for any inadequacy of insurance coverage.

ARTICLE 13: LOSSES AND LOSS PROCEEDS

- **13.1 Notice**. If either party becomes aware of any Casualty or any actual, threatened, or contemplated Condemnation, then such party shall promptly Notify the other.
- 13.2 Effect of Casualty. If any Casualty occurs, then: (a) no Rent shall abate; (b) this Lease shall not terminate or be impaired; and (c) Lessee shall Restore with reasonable promptness regardless of cost or the amount of Property Insurance Proceeds subject to terms of this Section 13.2. Lessee shall be responsible, using its own or borrowed funds, to pay for any difference between the estimated cost to Restore the Improvements and the amount of the Property Insurance Proceeds provided that such Property Insurance Proceeds shall be received by Lessee and are approved for use by Leasehold Mortgagee for the Restoration as set forth in Section 13.3 below. If the Casualty occurs within the last ten (10) years of the Term then Lessee shall use commercially reasonable efforts to determine whether such Casualty is a Substantial Casualty within ninety (90) days after the occurrence of such Casualty, or within such additional time as may be reasonably required under the circumstances, and shall promptly Notify Lessor as to whether the Casualty is a Substantial Casualty ("Substantial Casualty Notice"). If (a) the Casualty is a Substantial Casualty, or (b) Leasehold Mortgagee does not approve the use of the Property Insurance Proceeds for the Restoration, then in either case, Lessee may, by Notice to Lessor (and with the prior written consent of the Limited Partner) terminate this Lease effective thirty (30) days after such Notice. In the event that Lessee elects to terminate the Lease pursuant to this Section 13.2, the Property Insurance Proceeds shall be disbursed in the following order of priority (a) first, to Lessee to pay the cost of removing the destroyed Improvements and all debris remaining from the Casualty and any other obligations prior to surrender of the Property under this Lease (b) second, to Lessee (subject to the rights of any Leasehold Mortgagees) up to the Market Value of the Leasehold Estate as of the date of the Casualty, and (c) then, to the extent, if any, of any remaining Property Insurance Proceeds, to Lessee (subject to the rights of any Leasehold Mortgagees) and Lessor, respectively, in the proportion that the duration of the Term remaining at the date of the Substantial Casualty (determined as if the Term were to continue until the Scheduled Expiration Date) bears to the duration of the full Term. For purposes of further clarification of said proportion to Lessee as used in subparagraph (c) in the immediately preceding sentence, said proportion to Lessee shall be a maximum of one-thirteenth based upon a 75-year term.
- **Adjustment of Claims; Use of Property Insurance Proceeds**. Unless Lessee has validly elected a Casualty Termination, Lessee shall have the sole right and authority to adjust any insurance claim, subject to rights of any Leasehold Mortgagee. The Property Insurance Proceeds shall be disbursed: (a) in the case of an Immaterial Loss, to Lessee, to be held in trust to be applied first for Restoration; and (b) in the case of any other Casualty, to Depository or the first-position Leasehold Mortgagee, if required by such Leasehold Mortgagee, to be released in installments for Restoration. To obtain each such disbursement, Lessee shall deliver to Depository or the first-position Leasehold Mortgagee; if required by such Leasehold Mortgagee:
- **13.3.1** Architect's Certificate. A certificate of Lessee's licensed architect, confirming that in such architect's professional judgment: (a) the sum then being requested is then properly due and payable to contractors, subcontractors, or other Persons for Restoration; (b) Restoration is proceeding in substantial compliance with the applicable plans and specifications and otherwise satisfactorily; (c) the sum being requested does not exceed the amount then due and payable; (d) except in the case of the final

disbursement of Restoration Funds, the remaining Restoration Funds after disbursement are reasonably anticipated to suffice to pay for the remaining Restoration yet to be performed; and (e) in the case of the final disbursement of Restoration Funds, Lessee has substantially completed Restoration and obtained a temporary certificate of occupancy for the Restoration to the extent required by Law, and delivered (or simultaneously delivers in exchange for payment) final lien waivers from all Persons otherwise entitled to claim a Prohibited Lien because of the Restoration;

- **13.3.2** <u>Lien Waivers</u>. Progress lien waivers for Restoration completed and paid for through the date of the preceding disbursement; and
- **13.3.3** Other. Such other documents, deliveries, certificates and information as Depository reasonably requires.
- **Substantial Condemnation**. If a Substantial Condemnation occurs, then this Lease (except as it relates to allocation of the Condemnation Award) shall terminate on the Condemnation Effective Date. Rent shall be apportioned accordingly. The Condemnation Award shall be paid in the following order of priority: (a) first, Lessee (subject to the rights of any Leasehold Mortgagees) shall receive such portion of the Condemnation Award up to the Market Value of the Leasehold Estate condemned at the Condemnation Effective Date, (b) second, Lessor shall receive such portion of the remaining Condemnation Award up to the Market Value of the Fee Estate condemned at the Condemnation Effective Date, and (c) third, to the extent of any remaining Condemnation Award, Lessee (subject to the rights of any Leasehold Mortgagees) shall receive the remaining balance of the Condemnation Award.
- 13.5 <u>Insubstantial Condemnation</u>. If an Insubstantial Condemnation occurs, then any Condemnation Award shall be paid to Depository to be applied first for Restoration in the same manner as Property Insurance Proceeds. Whether or not the Condemnation Award is adequate, Lessee shall, at its expense, Restore in the same manner as Restoration upon Casualty. Any Condemnation Award remaining after Restoration shall be distributed in the same manner as if it arose from a Substantial Condemnation that affected only the part of the Premises taken.
- **Temporary Condemnation**. A Temporary Condemnation shall not terminate this Lease or excuse Lessee from full performance of its covenants or any other obligations hereunder capable of performance by Lessee during the period of such Temporary Condemnation, but in such case, Lessee shall receive any Condemnation Award for the Temporary Condemnation (to the extent applicable to periods within the Term).
- **13.7** <u>Immaterial Loss</u>. If an Immaterial Loss occurs, then Lessee shall receive any Condemnation Award in trust to be applied first to Restoration. Lessee shall Restore in accordance with this Lease. After Restoration, Lessor shall receive any remaining Condemnation Award.
- **13.8** Surrender. If this Lease is terminated as a result of a Condemnation, Lessee shall surrender the Premises in accordance with the applicable surrender provisions of Article 23, and Lessor and Lessee shall thereafter be relieved of any further obligation under this Lease.

ARTICLE 14: LESSOR'S RESERVED RIGHTS

14.1 <u>Inspections</u>. Subject to notice requirements that are specifically provided herein, Lessor and its agents, representatives, and designees may enter the Premises in cases of emergency, or upon reasonable Notice during regular business hours, to: (a) ascertain whether Lessee is complying with this Lease (including the review of Lessee's records, contracts and/or Subleases pertaining to the Premises); (b) cure

Lessee's Defaults, in accordance with this Lease; (c) inspect the Premises and any Construction; (d) reserved; (e) post notices of non-responsibility in accordance with this Lease; or (f) as reasonably required in connection with any sale, re-entitlements, or for other reasonable purposes determined by Lessor. In entering the Premises, Lessor and its designees shall not unreasonably interfere with operations on the Premises (provided that any such entry by Lessor into a Residential Rental Unit shall comply with advance written notice requirements under applicable Law) and shall comply with Lessee's reasonable instructions. If Lessor wishes to perform such tests, borings, and other analyses as Lessor determines may be necessary or appropriate relating to (non)compliance with any Law or possible Hazardous Substances Discharge, or to clean up any Pre-Existing Hazardous Substances Condition, Lessor shall Notify Lessee of such request, and Lessor and Lessee shall work in good faith to agree upon a schedule for Lessor to conduct such tests, borings, and other analyses that shall not adversely interfere with or disrupt Lessee's Construction schedule, or Lessee's Permitted Use of the Premises, including but not limited to, Lessee's tenants and subleases at the Project and upon completion, Lessor shall promptly restore the Premises to the condition existing prior to such tests, borings, and other analyses, as applicable.

- **Other Entries by Lessor**. In addition to Lessor's rights under Section 14.1, Lessor may enter the Premises at any time (a) to take possession due to any breach that has been without cure under any applicable cure period of any Event of Default under this Lease in the manner provided herein, subject to the rights of the Leasehold Mortgagee; and (b) to perform any covenants of Lessee that Lessee fails to perform (subject to any applicable notice and cure periods and the rights of the Leasehold Mortgagee). Lessor may make any such entries hereunder without abatement of Rent, and may take such reasonable steps as required to accomplish the stated purposes. In an emergency, Lessor shall have the right to use any means that Lessor reasonably deems proper to open the doors to the Improvements. Any entry into the Premises by Lessor in the manner hereinbefore described shall not be deemed to be a forcible or unlawful entry into, or a detainer of, the Premises, or an actual or constructive eviction of Lessee from any portion of the Premises (provided that any such entry by Lessor into a Residential Rental Unit shall comply with advance written notice requirements under applicable Law). No provision of this Lease shall be construed as obligating Lessor to perform any repairs, alterations or improvements except as otherwise expressly agreed to be performed by Lessor herein.
- **14.3** Water, Oil, Gas and Mineral Rights. Subject to applicable Laws, Lessor reserves to itself the sole and exclusive right to all water, oil, gas, or other hydrocarbon or mineral substances and accompanying fluids, including all geothermal resources, from the Land; but Lessor shall not undertake any extraction of such resources during the Term.

14.4 Easements.

- **14.4.1** Lessor's Right to Grant Licenses, Permits, Encroachments and Easements. Subject to Section 14.4.2, Lessor shall have the right, without payment to or charge from Lessee, to reserve to itself and to grant licenses, permits, encroachments or easements (collectively, "Easements") to any Person or Government on, over, under, across and through the Premises, to the extent such Easements are deemed by Lessor to be necessary or convenient for the construction, installation, operation, maintenance, repair and replacement of (a) improvements, (b) underground or overhead lines and other transmission facilities and appurtenances for electricity, gas, telephone, water, sewage, drainage and any other service or utility, (c) rights of way, curbs, pavements and other roadway improvements, and (d) landscaping, whether serving the Premises or other properties of Lessor or any other Person or Government.
- **14.4.2** <u>Limitations on Lessor's Rights</u>. Notwithstanding anything contained in above, Lessor may only grant or relocate Easements if taking such action will not have a material adverse effect on the design or use of Improvements and Common Area Amenities. Further, Lessor shall (or shall cause the grantee of such Easements to): (a) be responsible for payment of all costs in connection with the granting of such Easements and the construction, installation and restoration work in connection therewith; (b) carry out and

coordinate such work with Lessee so as to minimize disruption with Lessee's use of the Premises; and (c) following any work relating to any Easement, restore the Premises to substantially the same condition as existed prior to such work. If Lessor fails to perform its obligations under this <u>Section 14.4.2</u>, Lessee shall first Notify Lessor of such non-performance and the parties will negotiate in good faith to mutually agree upon the appropriate restoration work.

- **<u>Development</u>**. Lessee agrees that it will reasonably cooperate with Lessor in the event Lessor desires to develop roads, sidewalks or other public improvements owned by Lessor in the vicinity of the Premises; provided that such cooperation has no material adverse effect on the Premises or Lessee's use of the Premises, and such cooperation is at no cost to Lessee (unless otherwise approved by Lessee).
- **No Light or Air Easement**. Any diminution or shutting off of light or air by any structure now existing or hereafter erected by or on behalf of Lessor or Lessor's Affiliate on lands adjacent to the Premises shall in no way affect this Lease, shall not constitute a constructive eviction or grounds for reduction or abatement of Rent, or otherwise impose any liability on Lessor.
- **14.7 General**. In addition to Lessor's right, title and interest as the fee owner in the Land, Lessor also shall have such rights as it may have as grantee, benefitted party or other party other than the fee owner of the Land under any grants of easement, covenants, restrictions, Land Court orders, and other recorded instruments or maps encumbering or affecting the Land (for example and not by way of limitation, rights of Lessor as grantee under sewer or drainage easements on, through or under the Land).

ARTICLE 15: LESSOR'S TRANSFERS

- 15.1 <u>Transfer of Lessor's Interest</u>. Lessee acknowledges that, following the expiration of the Compliance Period, Lessor has the right to Transfer the Fee Estate in accordance with <u>Section 15.3</u>. Lessee agrees that in the event of any such Transfer, (a) Lessee shall look solely to such transferee for the performance of Lessor's obligations under this Lease after the date of Transfer, and such transferee shall be deemed to have fully assumed and be liable for all obligations of this Lease to be performed by Lessor after the date of Transfer; and (b) Lessee shall attorn to such transferee. Lessee hereby agrees to cooperate, at no cost to Lessee, with Lessor in connection with any Transfer; provided that Lessee's rights and obligations under this Lease shall not be amended in any material respect as a result of or in connection with any such Transfer. Following any such Transfer, Lessor shall deliver to Lessee a copy of the instrument evidencing the transferee's assumption of liabilities and obligations of Lessor under the Lease.
- **Release of Lessor**. Upon any Transfer of the entire Fee Estate in compliance with this Lease, the grantor automatically shall be freed and relieved from all liability (excluding liability previously accrued) for performance of any covenants or obligations to be performed by Lessor after the Transfer, provided that such successor Lessor assumes in writing Lessor's present and future obligations under this Lease. This Lease shall bind Lessor only while Lessor owns the Fee Estate, except as to any liabilities and obligations accrued before the date of Transfer of the Fee Estate.
- 15.3 No Right of First Refusal/No Option to Purchase. If Lessor desires to Transfer the Fee Estate during the Term following the expiration of the Compliance Period, it shall do so in full compliance with all Laws governing the County's sale of real property including, if applicable, a request for proposal or other public bidding process. If a request for proposal or other public bidding process is utilized or required, Lessor shall give Lessee written notice of Lessor's intent to Transfer the Fee Estate at least thirty (30) days prior to issuing such request for proposal or commencing such public bidding process. Lessee understands and agrees that nothing in this Lease grants Lessee an option or right of first refusal to purchase the Fee Estate from Lessor.

ARTICLE 16: REGULATORY PROVISIONS

16.1 Reserved.

16.1.1 Reserved.

ARTICLE 17: LESSEE'S TRANSFERS

Lessee's Transfer Right. Except as provided in Article 18 and Article 19, Lessee shall not Transfer this Lease or the Leasehold Estate, whether or not to an Affiliate, without the prior written consent of Lessor, which consent shall not be unreasonably withheld, conditioned, or delayed. Lessee agrees, however, that it shall be conclusively presumed to be reasonable for Lessor to withhold its consent unless the following have occurred: (A) Lessor's receipt of reasonably satisfactory evidence that: (a) Lessee is not in Default under this Lease or, if Lessee is in Default, that the transferee undertakes to cure any such Default to the reasonable satisfaction of Lessor; (b) the continued operation of the Premises after the Transfer shall comply with the provisions of this Lease; (c) the transferee has the financial capability and resources to operate and maintain the Premises as required by this Lease; (d) either (i) the transferee or its property manager has the experience, reputation, managerial and operational skills to operate and maintain the Premises, (ii) the transferee agrees to retain a property manager with the skills, experience and record described in clause (i) above, effective as of the date of the Transfer, or (iii) the transferor Lessee or its property manager will continue to manage the Premises, or another property management company reasonably acceptable to Lessor will manage the Premises, for at least one year following the Transfer; (e) the transferee is not delinquent in any tax payments and does not have pending against it any charges of, and does not have a record of, material building code violations or complaints concerning the maintenance, upkeep, operation, and regulatory agreement compliance of any of its projects as identified by any applicable local, state or federal regulatory agencies; and (f) the transferee is not in arrears or noncompliance with any obligations to the Government and is in good standing with respect to other Government agreements; (B) the execution by the transferee and delivery to Lessor of an assignment document specifically stating that the Transfer is made subject to all terms, covenants and conditions of this Lease, and all such terms, covenants, and conditions of this Lease shall be specifically assumed and agreed to by the transferee, along with such other documents reasonably requested by Lessor in connection with the Transfer but which do not expand, increase, diminish or otherwise modify the rights, duties and obligations of the "lessee" under this Lease; and (C) receipt by Lessor of all fees and/or expenses then currently due and payable to Lessor by Lessee as set forth in this Lease. It is hereby expressly stipulated and agreed that any Transfer in violation of this Section 17.1 shall be null, void and without effect, shall cause a reversion of title to Lessee, and shall be ineffective to relieve Lessee of its obligations under this Lease. The written consent of Lessor to any Transfer of this Lease or the Leasehold Estate shall constitute conclusive evidence that the Transfer is not in violation of this Section 17.1. Upon any Transfer by Lessee that complies with this Lease, Lessee shall be fully released from its obligations hereunder to the extent such obligations have been fully assumed in writing by the transferee except for: (x) any obligation to hold and apply Restoration Funds held by Lessee at the date of the Transfer (unless transferred to the transferee); (y) any unperformed obligations that arose or accrued prior to such Transfer and all Legal Costs of any proceeding relating thereto commenced before such Transfer for which the transferor is liable hereunder (unless specifically assumed in writing by the transferee); and (z) any indemnity obligation under this Lease (unless specifically assumed in writing by the transferee). Lessee shall pay all transfer and other taxes, if any, payable on account of any Transfer by Lessee or any holder of any Equity Interest in Lessee. Notwithstanding any provision to the contrary in this Lease, Lessee may, with the prior consent of Lessor, which consent shall not be unreasonably withheld, conditioned, or delayed, Transfer the Leasehold Estate and this Lease to an Affiliate of Lessee; provided that (a) Lessee shall

provide to Lessor a copy of the assignment and assumption agreement pursuant to which such Affiliate transferee shall assume and agree to be bound by and to observe and perform the covenants, duties and obligations of "lessee" under this Lease from and after the effective date of such assignment and assumption; and (b) Lessee provide Lessor with documentation reasonably evidencing that the Affiliate transferee qualifies as an "Affiliate" as defined under Article 1. Notwithstanding the foregoing, (a) any of the following shall not be considered a "Transfer" and shall not require Lessor's consent (provided that, in each instance, Lessor has received Notice thereof): (i) the exercise by the Limited Partner of its rights pursuant to the Partnership Agreement to remove the general partner of Lessee upon the existence of an event of default by the general partner, after the giving of written notice and expiration of any applicable cure period, and appoint the Limited Partner or an Affiliate thereof as interim general partner of Lessee; (ii) the exercise by the Limited Partner of its right to enforce any obligation of the general partner to repurchase the Limited Partner's partnership interest in Lessee as set forth in the Partnership Agreement; and/or (iii) a transfer by the Limited Partner of its partnership interest in Lessee (including transfers within a limited partner's interest of Lessee) in accordance with the terms of the Partnership Agreement, and (b) any amendment to the Partnership Agreement to effectuate any of the permitted transfers in the preceding clause (a) of this sentence shall not require Lessor's prior written consent.

- **17.2 No Partial Transfers**. Except in the case of a Sublease permitted pursuant to <u>Article 18</u> or a Leasehold Mortgage permitted pursuant to <u>Article 19</u>, in no event shall Lessee be permitted to Transfer less than its entire interest in this Lease or the Leasehold Estate, and Lessor may elect in its sole discretion to deny consent to any such partial Transfer.
- **Notice of Transfer**. If Lessee desires Lessor's consent to any Transfer, Lessee shall Notify Lessor in writing, which notice shall include (a) the proposed effective date of the Transfer; (b) the material terms of the proposed Transfer; (c) a copy of the signed purchase and sale agreement or other agreement between Lessee and the proposed transferee; (d) current financial statements of the proposed transferee certified, compiled or reviewed by an independent certified public accountant for the fiscal year most recently ended, and business credit, personal references and business history of the proposed transferee; and (e) such other reasonable information in connection with the proposed Transfer as Lessor shall reasonably request.
- **17.4 Expenses**. Within ten (10) days following demand, Lessee shall reimburse Lessor for Lessor's reasonable costs (including attorneys' fees) incurred in reviewing and approving or disapproving, or otherwise consulting with respect to, any Transfer.

ARTICLE 18: SUBLEASES

18.1 Residential Tenants.

- **18.1.1** Residential Spaces. Lessee acknowledges and agrees that the Premises are to be held, developed, managed and operated as a "housing project" within the meaning of Section 8-10.20 of the Revised Ordinances of Honolulu (ROH), as amended, in accordance with and subject to the provisions of the Development Agreement and this Lease regarding Residential Rental Units for Qualified Tenants at Affordable Rents and appurtenant facilities.
- **18.1.2** <u>Representations, Warranties, and Covenants</u>. The Lessee further represents, warrants and covenants as follows:
 - (a) Each Residential Rental Unit shall be a Dwelling Unit as defined in ROH §8-10.20(a). They will be and remain similarly constructed, and each Residential Rental Unit will

contain facilities for living, sleeping, eating, cooking and sanitation for a single person or a family which will be and remain an independent living unit complete, separate and distinct from other residential units and will include and will continue to include a sleeping area, at least one bathroom, and a single kitchen (as defined in ROH § 8-10.20(a)) equipped with a cooking range, oven, range hood, water heater, refrigerator, and sink. The Premises shall not be used as a hotel, motel, dormitory, fraternity or sorority house, rooming house, hospital, nursing home, sanitarium, or rest home, or in any manner in contravention of applicable law.

- (b) The Residential Rental Units shall at all times comply with all Federal and State Affordable Housing Requirements applicable thereto, including requirements established by such programs affecting both income limitations and allowable rent levels in respect of such residential units; provided, however that no Qualified Tenant in any Residential Rental Unit shall be required to make a payment toward rent in excess of the limits established by this Lease for such Residential Rental Unit. In the event of a conflict between the requirements of this Lease and any Federal and State Affordable Housing Requirements, the latter shall control. Upon the expiration of any Federal and State Affordable Housing Requirements, the requirements of this Lease shall remain in place and shall control for the balance of the Term. Nothing in this Lease shall preclude or be interpreted or deemed to prohibit Qualified Tenants from applying for and receiving rent and other subsidies that shall, as applicable under any program concerning such rent or other subsidies, subsidize or be in addition to the Affordable Rents specified in this Lease.
- (c) For the Term, subject to any applicable Federal and State Affordable Housing Requirements, all of the Residential Rental Units shall at all times be rented to and occupied by Qualified Tenants at the Affordable Rents specified herein.
- (d) Lessee shall establish a system to determine preferences by lot in the event the number of eligible applicants exceeds the number of Residential Rental Units available. The Residential Rental Units shall otherwise be rented or held available for rental, on a first-come first-served basis to members of the general public who are Qualified Tenants, on a continuous basis, and may not be converted to owneroccupied condominium units or other non-rental use. In renting Residential Rental Units to Qualified Tenants, the Lessee will not otherwise give preference to any particular protected class or group except Lessee may grant preferences for the following:
 - (1) Persons displaced by government action, provided that their tenant application is submitted between (a) the date of the initial displacement notice and (b) six (6) months after the actual displacement action, supported by sufficient documentation of such displacement;
 - (2) For residential units designated as accessible for persons with mobility, visual, hearing and/or mental impairment, households containing at least one person with such impairment will have first priority for those units. No special priority shall be given among such applicants based upon the specific type(s) of impairment of the household member; and,
 - (3) As may be required by any applicable Federal and State Affordable Housing Requirements and/or this Lease.
- (e) Lessee shall rent the units in accordance with a rental program approved by the Lessor, which rental program may be amended in writing upon mutual agreement of

Lessee and the Lessor from time to time, and which shall be incorporated by reference herein and binding upon Lessee. Such rental program shall establish:

- Requirements and procedures for reviewing applications, evaluating applicants, determining eligibility as a Qualified Tenant, selecting renters and renting to Qualified Tenants.
- (2) A system for determining preferences by lot, in the event the number of Qualified Tenants exceeds the number of Residential Rental Units available.
- (3) A description of how the rental program will ensure compliance with all applicable laws and standards relating to the rental of real property.
- (4) Affordable Rent for each Residential Rental Unit, based on unit type, household size, and income level.
- (5) In establishing the Affordable Rent, the HUD AMI for the following household sizes shall apply, based on the Residential Rental Unit type to be rented:

Unit Type Household Size Studio 1.5 persons 1 bedroom 1.5 persons

- (f) For the Term, the Lessee shall obtain, complete and maintain on file (i) at the time of initial occupancy of any Residential Rental Unit, (ii) upon the vacancy and reoccupancy of any Residential Rental Unit, and (iii) at least once annually, an income certification, in a form approved by the Lessor, an example of which is attached hereto as **Exhibit C**, which shall be subject to independent investigation and verification by the Lessor. The Lessee shall verify the information submitted by each Qualified Tenant at the time of submission, including taking, for other than Section 8 of the Housing Act certificate or voucher holders, the following steps as part of the verification process by the Lessee: (1) either (A) obtain a federal income tax return for such Qualified Tenant for the most recent tax year; and/or (B) obtain a written verification of employment from such Qualified Tenant's current employer; or (2) if such Qualified Tenant is not employed and has no tax return, obtain other verification of such Qualified Tenant's source of income. The Lessee shall file with the Lessor a certificate of continuing program compliance in a form approved by the Lessor, an example of which is attached hereto as Exhibit D, on or before the first (1st) day of each September during the Term setting forth the required information for the preceding calendar or fiscal year. The books and records of the Lessee pertaining to the incomes of Qualified Tenants must be open to inspection by any authorized representative of the Lessor. During any period that any Federal and State Affordable Housing Requirements are in effect, Lessee shall be deemed to have satisfied the requirements of this subparagraph if Lessee obtains, completes and maintains on file income certification and other forms that comply with the applicable Federal and State Affordable Housing Requirements and that contain substantially the same information, and certifications of compliance with such Federal and State Affordable Housing Requirements that establish compliance with the requirements herein.
- (g) All subleases shall be subordinate to this Lease and shall contain clauses, among others, wherein each Qualified Tenant: (i) certifies the accuracy of the statements made regarding household income and (ii) agrees that the family income and other eligibility requirements shall be deemed substantial and material obligations of the

tenancy of such tenant, that such Qualified Tenant will comply promptly with all requests for information with respect thereto from the Lessee, and that the failure to provide accurate income information or refusal to comply with a request for information shall be deemed a violation of a substantial and material obligation of the tenancy of such tenant.

- (h) In connection with all subleases of the Residential Rental Units, tenants shall be provided written disclosures of pre-existing uses in the surrounding community, including activities such as:
 - (1) Cultural and religious events at the nearby Church;
 - (2) Public library;
 - (3) 'Aiea Intermediate School
- 18.1.3 Affordable Rents. "Affordable Rent" means a monthly rent for a Residential Rental Unit which does not exceed the maximum rent established for persons with the annual income level (as adjusted for household size) necessary to qualify for that Residential Rental Unit, which shall be based upon guidelines published by the Hawaii Housing Finance and Development Corporation based on income levels and limits established by HUD, adjusted for household income, Unit and family size, and utility allowances. Affordable Rents shall be inclusive of tenant utility payments. Notwithstanding the foregoing, with respect to Qualified Tenants, if any, who are recipients of rent subsidies pursuant to Section 8 of the Housing Act (or any rent subsidy or other HUD, State, or County program), if the contract rent for a Residential Rental Unit under such program is greater than the Affordable Rent, the rent for such Residential Rental Unit may be such contract rent.
- 18.1.4 Non-discrimination. Lessee shall comply with the provisions of any applicable federal, State or local law prohibiting discrimination in housing on the basis of race, creed, color, sex, familial status, marital status, religion, national origin, age (except as to age, as may be expressly provided herein) or any other prohibited basis. Lessee shall also comply with the provisions of any applicable federal, State or local law prohibiting discrimination on the basis of race, creed, color, sex, familial status, marital status, religion, national origin, age, or any other prohibited basis, in connection with the employment or application for employment of persons for the Construction, operation and management of the Premises. The Lessee further agrees not to refuse to lease a Residential Rental Unit offered for rent, or otherwise discriminate in the terms of tenancy, solely because any tenant or prospective tenant is the holder of a certificate or a voucher under Section 8 of the Housing Act, or any successor legislation.
- **18.1.5** Condition of the Premises. The Lessee agrees that throughout the Term, it shall (1) maintain the Premises in good repair and condition in accordance with applicable County codes, and the Uniform Physical Condition Standards set forth in 24 CFR Part 5, Subpart G, as amended; (2) maintain and operate the Premises to provide decent, safe and sanitary housing, including the provision of all essential and appropriate services, maintenance and utilities; and (3) comply with the lead-based paint regulations set forth in 24 CFR Part 35, as amended.
- 18.1.6 Covenants to Run with the Property -- Termination of Covenants. The Lessor and the Lessee hereby covenant and agree that the covenants set forth herein that govern the use and occupancy of the Premises shall be and are covenants running with the Land for the Term and shall be binding upon all subsequent Lessees of the Premises for such Term, and are not merely personal covenants of the Lessor and the Lessee. The Lessee hereby agrees that any and all requirements of State Laws to be satisfied in order for the provisions of this Lease to constitute restrictions and covenants running with the Land shall be deemed to be satisfied in full, and that any requirements of privileges of estate are intended

to be satisfied, or, in the alternate, that an equitable servitude has been created to ensure that these restrictions run with the Land. For the Term each and every contract, deed or other instrument hereafter executed conveying the Premises or portion thereof shall expressly provide that such conveyance is subject to this Lease; provided, however, the covenants contained herein shall survive and be effective regardless of whether such contract, deed or other instrument hereafter executed conveying the Premises or portion thereof provides that such conveyance is subject to this Lease.

18.1.7 Real Property and General Excise Tax Exemptions. Lessee shall be responsible for paying and discharging all real property taxes for the Premises payable or accruing during the Term. Based on the recordation of this Lease, Lessee may file an application or claim for exemption from the assessment and payment of real property taxes, including any claim for exemption subject and pursuant to ROH §8-10.20 and §8-10.21, relating to low-income rental housing, and the County agrees to process such application in the ordinary course subject and pursuant to the requirements of §8-10.20 and §8-10.21. Lessee understands that Lessee must file for such exemption annually and that the County shall not be responsible or liable for Lessee's failure to timely file for such exemption on an annual basis.

Pursuant to Hawai'i Revised Statutes Section 46-15.1, Section 201H-36, and Section 237-29, the County may certify for exemption from general excise taxes any projects which meet the requirements of said statutes. The County will work with Lessee to issue such certifications provided that the requirements under said statutes and applicable laws are met. Lessee understands that Lessee is responsible for preparing and filing any exemption request and that the County shall not be responsible or liable for Lessee's failure to file for such exemptions. Lessee further understands that such exemptions are subject to all laws and rules applicable thereto.

The exemptions from real property taxes and general excise taxes currently available are subject to change by legislative or administrative action.

18.2 Commercial Tenants. [RESERVED]

- **18.3 Recordkeeping.** The financial records of the Premises are to be maintained by the Lessee in accordance with recognized industry-accepted accounting principles consistently applied. In addition to records and information required to be collected, prepared, maintained, and reported pursuant to federal and State laws and regulations and this Lease, the Lessee shall provide the following to Lessor at Lessor's reasonable request:
 - access to the books of account for the Premises;
 - records pertaining to the Residential Rental Units;
 - annual operating statements for the Project;
 - audited financial statements for the Project; and
 - annual operating budget for the Project.

ARTICLE 19: LEASEHOLD MORTGAGES

19.1 Leasehold Mortgage. Notwithstanding anything in this Lease to the contrary, Lessee shall have the absolute and unconditional right, without Lessor's consent, to execute and deliver a Leasehold Mortgage at any time and from time to time during the Term. The execution and delivery of a Leasehold Mortgage shall not be deemed to constitute such an assignment or transfer of this Lease that would require Lessee to obtain Lessor's consent under Article 17 of this Lease. Foreclosure of the Leasehold Mortgage (or any bona fide sale or assignment in lieu of foreclosure), whether by judicial proceedings or by virtue of any power contained in the Leasehold Mortgage, shall not require the consent of Lessor and shall not

constitute a breach of any provision or a Default under this Lease. Upon such foreclosure, sale or conveyance and provided that (a) such purchaser or assignee expressly assumes and agrees to be bound by and to observe and perform all covenants and obligations of the "lessee" under this Lease from and after the effective date of such conveyance, and (b) a copy of such fully executed assignment and assumption agreement is provided to Lessor, Lessor shall recognize such purchaser or assignee (or Leasehold Mortgagee or its designee) as the lessee hereunder. Any such purchaser or assignee (other than the Leasehold Mortgagee or its designee) who has acquired title to this Lease by way of foreclosure or assignment in lieu thereof may only assign its rights under this Lease, other than by way of mortgage, in compliance with Article 17. Lessor shall not be required to join in, or "subordinate the Fee Estate to," or to subordinate Lessor's interest under this Lease to, any Leasehold Mortgage, but shall execute and deliver such estoppel certificates, and non-disturbances and other certifications as any Leasehold Mortgagee shall reasonably require.

- **19.2** <u>Protection of Leasehold Mortgagee</u>. In the event Lessee subjects the Leasehold Estate to a Leasehold Mortgage, the provisions of this Article 19 shall apply with respect to such Leasehold Mortgage:
- 19.2.1 Concurrent Notices. Lessor shall, upon serving Lessee with any Notice pursuant to the provisions of this Lease, or other communication which may adversely affect the security of a Leasehold Mortgagee, including but not limited to, any Notice of Default, concurrently serve a copy of the Notice or other communication upon each Leasehold Mortgagee. Upon the execution and recording of a Leasehold Mortgage, such Leasehold Mortgagee shall send to Lessor a true copy of the Leasehold Mortgage with the recordation information noted, together with written Notice specifying the name and address of such Leasehold Mortgagee, or at such other address as such Leasehold Mortgagee may provide to Lessor by written Notice from time to time. Lessor shall be entitled to rely upon the last address of a Leasehold Mortgagee that was provided to Lessor by written Notice by such Leasehold Mortgagee.
- 19.2.2 Right to Cure. Each Leasehold Mortgagee and each Limited Partner shall have the right, but not the obligation, at any time prior to termination of this Lease and without payment of any penalty, to pay all Rent due hereunder, to effect any insurance, to pay any taxes or assessments, to make any repairs or improvements, to do any other act or thing required of Lessee under this Lease, and to do any act or thing which may be necessary and proper to be done in the performance and observance of the agreements, covenants and conditions of "lessee" under this Lease to prevent termination of this Lease; provided that no Leasehold Mortgagee shall have any duty, obligation, or liability under this Lease prior to the time of its entry and physical possession of the Premises or acquisition of the Leasehold Estate. Any of the foregoing done by Leasehold Mortgagee or the Limited Partner, as applicable, shall be effective to satisfy the obligation of Lessee under this Lease and Lessor shall accept such performance with the same force and effect to prevent a termination of the Lease as if the same had been done by Lessee. No action by Lessee or Lessor to voluntarily cancel or surrender (except in accordance with the terms herein) or materially modify the terms of this Lease or the provisions of this Article 19 shall be binding upon a Leasehold Mortgagee or Limited Partner without, in each case, its prior written consent. Notwithstanding any other provision to the contrary in this Lease, the Limited Partner shall have the same cure period after the giving of a Notice as provided to Lessee under Article 22, plus an additional period of thirty (30) days; provided that if such cure cannot be completed within such 30-day period, this cure period shall be extended so long as such Limited Partner thereafter diligently pursues such cure to completion.
- 19.2.3 <u>Cure Period</u>. Notwithstanding anything in this Lease to the contrary, if any Event of Default shall occur and be continuing beyond any applicable cure period which, pursuant to any provision of this Lease, entitles Lessor to terminate this Lease or exercise any other remedies hereunder, Lessor shall have no right to terminate this Lease or to exercise any other remedies hereunder unless (a) Lessor shall have given written notice to the Leasehold Mortgagee of such Event of Default and stating Lessor's intent to terminate this Lease or to exercise any other remedies hereunder ("Lessor's Notice"); and (b) the occurrence of any of the following: (i) in the event of a Monetary Default, cure shall not have been made or

commenced for such Monetary Default within thirty (30) days after the service of the Lessor's Notice, or (ii) in the event of any Nonmonetary Default under this Lease susceptible of being cured by Leasehold Mortgagee, no cure of such Nonmonetary Default is commenced within sixty (60) days after the service upon the Leasehold Mortgagee of such Lessor's Notice and diligently pursued to completion of the cure of such Nonmonetary Default; provided, however, that in the Event of a Default under this Lease which consists of the existence or nonpayment of a lien, such Default shall be deemed to be cured if, within such 60-day period, the Leasehold Mortgagee shall have commenced foreclosure and shall thereafter diligently pursue such proceedings to completion, or shall have commenced and shall thereafter diligently pursue steps to obtain title to the Leasehold Estate by means of an assignment in lieu of foreclosure. If any such Default susceptible of being cured by the Leasehold Mortgagee cannot be cured by the Leasehold Mortgagee without the Leasehold Mortgagee first obtaining possession of the Premises or title to the Leasehold Estate or if the Default is not susceptible of being cured by the Leasehold Mortgagee, such Default shall be deemed to be cured if: (A) within sixty (60) days after the receipt by the Leasehold Mortgagee of such written notice, the Leasehold Mortgagee shall have commenced foreclosure and thereafter diligently pursues such proceedings to completion, or (B) the Leasehold Mortgagee commences, within such 60-day period, and thereafter diligently pursues, steps to obtain title to the Leasehold Estate by means of an assignment in lieu of foreclosure. During the course of any such proceedings, such Leasehold Mortgagee shall pay or cause to be paid all Rent as and when the same becomes due and payable under this Lease and shall perform all other obligations of the Lease reasonably susceptible of being performed by Leasehold Mortgagee (but excluding any Personal Default of Lessee).

19.2.4 <u>Time Extensions</u>. If the Leasehold Mortgagee is prohibited from commencing or prosecuting foreclosure or other appropriate proceedings in the nature thereof by any process or injunction issued by any court or by reason of any action by any court having jurisdiction over any bankruptcy or insolvency proceeding involving Lessee, the times specified in <u>Section 19.2.3</u> for commencing or prosecuting foreclosure or other proceedings shall be extended for the period of the prohibition, provided that no Monetary Default under this Lease shall be continuing.

19.2.5 New Lease. Lessor agrees that, in the event of termination of this Lease for any reason (including, but not limited to, any Default by Lessee or by reason of the disaffirmance or reject hereof by Lessee or any receiver, liquidator, or trustee of Lessee or its property), Lessor, if requested by the Leasehold Mortgagee, will enter into a new lease of the Premises ("New Lease") with the most senior Leasehold Mortgagee requesting a new lease, which new lease shall commence as of the date of termination of this Lease and shall run for the remainder of the Term, at the Rent and upon the same terms, covenants and conditions herein contained and subject to the rights, if any, of the parties then in possession of any part of the Premises (including any Sublessee of a Residential Rental Unit), provided that: (a) such Leasehold Mortgagee shall make written request upon Lessor for the new lease within sixty (60) days after the date such Leasehold Mortgagee receives written Notice from Lessor that the Lease will be terminated; (b) such Leasehold Mortgagee shall pay to Lessor at the time of the execution and delivery of the New Lease any and all sums, including Rent, which would, at that time, be due and unpaid pursuant to this Lease but for its termination, and in addition thereto all reasonable expenses, including reasonable attorneys' fees, which Lessor shall have incurred by reason of such termination; (c) such Leasehold Mortgagee shall cure any Default which Lessee under the terminated Lease was obligated to cure under its terms, to the extent the same are reasonably susceptible of being cured by the Leasehold Mortgagee; and (d) the lessee under the New Lease shall perform and observe all covenants which Lessee was obligated to perform and observe under the terminated Lease, and shall have the same right, title and interest in and to the Premises as Lessee had under the terminated Lease immediately prior to its termination. Notwithstanding any provision to the contrary in this Lease, any New Lease made pursuant to this Section 19.2.5 shall be prior to any Leasehold Mortgage or other lien, charge or encumbrance on the Premises, to the same extent as the terminated Lease, and shall be accompanied by a conveyance of title to the existing Improvements (free of any mortgage, lien, change or encumbrance created by Lessor) for a term of years equal to the remaining term of this Lease, subject to the reversion in favor of Lessor upon

expiration or sooner termination of the New Lease. The right to a New Lease pursuant to this <u>Section 19.2.5</u> shall survive the termination of this Lease. If a Leasehold Mortgagee shall elect to demand a New Lease under this <u>Section 19.2.5</u>, Lessor agrees, at the request of, on behalf of, and at the sole cost and expense of the Leasehold Mortgagee, to institute and pursue diligently to conclusion any appropriate legal remedy or remedies to oust or remove Lessee from the Premises, as designated by the Leasehold Mortgagee. Unless and until Lessor has received notice from all Leasehold Mortgagees that the Leasehold Mortgagees elect not to demand a New Lease as provided in this <u>Section 19.2.5</u>, or until therefor has expired, Lessor shall not cancel or agree to the termination or surrender of any existing subleases (except if, after the giving of written notice to such subtenant and all Leasehold Mortgagees and an opportunity to cure such default as set forth in such sublease, a default has occurred and is continuing beyond any applicable cure period under an existing sublease) nor enter into any new sublease hereunder without the prior written consent of the Leasehold Mortgagees. Notwithstanding the foregoing, nothing herein contained shall require any authorized Leasehold Mortgagee to enter into a New Lease pursuant to this <u>Section 19.2.5</u>, nor to cure any Default of Lessee referred to above.

- 19.2.6 Lessor's Consent. Pursuant to Section 19.1, Lessor's written consent, which shall not be unreasonably withheld, is required for any assignment or transfer of the Leasehold Estate by any third-party purchaser or assignee (other than by Leasehold Mortgagee or its designee in connection with its further assignment or transfer of the Leasehold Estate) which acquires its interest in the Leasehold Estate and this Lease pursuant to a foreclosure (whether by judicial proceedings or by virtue of any power of sale contained in any Leasehold Mortgage), or assignment in lieu of foreclosure. Lessor shall execute a written consent to such transfer, or provide a written denial of consent (which will include specific reasons for Lessor's denying consent), within forty-five (45) calendar days of receipt of such third-party purchaser or assignee's written request for such consent. If, in connection with Lessee's financing of its interest under this Lease, a prospective lender requests that additional or modified protections be incorporated into this Lease, Lessor shall review and reasonably approve such requests and timely amend this Lease as necessary and appropriate; provided, however, that such additions or modifications requested are generally applicable and utilized in financings of leasehold estates similar to the Leasehold Estate under this Lease, and that such requests do not materially and adversely affect Lessor's rights or materially increase Lessor's obligations. For the avoidance of doubt and notwithstanding any provision to the contrary in this Lease, the Leasehold Mortgagee or its designee who has acquired title to this Lease by way of foreclosure or assignment in lieu thereof shall have the right to sell and assign its interest in the Leasehold Estate without the consent of Lessor.
- 19.2.7 <u>Liability Limits</u>. In the event any third party or Leasehold Mortgagee acquires the Leasehold Estate upon foreclosure (whether judicial or non-judicial in nature) or by assignment in lieu of foreclosure, or acquires a leasehold estate in the Premises pursuant to the terms of a New Lease, such party, as the new lessee, shall be personally liable only for the obligations of the Lessee under this Lease (or, if applicable, the New Lease) arising during the period of time that such party holds title to the Leasehold Estate created hereby (or, if applicable, the New Lease), and such personal liability shall be limited to such new lessee's interest in the Project and the Premises.
- 19.2.8 <u>Notice of Proceedings</u>. Lessee shall give all Leasehold Mortgagees Notice of any arbitration, litigation, or condemnation proceedings, or of any pending adjustment of insurance claims, as each may relate to the Premises and any Leasehold Mortgagee shall have the right to intervene therein and shall be made a party to such proceedings. The Parties hereby consent to such intervention. In the event that any Leasehold Mortgagee shall not elect to intervene or become a party to the proceedings, such Leasehold Mortgagee shall receive Notice of and a copy of any award or decision made in connection therewith.

ARTICLE 20: EQUIPMENT LIENS

- **20.1** Lessee's Rights. If at any time or from time to time Lessee desires to enter into or grant any Equipment Lien that otherwise complies with this Lease, and provided that no uncured Event of Default has occurred and is continuing, then upon Lessee's request Lessor shall enter into such customary documentation regarding the Financed FF&E as Lessee reasonably requests, providing for matters such as: (a) waiver or subordination of any right to take possession of such Financed FF&E upon an Event of Default; (b) waiver or subordination of any other right, title, or interest in the Financed FF&E; and (c) agreements to enable the holder of such Equipment Lien to repossess such Financed FF&E if such holder exercises remedies under its Equipment Lien.
- **20.2** Required Provisions for Equipment Liens. If Lessee enters into any Equipment Lien, then Lessee shall: (i) not file (or cause or permit to be filed) such Equipment Lien as a lien against the Fee Estate or any part of the Fee Estate, but Lessee shall be permitted to file or cause to be filed a fixture filing attaching to Lessee's interest in the Premises relating to any Financed FF&E; and (ii) cause to be inserted in the documents for such Equipment Lien a provision to the following effect:

Notwithstanding anything to the contrary herein, this chattel mortgage, conditional sales agreement, title retention agreement, or security agreement shall not create or be filed as a lien against the Fee Estate.

ARTICLE 21: QUIET ENJOYMENT

So long as this Lease has not been terminated, Lessor covenants that Lessee shall and may peaceably and quietly have, hold, and enjoy the Premises for the Term, subject to the terms, covenants, conditions, provisions and agreements set forth in this Lease, without hindrance or disturbance by or from Lessor or anyone lawfully claiming by or through Lessor, and free of any encumbrance created or suffered by Lessor, except Permitted Exceptions.

ARTICLE 22: EVENTS OF DEFAULT; REMEDIES

- **22.1 Definition of "Event of Default"**. An "**Event of Default**" means the occurrence of any one or more of the following:
- **22.1.1** Monetary Default. If a Monetary Default occurs and continues for ten (10) days after Notice from Lessor, specifying in reasonable detail the amount of money not paid and the nature and calculation of each such payment.
- **22.1.2** <u>Prohibited Liens</u>. If Lessee fails to comply with any obligation regarding Prohibited Liens and does not begin to remedy such failure within thirty (30) days after Notice from Lessor and, thereafter, diligently pursue such remedy to completion.
- **22.1.3** Bankruptcy or Insolvency. If Lessee ceases to do business as a going concern, ceases to pay its debts as they become due or admits in writing that it is unable to pay its debts as they become due, or becomes subject to any Bankruptcy Proceeding (except an involuntary Bankruptcy Proceeding dismissed within ninety (90) days after commencement), or a custodian or trustee is appointed to take possession of, or an attachment, execution or other judicial seizure is made with respect to, substantially

all of Lessee's assets or Lessee's interest in this Lease (unless such appointment, attachment, execution, or other seizure was involuntary and is contested with diligence and continuity and vacated and discharged within ninety (90) days).

- **22.1.4** Nonmonetary Default. If any other Nonmonetary Default occurs and Lessee does not cure it within thirty (30) days after Notice from Lessor describing it in reasonable detail, or, in the case of a Nonmonetary Default that cannot, with due diligence, be cured within thirty (30) days from such Notice, if Lessee shall not (a) within thirty (30) days from Lessor's Notice advise Lessor of Lessee's intention to take all reasonable steps to cure such Nonmonetary Default; (b) duly commence such cure within such period, and then diligently prosecute such cure to completion; and (c) complete such cure within a reasonable time under the circumstances (not necessarily limited to thirty (30) days).
- **22.1.5** Other Events. The occurrence of any other event described as constituting an "Event of Default" under Section 6.7 in this Lease, which default is not cured within the time frame described in Section 6.7, or, if no cure period is described, then within the time frame described in Section 22.1.4.
- **Remedies**. If an Event of Default occurs, then Lessor shall, at Lessor's option (unless prohibited by Law), have any or all of the following remedies, all cumulative (i.e., the exercise of one remedy shall not preclude exercise of another remedy), in addition to such other remedies as may be available at law or in equity or under any other terms of this Lease. Lessor's remedies include:
- **22.2.1** Termination of Lessee's Rights. Lessor may terminate Lessee's right to possess the Premises by any lawful means, in which case this Lease and the Term shall terminate, such date of termination shall be the Expiration Date, and Lessee shall immediately surrender possession to Lessor. Notwithstanding the foregoing, no re-entry or taking of possession of the Premises by Lessor shall occur unless a written Notice that this Lease is terminated is given by Lessor after the giving of written Notice and expiration of any applicable cure period as required under this Lease, and such termination is otherwise in compliance with the terms and conditions of this Lease, or pursuant to a court order stating that this Lease is terminated. The effective date of termination of this Lease shall be as of the date set forth or provided in the Notice or order, as the case may be.
- **22.2.2** Taking Possession. Lessor may re-enter and take possession of the Premises with process of law, whether by summary proceedings (unless prohibited by Law) or otherwise, and remove Lessee (and all property of Lessee), after this Lease has been terminated in accordance with Section 22.2.1, and without thereby being liable for damages or guilty of trespass. This is intended to constitute an express right of re-entry by Lessor. Except as expressly provided in this Lease or prohibited by Law, Lessee, for and on behalf of itself and all persons claiming by, through or under Lessee, expressly waives any and all right of redemption provided by any Law, or re-entry or repossession or to restore the operation of this Lease if Lessee is dispossessed by a judgment or by warrant of any court or judge or in case of re-entry or repossession by Lessor or any expiration or termination of this Lease in accordance with its terms. No re-entry by Lessor, whether had or taken under summary proceedings or otherwise, shall absolve or discharge Lessee from liability under this Lease. The terms "enter," "re-enter," "entry," and "re-entry," as used in this Lease, are not restricted to their technical legal meanings.
- **22.2.3** <u>Suits Before Expiration Date</u>. Lessor may sue for damages and/or to recover Rent from time to time at Lessor's election; nothing in this Lease requires Lessor to wait until the date when this Lease or the Term would have expired absent an Event of Default and a resulting termination of this Lease.
- **22.2.4** Receipt of Moneys. No receipt of money by Lessor from Lessee after termination of this Lease, or after the giving of any notice of termination of this Lease, shall reinstate, continue, or extend this Lease or affect any notice theretofore given to Lessee, or waive Lessor's right to enforce payment of any Rent payable or later falling due, or Lessor's right to recover possession by proper remedy, except as this

Lease expressly states otherwise, it being agreed that after service of Notice to terminate this Lease or the commencement of suit or summary proceedings, or after final order or judgment for possession, Lessor may demand, receive, and collect any moneys due or thereafter falling due without in any manner affecting such notice, proceeding, order, suit or judgment, all such moneys collected being deemed payments on account of Lessee's liability.

- **22.2.5** No Waiver. No failure by Lessor to insist upon strict performance of any covenant, agreement, term, or condition of this Lease or to exercise any right or remedy upon a Default, and no acceptance of full or partial Rent during continuance of any such Default, shall waive any such Default or such covenant, agreement, term or condition. No covenant, agreement, term or condition of this Lease to be performed or complied with by Lessee, and no Default, shall be Modified except by a written instrument executed by Lessor. No waiver of any Default shall Modify this Lease. Each and every covenant, agreement, term, and condition of this Lease shall continue in full force and effect with respect to any other then-existing or subsequent Default of such covenant, agreement, term or condition of this Lease.
- **22.2.6** Receiver. Lessor shall be entitled as a matter of right, by *ex parte* order or otherwise, to the appointment without bond of a receiver of the Premises, and of the rents, revenues, income and profits generated from the Premises, without regard to the value of the Premises or the solvency of any Person liable for the payment of any monetary obligation under this Lease, and regardless of whether Lessor has an adequate remedy available to Lessor under this Lease or under applicable Laws.
- **22.2.7** <u>Damages</u>. Lessor may recover from Lessee all damages Lessor incurs by reason of an Event of Default that occurs and is continuing beyond any applicable cure period, including reasonable costs of recovering possession, re-letting the Premises, and any and all other damages legally recoverable by Lessor, and reimbursement of Lessor's reasonable out-of-pocket costs, including Legal Costs. Lessor may recover such damages at any time after the occurrence and continuation of an Event of Default beyond any applicable cure period, including after expiration of the Term. Notwithstanding any Law to the contrary, Lessor need not commence separate actions to enforce Lessee's obligations for each month's accrual of damages for the occurrence and continuation of an Event of Default beyond any applicable cure period, but may bring and prosecute a single combined action for all such Rent and damages.
- **22.2.8** <u>Injunction of Breaches.</u> Whether or not an Event of Default has occurred, Lessor may obtain a court order enjoining Lessee from continuing any Default or from committing any threatened Default. Lessee specifically and expressly acknowledges that damages may not constitute an adequate remedy for any Nonmonetary Default.
- **22.2.9** <u>Continue Lease</u>. Lessor may at Lessor's option maintain Lessee's right to possession. In such case, this Lease shall continue and Lessor may continue to enforce it, including the right to collect Rent when due and any remedies for nonpayment.
- **22.2.10** Restoration Funds. Upon any termination of this Lease resulting from an Event of Default, to the extent that Lessor or Depository then holds any Restoration Funds, such Restoration Funds shall be applied first toward the applicable Restoration with any Restoration Funds remaining after completion of Restoration being applied in the manner set forth in Section 13.2 for the disposition of Property Insurance Proceeds upon a Casualty Termination, subject to any claims for damages resulting from such Event of Default.
- **22.3 Proceeds of Reletting**. Lessor shall apply any proceeds of any re-letting as follows, without duplication, but including Default Interest on all such sums:
- **22.3.1** <u>Lessor's Costs</u>. First, to pay to itself the cost and expense of terminating this Lease, re-entering, retaking, repossessing, repairing, performing any Construction, and the cost and expense of

removing all persons and property therefrom, including in such costs reasonable and customary brokerage commissions and Legal Costs;

- **22.3.2** <u>Preparation for Reletting.</u> Second, to pay to itself the cost and expense reasonably sustained in securing any new lessees and other occupants, including in such costs all brokerage commissions, Legal Costs, and any other reasonable costs of preparing the Premises for re-letting;
- **22.3.3** Costs of Maintenance and Operation. Third, to the extent that Lessor shall maintain and operate the Premises, to pay to itself the reasonable cost and expense of doing so; and
- **22.3.4** Residue. Fourth, after payment to itself of any balance remaining on account of Lessee's liability to Lessor, to Lessee or any Leasehold Mortgagee, as applicable.
- **22.4** Lessee's Late Payments; Late Charges. If Lessee fails to make any payment to Lessor required under this Lease within ten (10) days after such payment is first due and payable, then in addition to any other remedies of Lessor, and without reducing or adversely affecting any of Lessor's other rights and remedies, Lessee shall pay Lessor within ten (10) days after demand Default Interest on such late payment, beginning on the date such payment was first due and payable and continuing until the date when Lessee actually makes such payment. In addition, and without limiting any other rights or remedies of Lessor, Lessee shall pay Lessor, as Additional Rent, an administrative charge equal to five percent (5%) of any payment that Lessee fails to pay within thirty (30) days after such payment is first due and payable. Such administrative charge is intended to compensate Lessor for the inconvenience and staff time incurred by Lessor to handle the late or missed payment, shall not be deemed a penalty or compensation for use of funds, and shall not be credited against any other obligations of Lessee under this Lease.
- **Lessor's Right to Cure**. If Lessee at any time fails to make any payment or take any action this Lease requires, then Lessor, after twenty (20) Business Days' Notice to Lessee and to any Leasehold Mortgagee, or in an emergency with such notice (if any) as is reasonably practicable under the circumstances, and without waiving or releasing Lessee from any obligation or Default and without waiving Lessor's right to take such action as this Lease may permit as a result of such Default, may (but need not) make such payment or take such action. Lessee shall reimburse Lessor, as Additional Rent, for an amount equal to (a) all reasonable sums paid, and reasonable costs and expenses (including Legal Costs) incurred, by Lessor in exercising its cure rights under this <u>Section 22.5</u>; and (b) Default Interest on the amounts in clause (a) above.
- **22.6** Holding Over. If for any reason or no reason Lessee remains in the Premises after the Expiration Date, then Lessor will suffer injury that is substantial, difficult, or impossible to measure accurately. Therefore, if Lessee remains in the Premises after the Expiration Date, for any reason or no reason, then in addition to any other rights or remedies of Lessor, Lessee shall pay to Lessor, as liquidated damages and not as a penalty, for each month (prorated daily for partial months) during which Lessee holds over after the Expiration Date, a sum equal to twice the then market rental rate for the Premises, plus all Additional Rent otherwise payable under this Lease during the holdover period. Nothing contained in this Section 22.6 shall be construed as consent by Lessor to any holding over by Lessee, and Lessor expressly reserves the right to require Lessee to surrender possession of the Premises to Lessor as provided in this Lease upon the expiration or earlier termination of this Lease.
- **Waivers: Jury Trial, Redemption**. Lessor and Lessee irrevocably waive all rights to trial by jury in any action, proceeding, counterclaim, or other litigation arising out of or relating to this Lease, the relationship of Lessor and Lessee regarding the Premises, enforcement of this Lease, Lessee's use or occupancy of the Premises, any claim of injury or damage arising between Lessor and Lessee, or any actions of Lessor in connection with or relating to the enforcement of this Lease. Lessee waives any right of redemption provided for by Law.

- **22.8** Accord and Satisfaction; Partial Payments. No payment by Lessee or receipt by Lessor of a lesser amount than the amount owed under this Lease shall be deemed to be other than a part payment on account by Lessee. Any endorsement or statement on any check or letter accompanying any check or payment of Rent shall not be deemed an accord or satisfaction. Lessor may accept any such check or payment without prejudice to Lessor's right to recover the balance of such Rent or pursue any other remedy.
- **22.9** Lessor's Default. Lessor shall be in default under this Lease if Lessor fails to cure any breach of its obligations under this Lease within thirty (30) days after Notice from Lessee describing such breach in reasonable detail, or, in the case of a breach that cannot, with due diligence, be cured within thirty (30) days from such Notice, if Lessor shall not (a) within thirty (30) days from Lessee's Notice advise Lessee of Lessor's intention to take all reasonable steps to cure such default; (b) duly commence such cure within such period, and then diligently prosecute such cure to completion; and (c) complete such cure within a reasonable time under the circumstances (not necessarily limited to thirty (30) days).
- **22.10** <u>Miscellaneous</u>. Lessor and Lessee further agree as follows with respect to any Defaults and Lessor's rights and remedies:
- **22.10.1** Survival. Termination or expiration of this Lease resulting from a Default shall not relieve any party of any claims against it that arise under this Lease before the Lease expires or is terminated.
- **22.10.2** No Double Recovery. In no event shall Lessor be entitled, directly or indirectly, to recover twice for the same element of Lessor's damages.

ARTICLE 23: END OF TERM

Upon any Expiration Date: (a) all Improvements, FF&E, and Building Equipment shall become Lessor's property; (b) Lessee shall deliver to Lessor possession of the Premises, in the condition this Lease requires, subject to any Loss that this Lease does not require Lessee to Restore; (c) Lessee shall surrender any right, title, or interest in and to the Premises and deliver such evidence and confirmation thereof as Lessor reasonably requires; (d) Lessee shall deliver the Premises free and clear of all: Subleases, and liens except (1) Permitted Exceptions existing as of the Commencement Date or consented to by Lessor, (2) Subleases executed pursuant to this Lease or consented to by Lessor, (3) liens that Lessor or any of its agents caused, and (4) Subleases or other agreements required to remain in place due to certain tenants' rights set forth in applicable Law; (e) Lessee shall assign to Lessor, and give Lessor copies or originals of, all assignable licenses, permits, contracts, warranties, and guarantees then in effect for the Premises, along with copies of all operating manuals and similar documentation relating to all Improvements, FF&E, and Building Equipment, and the current year's operating budget for the Premises (including applicable back-up information); (f) the parties shall cooperate to achieve an orderly transition of operations from Lessee to Lessor without interruption, including delivery of such information, books and records (or copies thereof) as Lessor reasonably requires; (g) if such plans are available, Lessee shall provide Lessor with a complete set of as-built plans and specifications for all Improvements, if any, added to the Premises since the Commencement Date; (h) the parties shall adjust for Real Estate Taxes and all other expenses and income of the Premises and any prepaid Rent and shall make such payments as shall be appropriate on account of such adjustment in the same manner as for a sale of the Premises (but any sums otherwise payable to Lessee shall first be applied to cure any Default); (i) the parties shall terminate the recorded Lease; and (j) to the extent assignable, Lessee shall assign to Lessor, and Lessor shall reimburse Lessee for, all utility and other service provider deposits for the Premises.

ARTICLE 24: NOTICES

All Notices shall be in writing and addressed to Lessor and Lessee, as applicable, and their designated copy recipients, which shall be delivered simultaneously with such Notice to such party, as set forth in **Exhibit E** attached hereto. Notices (including any required copies as set forth in **Exhibit E**) shall be delivered by (a) United States certified or registered mail, postage prepaid, return receipt requested, or (b) a nationally recognized overnight courier service, to the addresses set forth in **Exhibit E**. Notices shall be deemed delivered (a) three (3) Business Days after the date it is posted if sent by U.S. Mail (provided no postal strike or other disruption of postal service is then in effect), or (b) the date the overnight courier delivery is made (or when delivery has been attempted, as evidenced by the written report of the courier service) to such address(es). Either party may change its address by Notice in compliance with this Lease. Notice of such a change shall be effective only upon receipt. Any party giving a Notice may request the recipient to acknowledge receipt of such Notice. The recipient shall promptly comply with any such request, but failure to do so shall not limit the effectiveness of any Notice. Any attorney may give any Notice on behalf of its client.

ARTICLE 25: ADDITIONAL DELIVERIES; THIRD PARTIES

- **25.1** Estoppel Certificates. As often as may be necessary, each party to this Lease (a "Requesting Party") may require the other party (a "Certifying Party") to execute, acknowledge, and deliver to the Requesting Party (or directly to a designated third party) up to four (4) original counterparts of an estoppel certificate in such form as may be reasonably required by the Requesting Party, indicating therein any exceptions thereto that may exist at that time, and shall also contain any other information reasonably requested by the Requesting Party. The Certifying Party shall sign, acknowledge, and return such estoppel certificate within fifteen (15) days after request, even if the Requesting Party is in Default. Any estoppel certificate may be relied upon by the Requesting Party (and any Person on behalf of whom the Requesting Party requested such estoppel certificate) and shall bind the Certifying Party.
- **25.2 Further Assurances**. Each party shall execute and deliver such further documents, and perform such further acts, as may be reasonably necessary to achieve the parties' intent in entering into this Lease.
- **25.3** Memorandum of Lease. This Lease shall not be recorded; provided, however, that either Lessor or Lessee may elect to have a memorandum of this Lease recorded in the Bureau or filed in the Office of the Assistant Registrar of the Land Court of the State of Hawai'i, as appropriate. Such memorandum shall be sufficient to give constructive notice of the tenancy hereby created and setting forth a description of the Premises, the term of this Lease and any other provisions agreed to by the parties hereto (or required by a Leasehold Mortgagee), and shall be executed by the parties hereto. If the parties amend this Lease, then the parties shall record a memorandum of such amendment. Notwithstanding the foregoing, this Lease shall be recorded if such recordation is required by a Leasehold Mortgagee or a prospective Leasehold Mortgagee.
- **25.4** <u>Modification</u>. Any Modification of this Lease must be in writing signed both Lessor and Lessee to be binding on the parties. Any material Modification, including but not limited to, any restatement, cancellation, rejection, surrender or termination of this Lease (a "Material Modification") that may be agreed upon between Lessor and Lessee, shall require Lessee to obtain the prior written consent of the most senior Leasehold Mortgagee and, if such Material Modification would be effective during the Compliance Period, the prior written consent of any Limited Partner.

- **25.5** Lessor's Right to Amend. Any provision herein to the contrary notwithstanding, during the term of this Agreement, Lessor reserves the right, at any time, to amend this Lease in order to assure compliance with all applicable HUD, County, State and other federal statutes, laws, and regulations. Lessor shall provide all such amendments in writing to the Lessee. Lessee agrees that it shall immediately take any and all reasonable steps to comply with such amendments as required by Law from and after the effective date of such amendment; provided that if Lessee, in its sole discretion, determines that any such amendment would render the continued operation of the Project to be in violation of Lessee's affordability covenants and/or economically infeasible, then notwithstanding any provision to the contrary in this Lease, Lessee shall have the right, among other rights and remedies, to terminate this Lease upon one hundred eighty (180) days advance written notice to Lessor.
- **25.6** Successors and Assigns. This Lease shall bind and benefit Lessor and Lessee and their successors and assigns, but this shall not limit or supersede any Transfer restrictions. Nothing in this Lease confers on any Person (except Lessor, Lessee, and any Leasehold Mortgagees) any right to insist upon, or to enforce against Lessor or Lessee, the performance or observance by either party of its obligations under this Lease.

ARTICLE 26: CULTURAL AND ARCHEOLOGICAL

26.1 <u>Native Hawaiian Rights</u>. Lessee shall respect and recognize any and all rights of native Hawaiians to exercise traditional rights, customs, practices, prerogatives, privileges and usufructs on the Premises, if any, subject to and in accordance with applicable Laws.

26.2 Human Remains; Artifacts; Historical Items.

- **26.2.1** Discovery. In the event any human remains, traditional cultural items, artifacts or historical items (collectively "Historic Items") are discovered on the Premises, Lessee shall immediately report such discovery to Lessor. Upon such discovery and subject to Lessor's approval and if required by applicable Laws, Lessee shall, at Lessee's sole expense: (a) cause all excavation or other activity in the immediate area that may damage the Historic Items or the potential historic site to cease; (b) cause the site to be stabilized and secured to temporarily protect the Historic Items against damage, theft, or both; and (c) cause the Historic Items to be left untouched so that their cultural, archaeological or historical context may be accurately documented and to honor cultural sensitivities related to the Historic Items; provided, however, that if artifacts or historical items are found without human remains, and if leaving the artifacts or historical items in their stabilized and secured site poses a substantial risk of loss or damage to all or part of them, Lessee shall cause such Historic Items to be removed and safeguarded elsewhere.
- **26.2.2** <u>Human Remains.</u> In the event Lessee discovers human remains, Lessee shall, at Lessee's sole expense and in addition to the duties set forth in <u>Section 26.2.1</u>, (a) report the discovery as soon as possible to Lessor, the Historic Preservation Division of the Department of Land and Natural Resources of the State ("**SHPD**"), the appropriate medical examiner or coroner, and the appropriate police department, and (b) cause to be prepared, by an archeologist reasonably acceptable to Lessor, a mitigation and/or burial treatment plan reasonably acceptable to Lessor and to SHPD or the burial council having jurisdiction over such matters. Lessor and Lessee shall comply with all Laws applicable to the handling of such human remains, and shall work together to formulate and carry out such mitigation or burial treatment plan.
- **26.2.3** <u>Lessor's Reservation</u>. If any Historic Items are discovered, then Lessor shall have the right at all reasonable times to enter the Premises for the purposes of searching for, exploring for, and removing any of the Historic Items for preservation as permitted by Law.

- **26.2.4** <u>Studies by Lessee</u>. In the event any archaeological studies or historic preservation studies are sought to be conducted in or on the Premises, by Lessee or anyone acting by or through Lessee, Lessee shall provide a complete copy of the results of such studies to Lessor promptly upon completion thereof.
- **26.2.5** Lessor's Right to Historic Items; No Liability. Lessee shall have no right, title or interest whatsoever with respect to any Historic Items discovered on or about the Premises. As between Lessor and Lessee, Lessor shall retain ownership of any Historic Items discovered on or about the Premises to the extent private ownership of the Historic Items by Lessor is permitted under applicable Laws, and in any event, Lessor shall retain the exclusive right to act as, and to exercise all rights of, the landowner under applicable Laws. Lessor shall not be responsible for any damages or other liabilities that may result from cessation of excavation or construction, or from Lessee's compliance with provisions of this Article 26 and applicable Laws.

ARTICLE 27: MISCELLANEOUS

- Due Authorization and Execution. Lessor represents and warrants that Lessor is the owner of the Fee Estate, has full right, title, authority and capacity to execute and perform this Lease and any other agreements and documents to which Lessor is a party and referred to or required by this Lease (collectively, the "Lease-Related Documents"); the execution and delivery of the Lease-Related Documents have been duly authorized by all requisite actions of Lessor; the Lease-Related Documents constitute valid, binding, and enforceable obligations of Lessor; and neither the execution of the Lease-Related Documents nor the consummation of the transactions they contemplate violates any agreement (including Lessor's organizational documents), contract, or other restriction to which Lessor is a party or is bound. Lessee makes to Lessor representations and warranties reciprocal to those in the preceding sentence and, in addition, represents and warrants that Lessee is qualified to do business in the State. Except for the representation and warranty set forth in the last sentence of this Section 27.1, both parties' representations and warranties in this Section 27.1 shall continue to apply in full force and effect throughout the Term as if made continuously during the Term. Lessor represents and warrants to Lessee, as of the Commencement Date, to Lessor's actual knowledge and without undertaking any independent investigation, (a) there are no actions, suits, material claims, legal proceedings or any proceeding pending or threatened before any court or Governmental agency, with respect to the Premises, including but not limited to, eminent domain, and (b) Lessor has not received notice of any actions, suits, material claims, legal proceedings or any other proceeding impairing its interests in the Fee Estate that could prevent Lessor's execution nor performance of this Lease or prevent Lessee's ability to construct, develop and operate the Project.
- **Costs and Expenses; Legal Costs**. In the event of any litigation or dispute between the parties, or claim made by either party against the other, arising from this Lease or the Lessor-Lessee relationship under this Lease, or Lessor's enforcement of this Lease upon a Default, or to enforce or interpret this Lease or seek declaratory or injunctive relief in connection with this Lease, or to exercise any right or remedy under or arising from this Lease, or to regain or attempt to regain possession of the Premises or terminate this Lease, or in any Bankruptcy Proceeding affecting the other party to this Lease, the prevailing party shall be entitled to reimbursement of its Legal Costs with Default Interest and all other reasonable costs and expenses incurred in enforcing this Lease or curing the other party's default.
- **27.3 No Consequential Damages**. Whenever either party may seek or claim damages against the other party (whether by reason of a breach of this Lease by such party, in enforcement of any indemnity obligation, for misrepresentation or breach of warranty, or otherwise), neither Lessor nor Lessee shall seek, nor shall there be awarded or granted by any court, arbitrator, or other adjudicator, any speculative, consequential, collateral, special, punitive, or indirect damages, whether such breach shall be willful,

knowing, intentional, deliberate, or otherwise, except as otherwise expressly permitted by this Lease. The parties intend that any damages awarded to either party shall be limited to actual, direct damages sustained by the aggrieved party. Neither party shall be liable for any loss of profits suffered or claimed to have been suffered by the other.

- **27.4 No Waiver by Silence**. Failure of either party to complain of any act or omission on the part of the other party shall not be deemed a waiver by the non-complaining party of any of its rights under this Lease. No waiver by either party at any time, express or implied, of any breach of this Lease shall waive such breach or any other breach.
- **Performance Under Protest**. If a dispute arises about performance of any obligation under this Lease, the party against which such obligation is asserted shall have the right to perform it under protest, which shall not be regarded as voluntary performance. A party that has performed under protest may institute appropriate proceedings to recover any amount paid or the reasonable cost of otherwise complying with any such obligation.
- **27.6 Survival**. All rights and obligations that by their nature are to be performed after any termination of this Lease shall survive any such termination.
- **27.7 Unavoidable Delay**. Each party's obligation to perform or observe any nonmonetary obligation under this Lease shall be suspended during such time as such performance or observance is prevented or delayed by Unavoidable Delay.
- **27.8 Broker**. Each party: (a) represents and warrants that it did not engage or deal with any broker or finder in connection with this Lease, and no person is entitled to any commission or finder's fee on account of any agreement or arrangement made by such party; and (b) shall Indemnify the other party against any breach of such representation.
- **27.9 Service of Process**. Lessee and every assignee shall either be domiciled in the State or shall, effective upon the date of this Lease (for the original Lessee) or upon the date of said assignment (for an assignee), designate in writing an agent who is domiciled in the State upon whom service of notice or process may be made at all times (if applicable, Lessee's first such agent for service of process is designated in **Exhibit F)**. Service of summons or other legal process upon said agent shall be conclusively deemed to be complete upon Lessee and shall authorize the court from which such summons or legal process has issued to proceed in all respects as in the case of service personally made upon an individual. In the event Lessee fails to designate said agent for the service of process, or upon the death or absence of said agent, unless a successor shall be promptly named, the Director of Commerce and Consumer Affairs of the State shall be deemed Lessee's or assignee's agent for service of notice and process, and any notice or process served upon said designee or said Director of Commerce and Consumer Affairs shall have the force and effect of personal service upon Lessee or said assignee in all matters respecting this Lease and the enforcement thereof. Lessee and every assignee shall be duly qualified by the Director of Commerce and Consumer Affairs to do business in the State.
- **27.10** Sexual Harassment Policy. At all times during the Term, Lessee shall have and enforce a policy prohibiting sexual harassment in accordance with Article 18 of Chapter 1 of the Revised Ordinances of Honolulu 1990. Lessee may obtain a copy of said Article at the Office of the City Clerk, Honolulu Hale, 530 South King Street, Honolulu, Hawai'i.
- **27.11 Non-Discrimination Policy**. Lessee shall not discriminate against any employee or applicant for employment based on race, color, national origin, religion, sex, sexual orientation, familial status, or disability, and Lessee shall comply with the provisions included in any agreement with the County pertaining to discrimination.

27.12 Neither Party Agent, Joint Venturer or Partner of the Other. Neither party hereto shall be construed to be an agent of, or a joint venture or partner with, the other party.

ARTICLE 28: INTERPRETATION, EXECUTION, AND APPLICATION OF LEASE

- **28.1** <u>Captions</u>. The captions of Articles, Sections, items and paragraphs are for convenience and reference only and shall not be deemed to limit, construe, affect or alter the meaning of such Articles, Sections, items and paragraphs.
- **28.2** <u>Counterparts</u>. This Lease may be executed in counterparts with the same effect as if both parties hereto had executed the same document. Both counterparts shall be construed together and shall constitute a single lease document.
- **28.3 Delivery of Drafts**. Neither party shall be bound by this Lease unless and until such party shall have executed and delivered this Lease. The submission of draft(s) or comment(s) on drafts shall not bind the parties, nor shall such draft(s) and comment(s) be considered in interpreting this Lease.
- **Entire Agreement**. This Lease contains all terms, covenants, and conditions about the Premises. The parties have no other understandings or agreements, oral or written, about the Premises or Lessee's use or occupancy of, or any interest of Lessee in, the Premises except for any agreements referenced in this Lease, and except for any provisions from the Development Agreement that, by their terms are applicable to the Premises and intended to survive the closing of this leasing transaction (in which case such provision shall not be deemed legally merged into this Lease but, instead, shall be deemed incorporated into this Lease to the extent applicable).
- **28.5 Governing Law and Venue.** This Lease, its interpretation and performance, the relationship between the parties, and any disputes arising from or relating to any of the foregoing, shall be governed, construed, interpreted, and regulated under the laws of the State, without regard to principles of conflict of laws. Any legal action hereunder shall be filed in the Hawai'i judicial system only, and Lessor and Lessee hereby unconditionally submit themselves to the jurisdiction of the courts of the State in the circuit where the Premises are located, and the United States District Court for the District of Hawai'i, and waive the right to assert that such courts are in an inconvenient forum.
- **28.6 Partial Invalidity**. If any term or provision of this Lease or its application to any party or circumstance shall to any extent be invalid or unenforceable, then the remainder of this Lease, or the application of such term or provision to persons or circumstances except those as to which it is invalid or unenforceable, shall not be affected by such invalidity. All remaining provisions of this Lease shall be valid and be enforced to the fullest extent Law allows.
- **Principles of Interpretation**. No inference in favor of or against any party shall be drawn from the fact that such party has drafted any part of this Lease. The parties have both participated substantially in its negotiation, drafting, and revision, with advice from counsel and other advisers. A term defined in the singular may be used in the plural, and vice versa, all in accordance with ordinary principles of English grammar, which also govern all other language in this Lease. The words "include" and "including" shall be construed to be followed by the words: "without limitation." Each of these terms shall be interpreted as if followed by the words "(or any part of it)" except where the context clearly requires otherwise: Building Equipment; FF&E; Fee Estate; Improvements; Land; Leasehold Estate; Premises; and any other similar collective noun. Every reference to any document, including this Lease, refers to such document as Modified from time to time (except, at Lessor's option, any Modification that violates this Lease), and includes all exhibits, schedules and riders to such document.

- **28.8** <u>Time of the Essence</u>. Except as otherwise expressly provided in this Lease, time is of the essence with respect to all provisions of this Lease.
- **28.9** <u>Computation of Deadlines</u>. If a due date determined under this Lease falls on a Saturday, Sunday or official State, County or federal holiday, such due date will be deemed to be the next Business Day.
- **28.10** <u>Joint and Several</u>. If there is more than one Person comprising Lessee, the obligations imposed upon such Persons under this Lease shall be joint and several.
- **28.11** Relationship Between Lease and Development Agreement. In the event of any conflict or inconsistency between this Lease and the Development Agreement, this Lease shall control.

[SIGNATURES APPEAR ON NEXT PAGE]

IN TESTIMONY WHEREOF, Lessor and Lessee have caused this Lease to be signed as of the day and year first above written.

LESSOR:	
Date of execution by Lessor:, 20	
APPROVAL RECOMMENDED:	CITY AND COUNTY OF HONOLULU, a municipal corporation of the State of Hawai'i
Catherine A. Taschner, Acting Director Department of Land Management	By Department of Budget and Fiscal Services By Name: Andrew T. Kawano Title: Director
APPROVED AS TO FORM AND LEGALITY:	Title. Director
Deputy Corporation Counsel Department of the Corporation Counsel	
LESSEE: Date of execution by Lessee:, 20	HALEWILIKO HIGHLANDS LP, a Hawaii limited partnership
	By: HKI Halewiliko LLC, a Hawaii limited liability company Its General Partner
	By: Hui Kauhale, Inc., a Hawaii nonprofit corporation Its Sole Member
	By Name: Marian Gushiken Title: President

EXHIBIT A

LEGAL DESCRIPTION AND ENCUMBRANCES

All that certain parcel of land situate at Aiea, District of Ewa, City and County of Honolulu, State of Hawaii, described as follows:

Lot 6, area 150,144 square feet, more or less, as delineated on the map entitled "AIEA INDUSTRIAL SUBDIVISION", which said map was filed in the Bureau of Conveyances of the State of Hawaii (the "Bureau") as File Plan No. 2282.

Being the premises conveyed to the City and County of Honolulu, a municipal corporation of the State of Hawaii, by Warranty Deed and Reservation of Easement dated December 5, 2002, recorded in the Bureau;

SUBJECT, HOWEVER, to:

- 1. Mineral and water rights of any nature in favor of the State of Hawaii.
- 2. The rights of the United States of America, State of Hawaii, the municipality and the public, in and to that part of the premises in question falling in the bed of Aiea Stream; also the rights of the riparian owners in and to the free and unobstructed flow of the water of said stream, if any.
 - Any easement for water course over that portion of said land lying within the banks of Aiea and any changes in the boundary lines of said land that have occurred or may hereafter occur from natural causes.
- 3. Cable Line Easement dated May 2, 1950 in favor of the United States of America for communication cable purposes, recorded June 9, 1950 in the Bureau in Liber 2354, Page 150.
 - NOTE: The United States of America remises, releases and forever quitclaims a portion of the property acquired by the foregoing easement by Quitclaim Deed recorded January 23, 1970 in the Bureau in Book 6865, Page 32.
- 4. Cable Line Easement dated April 18, 1960 in favor of the United States of America for communication cable purposes, recorded April 27, 1960 in the Bureau in Liber 3823, Page 104.
- 5. Final Order of Condemnation (Civil No. 6683) claimed by the City and County of Honolulu, a municipal corporation of the State of Hawaii, for an easement for a right-of-way for sanitary sewer purposes, dated February 14, 1963, recorded February 25, 1963 in the Bureau in Liber 4464, Page 228.
- 6. Final Order of Condemnation (Civil No. 11973) claimed by the City and County of Honolulu, a municipal corporation of the State of Hawaii, for an easement for a right-of-way for sanitary sewer purposes, dated October 14, 1963, recorded October 20, 1963 in the Bureau in Liber 4620, Page 445.
- 7. Terms and provisions as contained in Unilateral Agreement for Conditional Zoning executed by California and Hawaiian Sugar Company, a California corporation, dated May 24, 1974, recorded May 28, 1974 in the Bureau in Liber 9926, Page 439, as amended by instrument recorded February 19, 1999 in the Bureau as Document No. 99-025334.

- 8. Grant dated August 10, 1982 in favor of the City and County of Honolulu, a municipal corporation of the State of Hawaii, for an easement for sanitary sewer purposes over and across Easement A, recorded November 2, 1984 in the Bureau in Liber 18247, Page 400.
- 9. Restriction of vehicle access rights along Hakina Street as shown on File Plan No. 2282.
- 10. Easement E for sanitary sewer purposes as shown on File Plan No. 2282.
- 11. Easement F for landscaping purposes as shown on File Plan No. 2282.
- 12. Terms and provisions as contained in Limited Warranty Deed recorded November 30, 1994 in the Bureau as Document No. 94-196299.
- 13. Terms and provisions as contained in Warranty Deed and Reservation of Easement dated December 5, 2002, recorded December 5, 2002 in the Bureau as Document No. 2002-216975.
- 14. Terms and provisions as contained in instrument, entitled 201H Agreement Halewiliko Highlands Senior Housing, executed by the City and County of Honolulu, a political subdivision of the State of Hawaii, dated February 15, 2022, recorded March 2, 2022 in the Bureau as Document No. A-80960285.
- 15. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Julee N. Nishimura, Trustee of the Julee N. Nishimura Trust, an unrecorded Revocable Living Trust dated November 25, 2005, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated April 26, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060512.

NOTE: A tile wall and a CRM wall from Lot 170-B encroaches onto Lot 6.

16. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Gwen N. Zakahi, Trustee of the Gwen N. Zakahi Revocable Living Trust dated March 10, 2010, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated May 31, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060513.

NOTE: A tile wall from Lot 170-A encroaches onto Lot 6.

17. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Paul Masao Murato, Trustee under that certain unrecorded Paul Masao Murato Revocable Living Trust dated August 30, 1995, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated April 25, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060514.

NOTE: A tile wall from Lot 171 encroaches onto Lot 6.

18. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Todd S. Inouye and Jill M. Y. N. W. Inouye, Trustees of the Todd S. Inouye Revocable Trust dated January 16, 2021, and Jill M. Y. N. W. Inouye and Todd S. Inouye, Trustees of the Jill M. Y. N. W. Inouye Revocable Trust dated January 16, 2021, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated May 31, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060515.

NOTE: A vinyl fence from Lot 64 encroaches onto Lot 6.

19. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by David N. Kodama, Trustee of the David N. Kodama Trust, and Karen K. Kodama, Trustee of the Karen K. Kodama Trust, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated April 7, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060516.

NOTE: A tile wall and a CRM wall from Lot 172 encroach onto Lot 6.

20. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Ronald Keola Taira and Trina Huynh Taira, husband and wife, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated May 31, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060517.

NOTE: A CRM wall from Lot 59 encroaches onto Lot 6.

21. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Loren Koon Sun Chang, Trustee under that certain unrecorded Loren K. S. Chang Revocable Living Trust dated June 9, 1995, and Laurie Hiroko Chang, Trustee under that certain unrecorded Laurie H. Chang Revocable Living Trust dated June 9, 1995, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated April 21, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060518.

NOTE: A tile wall and a concrete retaining wall with a fence post from Lot 58 encroach onto Lot 6.

22. Terms and provisions as contained in Encroachment Agreement and License (Wall) executed by Earl F. Ohara and Mae S. Ohara, Co-Trustees under that certain unrecorded Trust Agreement known as the Earl F. Ohara Revocable Trust dated October 2, 2014, and Mae S. Ohara and Earl F. Ohara, Co-Trustees under that certain unrecorded Trust Agreement known as Mae S. Ohara Revocable Trust dated October 2, 2014, and the City and County of Honolulu, a municipal corporation of the State of Hawaii, dated March 28, 2022, recorded June 20, 2022 in the Bureau as Document No. A-82060519.

NOTE: A tile wall from Lot 54 encroaches onto Lot 6.

END OF EXHIBIT A

EXHIBIT B

ADDITIONAL PERMITTED EXCEPTIONS

Permitted Exceptions shall include all of the following, as they existed on the Commencement Date:

- 1. All leases, Subleases, tenancies and rights of occupancy affecting the Premises caused or permitted by Lessee or by anyone claiming by, through, or under Lessee;
- 2. All rights, if any, for electricity, gas, telephone, water, cable television, and any other utilities to maintain and operate lines, cables, poles, and distribution boxes in, over, and upon the Premises;
- 3. Possible projections or encroachments of retaining walls, foundations, stoops, areas, steps, sills, trim, cornices, standpipes, fire escapes, casings, ledges, water tables, lintels, porticos, keystones, windows, hedges, copings, sidewalk elevators, fences, fire escapes, and the like, or similar projections or objects upon, under, or above any adjoining buildings or streets or avenues or those belonging to adjoining premises which encroach upon the Premises or within any set-back areas, and variations between the lines of record title and fences, retaining walls, hedges, and the like;
 - 4. Variations between the tax map and the record description of the Land;
- 5. Zoning, environmental, municipal, building, and all other laws, regulations or similar matters imposed by any federal, state, municipal, or local government or any public or quasi-public board, authority, or similar agency having jurisdiction over the Premises or any portion thereof;
- 6. All notes or notices of any violation of law or municipal ordinances, orders, or requirements noted in or issued by any Government having or asserting jurisdiction, now or hereafter affecting the Premises: and
- 7. The lien for all taxes, charges, rents, assessments, and any other governmental charges which are not yet due and payable.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

EXHIBIT C

FORM OF INCOME CERTIFICATION

VERIFICATION OF INCOME

RE:	Halewiliko Higl 99-385 Pohai Pl	nlands ace Aiea, Hawaii 967	01		
	Apartmen	t Number:	Initi	al Occupancy Date	e:
the u	ered fully, and truth	undersigned, being fir fully each of the follo tment development fo	wing quest	ions for all persons	s who are to occupy
	Α.	В.	C.	D.	E.
	Name of Members of the Household	Relationship to Head of Household	Age	Social Security Number	Place of Employment
	of the Household	Head of Household			
		Spouse			
•	d beginning this dat ne described in subi (a) <u>The amou</u>	ne anticipated income e, including income d tem (b) below, is \$ nt set forth above incl described in (b) below	escribed in udes all of	Item 1(a) below, b	out excluding all
		all wages and salarie payroll deductions;	s, overtime	pay, commissions	s, fees, tips and
	rental of real or p) net income from the ersonal property (with f capital indebtedness	out deduct	ing expenditures fo	or business expansio
	(ii Item 2(h) below):	i) interest and divider	nds (include	e all income from a	assets as set forth in

- (iv) the full amount of periodic payments received from social security, annuities, insurance policies, retirement funds, pensions, disability or death benefits and other similar types of periodic receipts;
- (v) payments in lieu of earnings, such as unemployment and disability compensation, workmen's compensation and severance pay;
- (vi) the maximum amount of public assistance available to the above persons;
- (vii) periodic and determinable allowances, such as alimony and child support payments and regular contributions and gifts received from persons not residing in the dwelling;
- (viii) all regular pay, special pay and allowances of a member of the Armed Forces (whether or not living in the dwelling) who is the head of the household or spouse; and
- (ix) any earned income tax credit to the extent it exceeds income tax liability.
 - (b) The following income is excluded from the amount set forth above:
 - (i) casual, sporadic or irregular gifts;
- (ii) amounts which are specifically for or in reimbursement of medical expenses;
- (iii) lump sum additions to family assets, such as inheritances, insurance payments (including payments under health and accident insurance and worker's compensation), capital gains and settlement for personal or property losses;
- (iv) amounts of educational scholarships paid directly to a student or an educational institution, and amounts paid by the government to a veteran for use in meeting the costs of tuition, fees, books and equipment, but in either case only to the extent used for such purposes;
- (v) hazardous duty to a member of the household in the armed forces who is away from home and exposed to hostile fire;
- (vi) relocation payments under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970;
- (vii) income from employment of children (including foster children) under the age of 18 years;
 - (viii) foster child care payments;

((ix) the value of coupon allotments under the Food Stamp Act of 1977;		
1973;	(x) payments to volunteers under the Domestic Volunteer Service Act of		
((xi) payments received under the Alaska Native Claims Settlement Act;		
	(xii) income derived from certain submarginal land of the United States rust for certain Indian tribes;		
	(xiii) payments on allowances made under the Department of Health and es' Low-Income Home Energy Assistance Program;		
((xiv) payments received from the Job Partnership Training Act;		
of Ottawa India	(xv) income derived from the disposition of funds of the Grand River Band ans; and		
(xvi) the first \$2000 of per capita shares received from judgment funds awarded by the Indian Claims Commission or the Court of Claims or from funds held in trust for an Indian tribe by the Secretary of Interior.			
2. If any of the persons described in Column A in the table above (or any person whose income or contributions were included in Item 1 above) has any savings, stocks, bonds, equity in real property or other form of capital investment (excluding interests in Indian trust lands), provide:			
	the total value of all such assets owned by all such persons:, and		
	the amount of income expected to be derived from such assets in the 12-month period commencing this date: \$		
3. (a) Will all of the persons listed in Column A above be, or have they been, full-time students during five calendar months of this calendar year at an educational institution (other than a correspondence school) with regular faculty and students?			
Yes	No		
(b) (Complete only if the answer to Item 3(a) is "Yes"). Is any such person (other than nonresident aliens) married and eligible to file a joint federal income tax return?			
Yes	No		

We acknowledge that all of the foregoing information is relevant to verification of our status as Tenants of a Regulated Unit and compliance with City affordable housing policies and ordinances and HUD affordable housing program requirements. We consent to the disclosure of such information by the City to HUD and other third-parties involved in the audit, monitoring or enforcement of such policies or programs.

We declare under pena	lty of perjury that the foregoing is true	e and correct.
Date:		
•	Head of Household	
	Spouse	

FOR COMPLETION BY PROJECT LESSEE ONLY:

I.	Calculation of eligible income:				
	(A)	Enter amount entered for entire household from \$	m Item 1 above:		
	(B)	Is the amount entered in Item 2(a) above is gre	eater than \$5,000? Yes / No		
	If the	If the answer to Item I(B) is "Yes," please complete subitems (i)-(iii) below:			
	(i)	The product of the amount entered in Item 2(a) above and the current passbook savings rate as determined by HUD: \$			
	(ii)	The amount entered in Item 2(b) above: \$			
	(iii)	Item I(i) minus Item I(ii) (if less than \$0, enter	r \$0): \$		
	(C)	TOTAL ELIGIBLE INCOME (Item I(A) plus \$	s, if applicable, Item I(B)(iii)):		
II.	Incon	ne Qualification:			
	 (A) Is the amount entered in Item I(C) less than or equal to thirty percent (30%) of median income for the Area? Yes / No (B) Is the amount entered in Item I(C) less than or equal to sixty percent (60%) of me income for the Area? Yes / No 				
(C) Is the amount entered in Item I(C) less than or equal to eighty percent (80%) median income for the Area? Yes / No			ual to eighty percent (80%) of		
III.	Numb	per of apartment unit assigned:	(enter here and on page one)		
		_			
			Lessee		

EXHIBIT D

[FORM OF CERTIFICATE OF CONTINUING PROGRAM COMPLIANCE]

CERTIFICATE OF CONTINUING PROGRAM COMPLIANCE

Capitalized ter	ms not otherwise	defined in thi	s Certificate of Contin	uing Program
Compliance shall have the me	eanings set forth	in that certain	Lease dated	, by and
between the City and County	of Honolulu, a	municipal cor	poration of the State of	of Hawaii (the
"City"), and	, a		(the "Lessee"),	with respect to
"City"), and the below-referenced Project	(such agreement,	the "Agreeme	ent").	
			, the under	
leased from the City that of				
Halewiliko Highlands (the "P				
Project was continually in cor				
Project were occupied by Quertify that the representations				
knowledge and belief.	s set fortil flereili a	are true and co	incer to the best of the	undersigned s
knowledge and benefit.				
Set forth belo	w are the name:	s of the Qua	lified Tenants who c	ommenced or
terminated occupancy during		_		
	ced Occupancy			
1		1		
2		2		
3		3		
Attached is a se	marata chaat lictin	a the number of	of each residential unit	and indicating
which residential units are oc	-	-		-
such residential units and the				
of residential units during the	_	or Quarried	Tonants who commen	oca occupancy
	No. of		Total Eligible	Size
Unit No. Class of Unit	Bedrooms	Rent	Income	(Sq. Ft.)
			<u> </u>	
<u> </u>				
				
				
Total Number of Units:				
Number of Lower Income Te	nants commencin	g occupancy t	his year:	

EXHIBIT E

NOTICE ADDRESSEES (INCLUDING REQUIRED COPY RECIPIENTS)

LESSOR: City and County of Honolulu

Department of Budget and Fiscal Services

530 South King Street Honolulu, Hawaii 96813

Attn: Director

With a copy to:

City and County of Honolulu Department of Land Management 650 South King Street, 3rd Floor Honolulu, Hawaii 96813

Attn: Director

With a copy to:

City and County of Honolulu

Department of the Corporation Counsel

530 South King Street, Room 110

Honolulu, Hawaii 96813 **Attn**: Corporation Counsel

LESSEE: Halewiliko Highlands LP

1001 Bishop Street, Suite 2880

Honolulu, Hawaii 96813

Attn: Marian Gushiken, President Telephone: (808) 523-8093

Facsimile: (808) 523-8826 Email: marian.gushiken@eahhousing.org

With a copy to:

Schneider Tanaka Radovich Andrew & Tanaka, LLLC

1100 Alakea Street, Suite 2100

Honolulu, Hawaii 96813 Attn: Tracy D. Tanaka Telephone: (808) 792-4207 Facsimile: (808) 792-3920

Email: tracy.tanaka@stratlaw.com

With a copy to:

U.S. Bancorp Community Development Corporation USB Hawaii State Investor II, LLC Project No. 28770 1307 Washington Avenue, Suite 300 Mail Code: SL MO RMCD

Mail Code: SL MO RMC St. Louis, MO 63103

Attention: Director of Asset Management

Telephone: (314) 335-2600 Facsimile: (314) 335-2601

With a copy to:

Kutak Rock LLP 1650 Farnam Street Omaha, NE 68102

Attention: Jill Goldstein, Esq.

EXHIBIT F

SERVICE OF PROCESS

Karen S. Seddon 1001 Bishop Street, Suite 2880 Honolulu, Hawaii 96813